August 1988

A 199

4

1

X

Ę

2

2

E

MIR FILE COPY

UILU-ENG-88-2244 DC-108

COORDINATED SCIENCE LABORATORY

College of Engineering Decision and Control Laboratory

SIMULTANEOUS DESIGN OF COMMUNICATION STRATEGIES AND CONTROL POLICIES IN STOCHASTIC SYSTEMS

Rajesh C. Bansal



23

88

9

000

UNIVERSITY OF ILLINOIS AT URBANA-CHAMPAIGN

Approved for Public Release. Distribution Unlimited.

		REPORT DOCU	MENTATION	PAGE		، مربق بالأعل اب الأرب الأمر
1a. REPORT SECURITY CLASSIFICATIO	N		1b. RESTRICTIVE	MARKINGS		
Unclassified			None			
2a. SECURITY CLASSIFICATION AUTH	IORITY		3. DISTRIBUTION / AVAILABILITY OF REPORT			
26. DECLASSIFICATION / DOWNGRADING SCHEDULE		Approved for public release; distribution unlimited				
PERFORMING ORGANIZATION REF	PORT NUMBE	R(S)	5. MONITORING	ORGANIZATION R	EPORT NUMBER	2(5)
UILU-ENG-88-2244 (DC	C-108)					
a. NAME OF PERFORMING ORGANI		6b. OFFICE SYMBOL	7a. NAME OF M	IONITORING ORGA	NIZATION	
Coordinated Science I	Lab	(if applicable)	Air Force	Office of Sc	ientific R	esearch
UNIVERSITY OF 111100	1.S 	N/A		the State and 710	Codel	
1101 W. Sprinofiald A	Avenue		Bolling Ai	r Force Base		
Urbana, IL 61801			Washington	, DC 20332	-	
BA. NAME OF FUNDING / SPONSORING	G Office	8b. OFFICE SYMBOL (If applicable)	9. PROCUREMEN	IT INSTRUMENT ID	ENTIFICATION N	UMBER
of Scientific Resear	ch		AFOSR-88	-0174		
k. ADDRESS (City, State, and ZIP Cod	fe)		10. SOURCE OF	FUNDING NUMBER	85	
Bolling Air Force Ba	se		PROGRAM ELEMENT NO.	PROJECT NO.	TASK NO.	WORK UNIT
Washington, DC 2033	2					
SIMULTANEOUS DESIGN SYSTEMS 2. PERSONAL AUTHOR(S) BANSAL, RAJESH C. 3a. TYPE OF REPORT Technologi	OF COMMUN	VERED	GIES AND CON	TROL POLICIE	Oay) 15. PAGE	ASTIC
SIMULTANEOUS DESIGN SYSTEMS 2. PERSONAL AUTHOR(S) BANSAL, RAJESH C. 3a. TYPE OF REPORT Technical 36. SUPPLEMENTARY NOTATION	OF COMMUN	NICATION STRATE	GIES AND CON 14. DATE OF REPO 1988 AUG	TROL POLICIE DRT (Year, Month, I SUST	ES IN STOCH Day) 15. page 215	
SIMULTANEOUS DESIGN SYSTEMS 12. PERSONAL AUTHOR(S) BANSAL, RAJESH C. 13a. TYPE OF REPORT Technical 15. SUPPLEMENTARY NOTATION	OF COMMUN	NICATION STRATE	GIES AND CON 14. DATE OF REPO 1988 AUG	TROL POLICIE DRT (Year, Month, I SUST	Oay) 15. PAGE 215	
SIMULTANEOUS DESIGN SYSTEMS 2. PERSONAL AUTHOR(S) BANSAL, RAJESH C. 33. TYPE OF REPORT Technical 36. SUPPLEMENTARY NOTATION 17. COSATI CODES	OF COMMUI	VERED TO	GIES AND CON 14. DATE OF REPC 1988 AUG Continue on reven	TROL POLICIE	Day) 15. PAGE 215	ASTIC COUNT
SIMULTANEOUS DESIGN SYSTEMS 12. PERSONAL AUTHOR(S) BANSAL, RAJESH C. 13a. TYPE OF REPORT Technical 16. SUPPLEMENTARY NOTATION 17. COSATI CODES FIELD GROUP SUB	OF COMMUN	NICATION STRATE VERED TO 18. SUBJECT TERMS (Stochastic cont nonclassical in	GIES AND CON 14. DATE OF REPO 1988 AUG Continue on reven trol, linear nformation p	TROL POLICIE ORT (Year, Month, 1 UST Polit necessary and -quadratic-G atterns, tea	Day) 15. PAGE 215 didentify by blo aussian pr am theory,	ASTIC COUNT ck number) oblems, information
SIMULTANEOUS DESIGN SYSTEMS 12. PERSONAL AUTHOR(S) BANSAL, RAJESH C. 13a. TYPE OF REPORT Technical 15. SUPPLEMENTARY NOTATION 17. COSATI CODES FIELD GROUP SUB	OF COMMUN	18. SUBJECT TERMS (Stochastic cont nonclassical in theory, zero-su	GIES AND CON 14. DATE OF REPC 1988 AUG Continue on revers trol, linear nformation p um games. number)	TROL POLICIE	Day) 15. PAGE 215 didentify by blo baussian pr im theory,	ASTIC COUNT ck number) oblems, information
SIMULTANEOUS DESIGN SYSTEMS 12. PERSONAL AUTHOR(S) BANSAL, RAJESH C. 13a. TYPE OF REPORT Technical 16. SUPPLEMENTARY NOTATION 17. COSATI CODES FIELD GROUP SUB 19. ABSTRACT (Continue on reverse We consider the pr policies in decentraliz because of the nonclass classes of such problem statistics for which th approach used consists joint strategies that a strategies fail to prov obtained that demonstra thesis can be viewed as for a general theory of and possibly partial st two-stage stochastic te	OF COMMUN 13b. TIME CC FROM I-GROUP if necessary oblem of ical nature ical n	18. SUBJECT TERMS (Stochastic cont nonclassical in theory, zero-su and identify by block (simultaneously astic systems. are of their in inear dynamics, lity of linear finding a lower is lower bound. ally optimal so nferiority of 1 nt prototype pro- age distributed l description. have made cont	GIES AND CON 14. DATE OF REPC 1988 AUG Continue on reven trol, linear nformation p um games. number) designing c Such proble formation st quadratic l strategies c r bound on t For some i lutions, exp inear design oblems, whic decision ma We have obt ributions to	TROL POLICIE ORT (Year, Month, I UST 	Day) 15. PAGE 215 215 215 215 215 215 215 215 215 215	ASTIC COUNT Count Co
SIMULTANEOUS DESIGN SYSTEMS 12. PERSONAL AUTHOR(S) BANSAL, RAJESH C. 13a. TYPE OF REPORT Technical 16. SUPPLEMENTARY NOTATION 17. COSATI CODES FIELD GROUP SUB 19. ABSTRACT (Continue on reverse We consider the pr policies in decentraliz because of the nonclass classes of such problem statistics for which th approach used consists joint strategies that a strategies fail to prov obtained that demonstra thesis can be viewed as for a general theory of and possibly partial st two-stage stochastic te 20. DISTRIBUTION/AVAILABILITY OF DUNCLASSIFIED/UNLIMITED	OF COMMUN 13b. TIME CC FROM -GROUP -GROUP 	18. SUBJECT TERMS (Stochastic coni nonclassical in theory, zero-su and identify by block / simultaneously astic systems. Jre of their in inear dynamics, lity of linear finding a lower is lower bound. ally optimal so inferiority of l nt prototype pro- age distributed l description. have made cont	GIES AND CON 14. DATE OF REPC 1988 AUG Continue on reven trol, linear nformation p um games. Dumber) designing c Such proble formation st quadratic l strategies c r bound on t For some i lutions, exp inear design oblems, whic decision ma We have obt ributions to 21. ABSTRACT SE Unclassi	TROL POLICIE ORT (Year, Month, I UST ommunication atterns, tea ommunication ins are diffi ructure. We oss function an be establ he cost, and nstances of licit nonlin is. The prob h are essent king under m ained some f wards the de CURITY CLASSIFIC	Coy) 15. PAGE 215 215 215 215 215 215 215 215 215 215	ASTIC COUNT ck number) oblems, information s and contr lve, mainly tified ussian tructing where linea gies are ed in this ng blocks l informati results fo of a genera
SIMULTANEOUS DESIGN SYSTEMS 12. PERSONAL AUTHOR(S) BANSAL, RAJESH C. 13a. TYPE OF REPORT Technical 16. SUPPLEMENTARY NOTATION 17. COSATI CODES FIELD GROUP SUB 19. ABSTRACT (Continue on reverse We consider the pr policies in decentraliz because of the nonclass classes of such problem statistics for which th approach used consists joint strategies that a strategies fail to prov obtained that demonstra thesis can be viewed as for a general theory of and possibly partial st two-stage stochastic te 20. DISTRIBUTION / AVAILABILITY OF DUNCLASSIFIED/UNLIMITED 22a. NAME OF RESPONSIBLE INDIVID	OF COMMUN 13b. TIME CC FROM 	NICATION STRATE NICATION STRATE TO 18. SUBJECT TERMS (Stochastic coni nonclassical in theory, zero-su and identify by block / simultaneously astic systems. are of their in inear dynamics, lity of linear finding a lower is lower bound. ally optimal so aferiority of l nt prototype pro- age distributed l description. have made cont PT. □ DTIC USERS	GIES AND CON 14. DATE OF REPC 1988 AUG Continue on reven trol, linear nformation p um games. number) designing c Such proble formation st quadratic l strategies c r bound on t For some i lutions, exp inear design oblems, whic decision ma We have obt ributions to 21. ABSTRACT SE Unclassi 22b. TELEPHONE	DRT (Year, Month, I DUST ommunication atterns, tea ommunication mus are diffi ructure. We oss function an be establ he cost, and nstances of licit nonlin is. The prob h are essent king under m ained some f wards the de CURITY CLASSIFIC fied (Include Area Code,	Coy) 15. PAGE 215 215 215 215 215 215 215 215 215 215	ASTIC COUNT Content oblems, information s and contr lve, mainly tified ussian tructing where linea gies are ed in this ng blocks l informati results for of a genera
SIMULTANEOUS DESIGN SYSTEMS 2. PERSONAL AUTHOR(S) BANSAL, RAJESH C. 13a. TYPE OF REPORT Technical 16. SUPPLEMENTARY NOTATION 17. COSATI CODES FIELD GROUP SUB 19. ABSTRACT (Continue on reverse We consider the pr policies in decentraliz because of the nonclass classes of such problem statistics for which th approach used consists joint strategies that a strategies fail to prov obtained that demonstra thesis can be viewed as for a general theory of and possibly partial st two-stage stochastic te 20. DISTRIBUTION / AVAILABILITY OF DUNCLASSIFIED/UNLIMITED 22a. NAME OF RESPONSIBLE INDIVID D FORM 1473, 84 MAR	OF COMMUN 13b. TIME CC FROM 	18. SUBJECT TERMS (Stochastic coni nonclassical in theory, zero-su and identify by block / simultaneously astic systems. ure of their in inear dynamics, lity of linear finding a lower is lower bound. ally optimal so aferiority of 1 nt prototype pro- age distributed l description. have made cont PT. OTIC USERS	GIES AND CON 14. DATE OF REPC 1988 AUG Continue on reven trol, linear nformation p um games. number) designing c Such proble formation st quadratic l strategies c r bound on t For some i lutions, exp inear design oblems, whic decision ma We have obt ributions to 21. ABSTRACT SE Unclassi 22b. TELEPHONE	TROL POLICIE ORT (Year, Month, I UST ommunication atterns, tea ommunication ms are diffi ructure. We oss function an be establ he cost, and nstances of licit nonlin is. The prob h are essent king under m ained some f wards the de CURITY CLASSIFIC fied Marcue Area Code, SECURITY O	Day) 15. PAGE 215 215 215 215 215 215 215 215 215 215	ASTIC COUNT Content oblems, information s and contribution information tructing where linear gies are ed in this ng blocks l information results for of a general YMBOL

UNCLASSIFIED SECURITY CLASSIFICATION OF THIS PAGE

theory for multistage (finite and infinite horizon) stochastic control and team problems with nonclassical information, in which the control (decision) variable affects not only the state trajectory but also the quality of information that is available to the decision makers.



Access	sion For	-
NTIS	GRA&I	
DTIC '	TAB	
Unanu	ounced	
Justi	riestion	
Distr Avai	ibution/ lability	Çul 9 9
	AV-12 5	istor .
Dist	Spente	.1
RV		

	CLASSIFIED	
SECURITY	CLASSIFICATION OF THIS PAGE	

: ?

Ŕ

141

ار با

* 84° . 84° . 84 . 94° . 98 . 48 . 48

SIMULTANEOUS DESIGN OF COMMUNICATION STRATEGIES AND CONTROL POLICIES IN STOCHASTIC SYSTEMS

BY

RAJESH C. BANSAL

B.Tech., Indian Institute of Technology, 1984 M.S., University of Illinois, 1986

THESIS

Submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Electrical Engineering in the Graduate College of the University of Illinois at Urbana-Champaign, 1988

Urbana, Illinois

SIMULTANEOUS DESIGN OF COMMUNICATION STRATEGIES AND CONTROL POLICIES IN STOCHASTIC SYSTEMS

Rajesh C. Bansal, Ph.D. Department of Electrical and Computer Engineering University of Illinois at Urbana-Champaign, 1988 Tamer Basar, Advisor

8

R

Ň

ğ

SN N

 $\hat{\lambda}$

88

We consider the problem of simultaneously designing communication strategies and control policies in decentralized stochastic systems. Such problems are difficult to solve, mainly because of the nonclassical nature of their information structure. We have identified classes of such problems with linear dynamics, quadratic loss functionals and Gaussian statistics for which the optimality of linear strategies can be established. The general approach used consists of first finding a lower bound on the cost, and then constructing joint strategies that attain this lower bound. For some instances of the cases where linear strategies fail to provide globally optimal solutions, explicit nonlinear strategies are obtained that demonstrate the inferiority of linear designs. The problems studied in this thesis can be viewed as important prototype problems, which are essential building blocks for a general theory of multistage distributed decision making under nonclassical information, and possibly partial statistical description. We have obtained some fundamental results for two-stage stochastic teams, and have made contributions towards the development of a general theory for multistage (finite and infinite horizon) stochastic control and team problems with nonclassical information, in which the control (decision) variable affects not only the state trajectory but also the quality of information that is available to the decision makers.

and the second second

i۳

ACKNOWLEDGEMENTS

I would like to express my gratitude and appreciation to Professor Tamer Basar for guidance and support throughout my graduate program, and to Mrs. Dixie Murphy for her excellent processing of this manuscript.

TABLE OF CONTENTS

CHAPTER

Ē

NX NX

8

See.

8

82 82

88

Ş.

Š,

Z.

XX m

PAGE

1. INTRODUCTION	1
 1.1. Motivation	1 5 9 11
2. THE DECENTRALIZED TWO-PERSON TEAM	14
 2.1. Introduction 2.2. Problem Formulation 2.3. Instances with Optimal Linear Solutions 2.4. Nonoptimality of Linear Strategies 2.5. Some Comments on Witsenhausen's Counterexample 2.6. Conclusion 	14 14 17 23 27 29
 3. SIMULTANEOUS COMMUNICATION AND CONTROL: FIRST-ORDER ARMA MODELS WITH FEEDBACK 3.1. Introduction 3.2. Problem Formulation 3.3. Construction of an Equivalent Problem 3.4. An Auxiliary Problem 3.5. Solutions to Problems P3 and P1 3.6 The Soft Constraint Version 	30 30 30 32 36 47
3.7. The Infinite Horizon Problem 3.8. Conclusion 4. SIMULTANEOUS COMMUNICATION AND CONTROL: GENERAL	58 74
ARMA MODELS WITH FEEDBACK	75
 4.1. Introduction	75 75 78 85 87 93 97

3
8
88
\$ 3
3 33
3 33
.
222
ŝ
555
2
<i>366</i>
76

5. THE DECENTRALIZED TWO-PERSON TEAM WITH MULTIPLE	
INFORMATION CHANNELS	
5.1 Introduction	00
5.2. Problem Formulation	
5.3. Instances with Optimal Linear Solutions	
5.4. Non-optimality of Linear Strategies x	
5.5. Conclusion	
6. DECENTRALIZED MULTIPERSON TEAMS	
6.1. Totanduction	
6.2 Dechlor Formulation	
6.2. The Problem with a Single Encoder and Multiple Decoder	
6.4. The Problem with Multiple Encoders and Multiple Decoders	
6.5. The Problem with Multiple Encoders and a Single Decoder	
6.6. Conclusion	
7. PROBLEMS WITH INCOMPLETE CHANNEL DESCRIPTION	
7.1. Introduction	134
7.2. Problem Description	135
7.3. Saddle Point Solutions for Games G_{12} , G_{13} and G_{14}	141
7.4. Analysis of Games G_{32} , G_{33} and G_{34}	150
7.5. Minimax and Maximin Strategies for Games G_{22} , G_{23} and G_{24}	
7.6. Conclusion	181
8 RECAPITULATION AND CONCLUSIONS	192
0. RECALLOLATION AND CONCLUSIONS	
APPENDIX A. REFORMULATION OF THE TEAM PROBLEM	
APPENDIX B. THE GAUSSIAN CHANNEL WITH SIDE INFORMATION	V
AT THE DECODER	195
KEFEKENCES	205
VITA	200
· · · · · ·	209

CHAPTER 1

INTRODUCTION

1.1. Motivation

巖

ß

8

33

3

8

Ř

8

8

ž

8

89

In many engineering situations a variety of possible measurements can be carried out on a physical system or process, and the question that naturally arises is what measurement process is optimal. Thus it may be necessary to communicate information through noisy channels to an observer, and a decision on what information to transmit may need to be made. Based on the observations received, the controller is required to make decisions that may affect the cost incurred as well as the further evolution of the system.

Consider, for example, the following decentralized systems:

(a) Control of a Space Probe from an Earth Station: We may think of a space probe and an earth station together constituting a decentralized system. The sensor aboard the space probe is required to transmit data over the downlink, after suitable encoding at some permissible power level. The earth station may be required to command the space probe to move along a certain trajectory, or it may wish to form estimates of the transmitted measurement. The problem, then, is one of simultaneously designing the communication strategy used by the probe and the control policy of the earth station. 222223

(b) *Tracking Systems:* In tracking systems it may be possible to allow for transmitting measurements at larger power levels (with an increase in the associated transmission cost) in order to combat noise and increase control efficiency. Here not only the communication strategy but also the transmission power level need to be designed.

Problems requiring simultaneous communication and control also arise in many socioeconomic situations, control of flexible structures, and other areas where the design

or allocation of limited transmission resources is important. The essentials of the situations involve

(i) A stochastic system, usually dynamic and/or decentralized, consisting of a number of decision makers (synonymously agents) along with a mathematical description of their interaction with the system and among themselves.

(ii) Elements of uncertainty (noises) entering the system, the underlying probability spaces for which are beyond the control of the decision makers.

(iii) An information structure, which characterizes the information gained and recalled by each decision maker.

(iv) A set of possible alternatives (decisions) for each agent, and permissible strategies which are mappings from the information space to the decision space of each agent.

(v) An objective functional that summarizes the preference ordering among various alternatives for the set of decision makers.

We note here that for the class of problems described above there exist very general formulations where, for example, the order in which the agents act may be determined by a chance mechanism (Witsenhausen [1971]) or the various agents exhibit a divergence of interests requiring a different objective functional for each agent (Basar and Olsder [1982]). Here we restrict our attention to classes of problems where in which

(i) The order in which the agents act is fixed in advance.

• 122223

(ii) All noises which may enter the system are zero mean Gaussian, and further

(iii) The problem involves the design of two kinds of strategies:

- Communication strategies for generating information-bearing signals.

- Control policies for forming estimates, minimizing errors and reducing costs.

Many such problems have been the subject of previous investigations. Athans [1972] considers the problem of selecting, at each instant of time, one measurement provided by one out of many sensors, with each measurement having an associated cost. For the class of linear stochastic dynamic systems and measurement subsystems, a weighted combination of prediction cost and accumulated observation cost is minimized, with the optimal measurement strategy being obtained by solving a deterministic optimal control problem. A similar formulation was considered earlier by Meier et al. [1967] who obtained a computational solution using dynamic programming. Herring and Melsa [1974] generalize the above results to allow the selection, at each instant of time, of the best combination of measurement devices as opposed to the best single device, and Mehra [1976] presents measurement schedules and sensor designs for linear stochastic systems subject to a constraint on the total measurement precision so as to minimize a norm of the error covariance matrix.

In contrast to the above studies, where the cost is the covariance of the error of the Kalman estimate. Chu [1978] considers the problem of finding the best measurement for static problems with arbitrary quadratic cost. A similar framework to that of Chu is adopted by Papavassilopoulos [1983] for the case of more than one decision maker, with the measurements still being restricted to be linear in the state.

Lafortune [1985] presents a general theorem for the computation of optimal solution to discrete time stochastic control problems when the decision makers have the additional freedom of choosing at each step among different sets of observations. Lafortune applies his results to finite state, controlled Markov chains as well as linear Gaussian systems with quadratic cost functionals. The control of Markov chains is also considered by

X

Ś

9

X

Ŕ

8

8

S

R

X

÷,

ŶŶ

ýX X

Varaiya and Walrand [1983], who obtain a complete solution for a class of symmetric channels with noiseless feedback.

Whittle and Rudge [1976] consider a situation where one agent observes a stationary ergodic Gaussian process and on its basis forms a sequence, through not necessarily a linear transformation. The other agent observes a noisy replica of the first agent's decision, and he is required to form an estimate of the original Gaussian sequence. This problem of simultaneous communication and control is solved using the concepts of statistical communication theory, and the solution requires that arbitrarily large signal blocks be available at all stages. In control theoretic terms their solutions are unrealizable, since the actions taken at a given time depend also on observations which lie in the future.

The problem considered by Whittle and Rudge is somewhat different from the ones mentioned earlier, because in the earlier formulations the measurement model is specified except for certain parameters, and the optimal selection of these parameters has been investigated. Here, we shall formulate and solve a class of problems related to that of Whittle and Rudge under the additional restriction of causality. When the design of the measurement strategy itself is part of the problem, the problem becomes considerably more difficult, since the action of one decision maker affects the information of the other, and there is no way in which this other decision maker can have access to the information on which the first one acted. Such information structures have been called *nonclassical* in the stochastic team literature. In the next section we study different information structures and the related computational considerations.

1.2. Information Structures and Computational Considerations

In this section we discuss the various types of information structures associated with stochastic team problems. These information structures characterize the precise static or dynamic information gained or recalled by each decision maker at each stage of a decision process, and are customarily distinguished as classical, quasiclassical or nonclassical. We then discuss some of the issues of computational complexity which arise when the numer-ical derivation of the optimal team solution is considered.

1.2.1. Classical and quasiclassical information structures

Classical information patterns include deterministic patterns and centralized information patterns. Deterministic patterns arise when the information is not noise-corrupted and may be of the *open-loop* type in which only the initial value of the state is available and no dynamic information is acquired, or of the *closed-loop* type where perfect information concerning the current value of the state is also acquired. Centralized patterns arise when all agents exchange their measurements without any delay and also recall the past information.

Under the deterministic or centralized stochastic information patterns, stochastic team problems become equivalent to stochastic control problems and the solution techniques for these (e.g., Bertsekas [1987], Kumar and Varaiya [1986]) are directly applicable. Thus for stochastic teams with classical information patterns, when

(i) the measurements are linear in the primitive random variables and controls,

(ii) the primitive random variables are jointly Gaussian.

(iii) the cost function is quadratic in the control vector and the primitive random variables, and

1.2.2. Nonclassical information structures

An information pattern is *nonclassical* if it is not partially nested. Alternatively, if the decision maker j's action affects the information of i, and there is no way by which i can infer the information available to j, then the information structure is said to be of the nonclassical type. Under the nonclassical information pattern, the derivation of the optimal team solution meets with formidable difficulties. One way of viewing these difficulties is that the control plays a triple role (Ho [1980]), viz.,

(i) the deterministic control effort of reducing the error;

(ii) improvement of future knowledge of uncertainty;

(iii) signalling to agents acting in the future some useful information that they do not necessarily acquire;

and these roles are in general conflicting.

Classes of tractable problems with nonclassical information have been very difficult to identify. Witsenhausen [1968] established that some of the simplest linear-quadratic-Gaussian (LQG) stochastic teams with nonclassical information do not admit optimal linear solutions; in fact the optimal solution to the problem formulated by Witsenhausen is not yet known. Very recently, however, some success has been reported for two-person stochastic teams with nonclassical information. Bansal and Basar [1987a] show that it is the presence of the product term between the decision variables which, coupled with the nonclassical information, makes the LQG problem intractable, and in the absence of this product term optimal solutions may readily be found. Some nonclassical multipath systems have also been shown to admit optimal linear solutions under quadratic costs and Gaussian noises (Bansal and Basar [1987b]).

Ě 8 24 1 8 8 <u>र</u> भू ĝ 8 R 2 -Ĵ, Ś 3 3 27

8

Here we are concerned with nonclassical information patterns because many problems involving the simultaneous design of communication and control strategies exhibit an information pattern of this type. Recall that the third role of controlling agents in nonclassical teams mentioned above is the role of "signalling" information to agents acting in the future, and this role is absent in the case of classical and quasiclassical information structures. When explicit design of the communication strategy itself is considered, it is the "signalling" policy that is being designed, and nonclassical patterns arise naturally.

It is for the above reasons that the problem considered by Whittle and Rudge [1976], where the communication strategy is to be designed, is significantly more complicated than the other problems, where the measurement model is specified, except for certain parameters, as in Athans [1972], Herring and Melsa [1974], Mehra [1976].

In the course of this work we shall identify some classes of stochastic teams with nonclassical information which are tractable in spite of the above mentioned difficulties.

1.2.3. Computational complexity of stochastic team problems

The severe difficulty encountered when the solution of stochastic team problems is attempted via numerical techniques has triggered some research into their complexity. Some recent results (Papadimitriou and Tsitsiklis [1982], Tsitsiklis and Athans [1985]) relate the complexity of stochastic team problems to that of known intractable problems. Using the tools of computational complexity the NP completeness of the discrete version of the static team problem has been established, and some progress has also been made in defining complexity concepts for continuous time problems by relating the complexity of the continuous version to that of its discretized counterpart (Papadimitriou and Tsitsiklis [1986]). In the same work the NP completeness of the discrete version of Witsenhausen's problem has also been established, thus explaining the failures, reported in literature, to attack it computationally (Ho and Chang [1980]).

Quite recently, Witsenhausen [1988] has developed a notion of equivalence between stochastic control problems that is suitable for complexity analysis. He has shown that a large class of problems with dynamic information can be reduced to equivalent static problems with a transformed cost functional, and this class includes all sequential discrete variable problems. Person-by-person optimality is then a sufficient condition for global optimality under smoothness and convexity conditions; however, when there is no convexity then person-by-person optimality is not sufficient.

While approaches based on the theory of NP completeness shed important light on the applicability (or nonapplicability) of numerical techniques for discretized versions of stochastic teams, being a worst case scenario they obscure important special cases which may admit relatively simple optimal solutions. For example, both the static team problem as well as Witsenhausen's counterexample have discretized versions which are NP complete, but the former is quite tractable under linear quadratic Gaussian assumptions (Radner [1962]), whereas the latter is still unsolved. The problem of carving out classes of tractable problems within the general class of nonclassical stochastic teams is an important one, and this thesis is a contribution to this direction.

1.3. Nonclassical Patterns and the Problem of Information Transmission

Consider the simplest problem of transmitting information, where the value of a random variable is to be transmitted, after suitable encoding, over a noisy channel. The noisy channel output is available at a decoder where it is used to construct an estimate of the input variable. In order to make this problem well defined, it is customary in the

9

の数

)

3

8

Ŕ

y.

Ń

Ň

S

communications literature to assume that the encoder operates under a hard power constraint, i.e., the mean-square level at the output of the encoder cannot exceed a certain predefined level.

This standard information transmission problem is depicted in Figure 1.1, where x is the input to be transmitted, u is the encoder output satisfying $E[u^2] \leq P^2$, w is the channel noise, and the output of the channel

$$y = u + w$$

is used by the decoder to form an estimate of x, designated by \hat{x} .

Note that the communication problem is one with a nonclassical information structure, since the action of the encoder affects the information of the decoder, but the decoder does not have access to the information of the encoder. (Indeed, if the decoder did have access to the information of the encoder, then the problem would be trivial, the estimate of x exact, and the channel redundant.)



Figure 1.1. The information transmission problem.

Information Theory provides the fundamental results for the analysis of the information transmission system of Figure 1.1. One of the most important results of Information Theory relates channel capacity to the rate distortion of the source. (A study of

channel capacity and rate distortion may be found, for example, in Ash [1965], Berger [1971] and Gallager [1968].)

Let R(D) denote the rate distortion function, (Berger [1971]), corresponding to a source under an appropriate distortion measure D. Shannon [1959] proved that it is not possible with any coding scheme to transmit the source under consideration through any channel of capacity less than R(D) without incurring an average distortion larger than or equal to D. Conversely, given any channel of capacity C > R(D), coding schemes exist (possibly using arbitrarily large block lengths) which result in an average distortion arbitrarily close to D when used over this channel. Thus, R(D) is the minimum channel capacity required to reproduce the given source at the decoder with average distortion at most D. For any channel, the minimum possible distortion D* that can result from its use is related to its capacity C by

$$R(D^*) = C.$$

We shall find the above relation between capacity and rate distortion quite useful for the analysis of some decentralized stochastic team problems, because the information transmission problem is a special case of problems with nonclassical information. We will also need the concepts of entropy and conditional entropy (see, for example, Abramson [1963]); these will be introduced in subsequent chapters as required.

1.4. Organization of the Thesis

Đ

Ø

3

X

8

3

<u>6</u>9

8

8

ŝ

8

Å,

2

Ņ

The organization of this thesis is as follows. In Chapter 2 we formulate and analyze some fundamental classes of stochastic team problems with two decentralized agents, exhibiting a nonclassical information pattern. We identify instances of such problems for which the optimal solutions are linear and may readily be found despite the intractability

7 k, Ċ, N.

of the general problem.

In Chapter 3 we consider a stochastic dynamic decision problem, where at each step two consecutive decisions must be made, one being what information-bearing signal to transmit, and the other what control action to exert. Such problems arise in the simultaneous optimization of both the observation and the control sequences in stochastic systems. This problem is solved completely for first-order auto-regressive moving average (ARMA) systems under the quadratic cost criterion. The results are further extended to cases where the time horizon is infinite and the cost function is discounted.

In Chapter 4 we consider stochastic dynamic decision problems requiring simultaneous optimization of both the observation and the control sequences for second- and higher-order systems, under quadratic cost criteria, and find strategies which are optimal over the affine class.

In Chapter 5 we generalize the results on the decentralized two-person teams of Chapter 2, by allowing the action of one agent to be transmitted to the other agent through a number of noisy channels simultaneously, instead of being transmitted through a single noisy channel. In Chapter 6 the results of Chapter 2 are extended in another direction, viz., to the case of more than two decision makers.

In Chapter 7 we expand on the framework by allowing an incomplete statistical description of the channel used to transmit measurements between the decentralized agents, and seek optimal solutions under a worst case analysis. It is assumed that the unknown part of the channel noise is controlled by an adversary or "jammer," and the situation is viewed as a zero sum game. The problems are studied for a variety of fidelity criteria, under hard and soft power constraints, for the transmitter as well as the jammer.

13

Finally, Chapter 8 provides a recapitulation of the results obtained, and concludes the thesis.

CHAPTER 2

THE DECENTRALIZED TWO-PERSON TEAM

2.1. Introduction

In this chapter we formulate and analyze some fundamental classes of stochastic team problems involving two decentralized agents, where the action of one agent affects the information of the other, and the other agent does not have access to the information upon which the first one acted. In Section 2.2 we formulate the general two-person stochastic team problem with quadratic costs and Gaussian noises. In Section 2.3 we identify those instances of the general problem for which the optimal solutions are linear. In Section 2.4 we show that for some instances in which we cannot show that linear solutions are globally optimal, the optimal linear policy may be outperformed by an appropriate nonlinear strategy. In Section 2.5 we comment on some aspects of the difficulties associated with Witsenhausen's counterexample, (Witsenhausen [1968]), and the concluding remarks in Section 2.6 end this chapter.

2.2. Problem Formulation

In this section, we formulate a two-person, decentralized, stochastic team problem with nonclassical information, where the action of one agent affects the information of the other and the information structure is nonnested (Basar and Cruz [1982]). The problem is to design the controls u_0 and u_1 so as to minimize the quadratic cost $J(\gamma_0, \gamma_1)$, where

$$J(\gamma_0,\gamma_1) = E[k_0 u_0^2 + s_0 u_0 x + s u_1^2 + s_1 u_1 x],$$

with

and

200

ŝ

8

8

est a

8

20

Š.

 $\hat{\boldsymbol{\Sigma}}$

ŝ

Ŷ,

 $\sum_{i=1}^{n}$

Ň

 $u_1 = \gamma_1(y),$

 $k_0 > 0$, s > 0, s₀ and s₁ being prespecified constants.

The variables z and y, upon which the two agents base their respective actions, are given by

$$z = ax + v \tag{2.1}$$

and

$$y = u_0 + bx + w$$
. (2.2)

Here x, v and w are zero mean Gaussian random variables, which are independent of one another, and have variances σ_x^2 , σ_v^2 and σ_w^2 , respectively.

The situation is depicted schematically in Figure 2.1 below.



Figure 2.1. The two-person, decentralized, stochastic team.

There are no general analytical or numerical tools which can be used to obtain the optimal solution of this problem, the difficulty arising mainly due to its nonclassical information structure. The separation principle (Wonham [1968]), for example, does not apply here,

 $u_0 = \gamma_0(z)$

 $\overline{\mathcal{X}}$

since the estimation part cannot be separated from the control action. Numerical techniques fail to provide any answers because the discretized version is NP complete (Tsitsiklis and Athans [1985]). To further understand the nature of the underlying difficulty, we may view the above *team* problem as an equivalent *control* (one-person) problem, with the control exhibiting a dual role. We complete the squares to obtain the cost functional

$$E[k_0 u_0^2 + s_0 u_0 x + s(u_1 + \frac{s_1}{2s} x)^2]$$

which implies that the optimal $\boldsymbol{\gamma}_1$ is necessarily given by

$$u_1 = \gamma_1(y) = -\frac{s_1}{2s} E(x | y)$$
 (2.3)

and hence we have the equivalent problem:

Minimize
$$E[k_0(u_0 + \frac{s_0}{2k_0}x)^2 - \frac{s_1^2}{4s}E(E(x+y)^2)]$$

subject to (2.2).

We thus have a quadratic term to be minimized as is usual in stochastic control formulations; in addition, there is a second *nonquadratic* term which is influenced by the information about x that y contains. In the absence of this second term the minimum is attained by choosing u_0 as a linear function of z:

$$u_{0} = \gamma_{0}(z) = -\frac{s_{0}}{2k_{0}} E(x | z) = -\frac{s_{0}}{2k_{0}} \left(\frac{a\sigma_{x}^{2}}{(a^{2}\sigma_{x}^{2} + \sigma_{x}^{2})} \right) z.$$
(2.4)

but the presence of this second term brings in the possibility of a conflict between the two roles of control and information transmission. Thus, even if affine laws are optimal,

indirect techniques need to be used to prove their optimality; but in fact, as to be elucidated below, affine strategies do not always continue to be optimal.

2.3. Instances with Optimal Linear Solutions

In this section we show that if either b = 0 or $\sigma_v^2 = 0$, then the stochastic team problem formulated in the preceding section admits optimal solutions which are linear in the observation variables.

(a) $\sigma_v^2 = 0$, b arbitrary

If $\sigma_v^2 = 0$, then we have a = 1 without loss of generality, and we define $u'_0 = u_0 + bx$ to obtain the equivalent Problem P1 below:

$$\begin{array}{c} \text{Minimize J}'(\gamma'_{0},\gamma'_{1}) \\ \gamma'_{0},\gamma'_{1} \end{array}$$

where

$$J'(\gamma'_{0},\gamma'_{1}) = E[k_{0}u'_{0}^{2} + s'_{0}u'_{0}x + \frac{s_{1}^{2}}{4s}(u'_{1} - x)^{2} + K]$$

$$u'_{0} = \gamma'_{0}(x),$$
 (2.5)

$$u'_{1} = \gamma'_{1}(u'_{0} + w),$$
 (2.6)

$$\mathbf{s}_0' \equiv \mathbf{s}_0 - 2\mathbf{k}_0 \mathbf{b} \,. \tag{2.7}$$

and K is a constant independent of u'_0 and u'_1 .

We thus obtain a problem of the type studied in Bansal and Basar [1987a], without a product term between the decision variables, for which linear strategies have been shown to be optimal. We briefly outline the approach here for completeness. Note that the situation is as depicted in Figure 2.2. In view of the discussion in Section 1.3, in which prob-

lems with nonclassical information have been related to problems involving information transmission, the agent taking action u_0 may be viewed as a generalized encoder and the agent taking action u_1 as a decoder, in the terminology of the information transmission system introduced in Chapter 1.



Figure 2.2. The transformed Problem P1.

We first consider the transformed problem under the additional restriction

$$E[u'_0^2] \leqslant P^2$$
. (2.8)

8

83

غتم

с, Ш

Х К

ß

3

3

X

3

×.

<u>N</u>

3

Ę,

5

S.

~ 77

Note that

$$I(x;y) \ge I(x;u'_{1}) \ge \frac{1}{2} \log \frac{\sigma_{x}^{2}}{E[(u'_{1}-x)^{2}]}$$
(2.9)

where I(a;b) denotes the mutual information of random variables a and b. The first inequality in (2.9) is the data processing inequality, and the second inequality follows from the definition of mutual information (see, for example, Wyner [1970]).

Further, we have

$$I(x;y) = H(y) - H(y|x)$$
(2.10)

19

where H(y) is the entropy of the random variable y and H(y|x) is the conditional entropy of y given x.

Using (2.10) along with the fact that for a given variance the entropy is maximized by a Gaussian random variable with that variance, (see, for example, Kagan, Linnik and Rao [1973]), we have

$$I(x;y) \leq \frac{1}{2} \log(\frac{P^2 + \sigma_w^2}{\sigma_w^2}).$$
 (2.11)

Using (2.11) along with (2.9) we obtain (under the restriction (2.8))

$$E[(u'_{1} - x)^{2}] \ge \frac{\sigma_{x}^{2} \sigma_{w}^{2}}{p^{2} + \sigma_{w}^{2}}.$$
(2.12)

Now let J'_{p} denote the minimum of $J'(\gamma'_{0},\gamma_{1})$ under the hard power constraint, i.e.,

$$J'_{p} \stackrel{\Delta}{=} \inf_{\substack{u'_{0}, u'_{1}, E[u'_{0}^{2}] = P^{2}}} J'(\gamma'_{0}, \gamma'_{1}).$$
(2.13)

We then have

$$\begin{split} \mathbf{I'_{p}} &\geq k_{0}P^{2} + \inf_{\substack{E[u'_{0}] = P^{2}}} E[s'_{0}u'_{0}x] + \inf_{\substack{E[u'_{0}^{2}] = P^{2}}} E[(u'_{1} - x)^{2}] \frac{s_{1}^{2}}{4s} + K \\ &\geq k_{0}P^{2} + \inf_{\substack{E[u'_{0}] \leq P^{2}}} E[s'_{0}u'_{0}x] + \inf_{\substack{E[u'_{0}^{2}] \leq P^{2}}} E[(u'_{1} - x)^{2}] \frac{s_{1}^{2}}{4s} + K \\ &= k_{0}P^{2} - |s'_{0}| P\sigma_{x} + \frac{\sigma_{x}^{2}\sigma_{w}^{2}}{(P^{2} + \sigma_{w}^{2})} \frac{s_{1}^{2}}{4s} + K \\ &\geq \min[k_{0}P^{2} - |s'_{0}| P\sigma_{x} + \frac{\sigma_{x}^{2}\sigma_{w}^{2}}{(P^{2} + \sigma_{w}^{2})} \frac{s_{1}^{2}}{4s} + K \\ &= k_{0}P^{*2} - |s'_{0}| P^{*}\sigma_{x} + \frac{\sigma_{x}^{2}\sigma_{w}^{2}}{(P^{2} + \sigma_{w}^{2})} \frac{s_{1}^{2}}{4s} + K \\ &= k_{0}P^{*2} - |s'_{0}| P^{*}\sigma_{x} + \frac{\sigma_{x}^{2}\sigma_{w}^{2}}{(P^{2} + \sigma_{w}^{2})} \frac{s_{1}^{2}}{4s} + K \\ &= k_{0}P^{*2} - |s'_{0}| P^{*}\sigma_{x} + \frac{\sigma_{x}^{2}\sigma_{w}^{2}}{(P^{2} + \sigma_{w}^{2})} \frac{s_{1}^{2}}{4s} + K \\ &= k_{0}P^{*2} - |s'_{0}| P^{*}\sigma_{x} + \frac{\sigma_{x}^{2}\sigma_{w}^{2}}{(P^{2} + \sigma_{w}^{2})} \frac{s_{1}^{2}}{4s} + K \\ &= k_{0}P^{*2} - |s'_{0}| P^{*}\sigma_{x} + \frac{\sigma_{x}^{2}\sigma_{w}^{2}}{(P^{2} + \sigma_{w}^{2})} \frac{s_{1}^{2}}{4s} + K \\ &= k_{0}P^{*2} - |s'_{0}| P^{*}\sigma_{x} + \frac{\sigma_{x}^{2}\sigma_{w}^{2}}{(P^{2} + \sigma_{w}^{2})} \frac{s_{1}^{2}}{4s} + K \\ &= k_{0}P^{*2} - |s'_{0}| P^{*}\sigma_{x} + \frac{\sigma_{x}^{2}\sigma_{w}^{2}}{(P^{2} + \sigma_{w}^{2})} \frac{s_{1}^{2}}{4s} + K \\ &= k_{0}P^{*2} - |s'_{0}| P^{*}\sigma_{x} + \frac{\sigma_{x}^{2}\sigma_{w}^{2}}{(P^{2} + \sigma_{w}^{2})} \frac{s_{1}^{2}}{4s} + K \\ &= k_{0}P^{*2} - |s'_{0}| P^{*}\sigma_{x} + \frac{\sigma_{x}^{2}\sigma_{w}^{2}}{(P^{2} + \sigma_{w}^{2})} \frac{s_{1}^{2}}{4s} + K \\ &= k_{0}P^{*2} - |s'_{0}| P^{*}\sigma_{x} + \frac{\sigma_{x}^{2}\sigma_{w}^{2}}{(P^{2} + \sigma_{w}^{2})} \frac{s_{1}^{2}}{4s} + K \\ &= k_{0}P^{*2} - |s'_{0}| P^{*}\sigma_{x} + \frac{\sigma_{x}^{2}\sigma_{w}^{2}}{(P^{2} + \sigma_{w}^{2})} \frac{s_{1}^{2}}{4s} + K \\ &= k_{0}P^{*2} - |s'_{0}| P^{*}\sigma_{x} + \frac{\sigma_{x}^{2}\sigma_{w}^{2}}{(P^{2} + \sigma_{w}^{2})} \frac{s_{1}^{2}}{4s} + K \\ &= k_{0}P^{*2} - |s'_{0}| P^{*}\sigma_{x} + \frac{\sigma_{x}^{2}\sigma_{w}^{2}}{(P^{*} + \sigma_{w}^{2})} \frac{s_{1}^{2}}{4s} + K \\ &= k_{0}P^{*} - |s'_{0}| P^{*}\sigma_{w} + \frac{\sigma_{x}^{2}\sigma_{w}^{2}}{(P^{*} + \sigma_{w}^{2})} \frac{s_{1}^{2}}{4s} + K \\ &= k_{0}P^{*} - |s'_{0}| P^{*}\sigma_{w} + \frac{\sigma_{x}^{2}}{(P^{*} + \sigma_{w}^{2})} \frac{s_{1}^{2}}{4s} + K \\ &=$$

We thus obtain a lower bound on the optimal cost, and the final task is to note that this lower bound is tight and is achieved by using the linear policy:

$$u_{0}^{*} = -(\operatorname{sgn} s_{0}') \frac{P^{*}}{\sigma_{x}} x. \qquad (2.15)$$

We now have Theorem 2.1 below:

Theorem 2.1.

(i) The stochastic team problem of Section 2.2 with $\sigma_v^2 = 0$, admits an optimal solution which is linear in the observation variables, and is given by

$$u_0^* = \gamma_0^*(x) = p^*x$$
 (2.16)

$$u_{1}^{*} = \gamma_{1}^{*}(y) = -\frac{s_{1}}{2s} \left[\frac{(p^{*}+b)\sigma_{x}^{2}}{(p^{*}+b)^{2}\sigma_{x}^{2} + \sigma_{w}^{2}} \right] y$$
(2.17)

where

 $\mathbf{p^*} = \lambda^* - \mathbf{b}$

and λ^* is given by the solution to the following parameter optimization problem:

$$\lambda^{*} = \arg \min_{\lambda} [k_{0}\lambda^{2}\sigma_{x}^{2} + s_{0}'\lambda\sigma_{x}^{2} + \frac{s_{1}^{2}}{4s} \frac{\sigma_{x}^{2}\sigma_{w}^{2}}{(\lambda^{2}\sigma_{x}^{2} + \sigma_{w}^{2})}].$$
(2.18)

(ii) The optimal value of the cost is

$$k_{0}\lambda^{*2}\sigma_{x}^{2} + s_{0}'\lambda^{*}\sigma_{x}^{2} + \frac{s_{1}^{2}}{4s} \frac{\sigma_{x}^{2}\sigma_{w}^{2}}{(\lambda^{*2}\sigma_{x}^{2} + \sigma_{w}^{2})} + K$$
(2.19)

where

$$K = \sigma_x^2 (-k_0 b^2 + s_0 b - s_1^2/4s) .$$

Remark 2.1. The parameter optimization problem (2.18) always admits a solution since as $\lambda \rightarrow \infty$, so does the expression to be minimized, implying that we can restrict the search to a compact set over which a continuous function always admits a minimum.

(b) b = 0, σ_y^2 arbitrary

8

8

ß

88

8

33

X

R

20

1

If b = 0, then the channel noise becomes independent of the input x, and the problem is a special case of the multipath system considered in Bansal and Basar [1987b] for which linear strategies have been shown to be optimal. For completeness we outline the approach here. We can assume, without any loss of generality, that a = 1, then introduce the random variable

$$\mathbf{m} = \mathbf{E}(\mathbf{x} \mid \mathbf{z}) \tag{2.20}$$

and make the following observations:

(i)
$$\gamma_1(y) = -\frac{s_1}{2s} E(x | y)$$

(ii) $E[E(x | y) - x)^2] = E[(E(x | y) - m)^2] + E[(m - x)^2]$
(iii) $E[u_0(z)x] = E[u_0(z)m]$

(since the random variable (x-E(x|z)) is independent of z).

(iv) The random variable m is Gaussian distributed, since

$$m = \frac{\sigma_x^2}{\sigma_x^2 + \sigma_y^2} z,$$

and has the variance

$$\sigma_{\rm m}^{2} = \sigma_{\rm x}^{4} / (\sigma_{\rm x}^{2} + \sigma_{\rm y}^{2}) \,.$$

RT.

In view of the above observations, the problem with b = 0 is equivalent to

$$\begin{array}{c} \text{Minimize } J''(\gamma_0,\gamma_1) \\ \gamma_0',\gamma_1 \end{array}$$

 $J''(u'_{0},u_{1}) = E[k_{0}u'_{0}^{2} + s_{0}u'_{0}m + \frac{s_{1}^{2}}{4s}(u_{1} - m)^{2} + K']$

where

with

$$u'_{0} = \gamma'_{0}(m)$$
$$u_{1} = \gamma_{1}(y)$$

and K' a constant independent of u'_0 and u_1 . Since m is zero-mean Gaussian, we have a problem of the type discussed in part (a) of this section, and the optimality of linear strategies follows.

Theorem 2.2.

(i) The stochastic team problem of Section 2.2 with b = 0, a = 1, admits an optimal solution which is linear in the observation variables, and is given by

$$u_{0}^{*}(z) = \lambda^{*}z$$

$$u_{1}^{*}(y) = -\frac{s_{1}}{2s} \left[\frac{\lambda^{*}\sigma_{x}^{2}}{\lambda^{*}(\sigma_{x}^{2} + \sigma_{y}^{2}) + \sigma_{w}^{2}} \right] y$$

where λ^* is given by the solution to the following parameter optimization problem:

$$\lambda^{*} = \arg \min_{\lambda} \left[k_{0} \lambda^{2} (\sigma_{x}^{2} + \sigma_{y}^{2}) + s_{0} \lambda \sigma_{x}^{2} + \frac{s_{1}^{2}}{4s} \left[\frac{\sigma_{x}^{2} (\lambda^{2} \sigma_{y}^{2} + \sigma_{w}^{2})}{\lambda^{2} (\sigma_{x}^{2} + \sigma_{y}^{2}) + \sigma_{w}^{2}} \right] \right]$$

(a solution to which may always be found as discussed in Remark (2.1)).

 $\gamma_1 = \gamma_1(y)$

(ii) The optimal value of the cost is

$$k_{0}\lambda^{*2}(\sigma_{x}^{2}+\sigma_{v}^{2})+s_{0}\lambda^{*}\sigma_{x}^{2}+\frac{s_{1}^{2}}{4s}\frac{\sigma_{x}^{2}(\lambda^{*2}\sigma_{v}^{2}+\sigma_{w}^{2})}{(\lambda^{*2}(\sigma_{x}^{2}+\sigma_{v}^{2})+\sigma_{w}^{2})}-\frac{s_{1}^{2}}{4s}\sigma_{x}^{2}.$$

2.4. Nonoptimality of Linear Strategies

In the preceding section we have identified those instances of the general two-person, decentralized, stochastic team problem for which the optimal strategies are linear and may be obtained through the solution of a related parameter optimization problem. These include the cases where either $\sigma_v^2 = 0$, i.e., an uncorrupted version of the variable x is available to the agent acting first; or b = 0, i.e., the channel noise is independent of the input x. In this section we show that when neither of the above two conditions hold, then it is possible to construct some problem instances where the optimal linear strategy is outperformed by an appropriately chosen nonlinear strategy. In the following, we first assume that b = 1, $\sigma_w^2 = 0$. Note that if these assumptions imply that linear strategies are not optimal, then we cannot expect linear strategies to be optimal, in general, for the class of two-person decentralized teams with nonzero b and σ_v^2 .

Problem PG

Minimize
$$E[k_0u_0^2 + s_0u_0x + su_1^2 + s_1u_1x]$$

 γ_0, γ_1

where

$$u_0 = \gamma_0(z)$$
, $u_1 = \gamma_1(y)$

Note that, since z = ax + v, x may be written as $x = \beta z + n$ where

$$\beta = a\sigma_x^2/(a^2\sigma_x^2 + \sigma_y^2)$$

and n is a Gaussian random variable which is independent of z, with variance

$$\sigma_n^2 = \sigma_x^2 \sigma_v^2 / (a^2 \sigma_x^2 + \sigma_v^2) .$$

Thus we obtain the equivalent problem

Minimize
$$E[k_0u_0^2 + \beta s_0u_0z + su_1^2 + s_1\beta u_1z + s_1u_1n]$$

 γ_0, γ_1

where

$$u_0 = \gamma_0(z)$$
, $u_1 = \gamma_1(u_0 + \beta z + n)$.

(In this restatement we have used the fact that $E[u_0n] = 0$, n being independent of z, and having zero-mean.) We next define $u_0 + \beta z = u'_0$ to obtain the cost functional

$$E[k_0(u'_0 - \beta z)^2 + \beta s_0(u'_0 - \beta z)z + su_1^2 + s_1\beta u_1 z + s_1u_1n]$$

and arrive at the problem below:

Minimize
$$E[k_0 u'_0^2 + (\beta s_0 - 2k_0\beta)u'_0z + su_1^2 + s_1u_1n + s_1\beta u_1z]$$

 $\gamma'_0\gamma_1$

where

$$u'_{0} = \gamma'_{0}(z)$$
, $u_{1} = \gamma_{1}(u'_{0} + n) \equiv \gamma_{1}(y)$.

We thus have the situation depicted in Figure 2.3 below.



Figure 2.3. Schematics for the transformed problem.

Note the presence of the term $E[u_1n]$ in the cost functional; in the absence of this term the

5

523

8

20

2

8

بر برک

optimality of linear strategies may be established via information theoretic bounds. We now have a problem which is similar to the Gaussian test channel problem (Gallager [1968]), with the difference that instead of having to estimate the input z, the decoder is required to estimate a linear combination of the input and the channel noise. We thus have the problem below:

Minimize
$$E[k_0u'_0^2 + s'_0u'_0z + s(u_1 + s_{11}n + s_{12}z)^2]$$

 $\gamma'_0.\gamma_1$

where

$$u'_{0} = \gamma'_{0}(z) , u_{1} = \gamma_{1}(u'_{0} + n)$$

$$s'_{0} \equiv \beta(s_{0} - 2k_{0})$$

$$s_{11} \equiv s_{1}/2s$$

$$s_{12} \equiv s_{1}\beta/2s .$$

We now note that since $n = y - u'_0$, we have

$$u_{1} + s_{11}n + s_{12}z$$

= $u_{1} + s_{11}(y - u'_{0}) + s_{12}z$
= $u_{1} + s_{11}y + s_{12}z - s_{11}u'_{0}$.

Defining

 $\mathbf{u'}_1 = \mathbf{u}_1 + \mathbf{s}_{11}\mathbf{y}$

we obtain the problem

$$\underset{\gamma'_{0},\gamma'_{1}}{\text{Minimize } E[k_{0}u'_{0}^{2} + s'_{0}u'_{0}z + s(u'_{1} + s_{12}z - s_{11}u'_{0})^{2}]}$$

where

$$u'_0 = \gamma'_0(z), \ u'_1 = \gamma'_1(y).$$

This is a problem of the type studied in Bansal and Basar [1987a] with a product term between the decision variables, and hence, as shown there, regions of the parameter space

exist where nonlinear strategies outperform the optimum linear strategies.

An Illustration

Consider the Problem PG with parameter values $k_0 = 0.01$, $s_0 = 6.02$, s = 1.0, $s_1 = 2.0$, $\sigma_x^2 = 1.0025$, a = -0.5/(1.0025) and $\sigma_y^2 = 100-(0.5)^2/1.0025$.

With the above choice we may write $x = \beta z + n$ where $\beta = -0.005$ and $\sigma_n^2 = 1.0$. (Recall that we have taken $\sigma_w^2 = 0$, b = 1 and $\sigma_z^2 = 100.0$).

We then obtain the transformed problem

Minimize
$$E[0.01u'_0^2 - 0.03u'_0x + (u'_1 - u'_0 - 0.005z)^2]$$

 $y'_0y'_1$

where

Contraction Official and Contraction

$$u'_0 = \gamma'_0(z)$$

$$u'_1 = \gamma'_1(u'_0 + n) \equiv \gamma'_1(y).$$

Now, if γ'_0 is linear, then optimal γ'_1 is also linear, and hence restricting γ'_0 to the form $\gamma'_0 = \lambda z$, and optimizing over λ , yields a cost of -1.2477 which is the best in the linear class. However, we can show that the nonlinear policy

$$\gamma'_{0}(z) = 10 \text{ sgn } z$$
$$\gamma'_{1}(y) = \begin{cases} 10 & \text{if } y \ge 0\\ -10 & \text{if } y < 0 \end{cases}$$

yields a cost of -1.3911 which is superior to the optimal cost in the linear class.

2.5. Some Comments on Witsenhausen's Counterexample

The following problem WI, studied by Witsenhausen [1968], is a two-person, decentralized, stochastic team problem with a product term between the decision variables, and may be viewed as a special case of the reduced version found in Section 2.4.

Problem WI

Minimize $E[k_0(u_0 - x)^2 + (u_0 - u_1)^2]$ $\gamma_0 \gamma_1$ résseres Orazza con Conferenci Orazza se Converses

where

$$u_0 = \gamma_0(x); u_1 = \gamma_1(y)$$

and

 $\mathbf{y} = \mathbf{u}_0 + \mathbf{w} \ .$

Noting that

$$(u_0 - u_1)^2 = (y - w - u_1)^2$$

and defining

 $u'_1 = y - u_1$,

we obtain the equivalent Problem WI' below.

Problem WI'

Minimize
$$E[k_0(u_0 - x)^2 + (u_1' - w)^2]$$

 γ_0, γ_1'

with

$$u_0 = \gamma_0(x); u'_1 = \gamma'_1(u_0 + w).$$

We, therefore, have a problem in which the second agent wishes to estimate the channel noise (which is independent of u_0), in sharp contrast to the communication type problems where the second agent wishes to estimate the observation of the first agent.

Since the second agent necessarily uses $u'_1 = E(w | y)$, the second term in the cost function for Problem WI' becomes

$$J_2(\gamma_0) = E[(w - E(w | y))^2].$$

It is a known result in probability and statistics that for every fixed power level $E[u_0^2] = P^2$, the linear strategy

$$u_0 = \gamma_0(x) = \frac{P}{\sigma_x} x$$

maximizes $J_2(\gamma_0)$, when $y = u_0 + w$. To prove this result, consider a zero sum game in normal form with kernel

$$G(u_0, u_1) = E[(w - u_1)^2]$$

which is to be maximized by a choice of $u_0 = \gamma_0(x)$ and minimized by a choice of $u_1 = \gamma_1(y)$, subject to the constraint $E[u_0^2] = P^2$. Note that

$$\max_{u_0} J_2(u_0) = \max_{u_0} \min_{u_1} G(u_0, u_1),$$

i.e., the unrestricted maximum of the function J_2 is equal to the lower value of the game with kernel G. But the upper and lower value of a game are equal if a saddle point exists, and to complete the proof one can show that

$$u_0 \sim N(0, P^2)$$
$$u_1 = \frac{\sigma_w^2}{P^2 + \sigma_w^2} y$$

provide a saddle-point solution for the game above.

In view of the above result, we see that for a given power level the linear choice of strategy γ_0 is the worst possible design as far as the minimization of the second term in the cost functional for Problem WI' is concerned. It is therefore not surprising that linear solutions are not optimal for the classes of problems which involve the estimation of channel noise as in the case of Witsenhausen's counterexample, as well as in the general case of Section 2.4.

2.6. Conclusion

In this chapter we have formulated and analyzed some fundamental classes of stochastic team problems, involving two decentralized agents, where the action of one agent affects the information of the other, and the information structure is nonnested. We have identified those instances of the general problem for which the optimal solutions are linear. For some instances in which we cannot show that linear strategies are globally optimal, we have shown that the optimal linear policies may be outperformed by appropriately chosen nonlinear strategies. We have commented on some aspects of the difficulty associated with Witsenhausen's problem. The important conclusion that can be drawn from the analysis in this chapter is that for two-person decentralized stochastic team problems there do exist classes of parameter values for which the optimal solutions are linear and may readily be found, despite the fact that the general problem is quite intractable.
CHAPTER 3

SIMULTANEOUS COMMUNICATION AND CONTROL: FIRST-ORDER ARMA MODELS WITH FEEDBACK

3.1. Introduction

In this chapter we consider stochastic dynamic teams where at each step two consecutive decisions must be taken, one being what information-bearing signal to transmit and the other regarding what control action to exert. The organization of this chapter is as follows. In Section 3.2 we shall formally pose the problem with hard power constraints as Problem P1. In Section 3.3 we shall use an intermediate Problem P2 to construct Problem P3. Problems P1 and P3 will be shown to be equivalent in the sense that the optimal solution of one may be constructed from the optimal solution of the other. In Section 3.4 we formulate and solve an auxiliary problem using some results from Information Theory. In Section 3.5 we shall show that the solution to the auxiliary problem may be used to provide a solution to Problem P3, thereby solving Problem P1. In Section 3.6 we study the associated "soft" constraint version, and in Section 3.7 we study the existence of stationary optimal policies for the infinite horizon problem with discounted cost. The concluding remarks in Section 3.8 then end this chapter.

3.2. Problem Formulation

In this section we give a precise formulation of the problem to be solved in the sequel.

The problem is to control a stochastic dynamic system, the description of which is available in the form of a general first-order difference equation, with the objective being to minimize a quadratic cost functional.

The control v_i is based on the observation vector $(y_0,...,y_i)$, where each of the y_k 's is a noise-corrupted version of the information-bearing signal u_i , the u_i 's being based on the current value of the state and all previous y_k 's. In the context of the space probe example of Section 1.1, (with the u_i 's being generated on board the space probe and the v_i 's being generated on the earth station), such a situation arises quite naturally because a feedback line (the uplink) is available which is many times more reliable than the forward link (the downlink).

The problem clearly has a nonclassical information structure, since the agent taking action v_i does not have access to the information based on which action u_i is taken. A precise formulation of this nonclassical stochastic team problem is provided next.

We specify the stochastic system by the following set of equations:

$$\mathbf{x}_{i+1} = \rho_i \mathbf{x}_i + \mathbf{m}_i - \mathbf{v}_i \tag{3.1a}$$

$$\mathbf{y}_i = \mathbf{u}_i + \mathbf{w}_i \tag{3.1b}$$

along with

$$E[u_i^2] \leq P_i^2 \tag{3.1c}$$

$$u_i = h_i(x_i, y^{i-1})$$
 (3.1d)

and

$$\mathbf{v}_{i} = \boldsymbol{\gamma}_{i}(\mathbf{y}^{i}) \quad . \tag{3.1e}$$

Here (3.1a) and (3.1b) are the state and measurement equations, respectively. The subscript $i_i(i=0,1,...)$ denotes that the realization of the random variable is at the i-th time instant, whereas the superscript denotes the history of the random variable up to that time instant, i.e.,

$$\mathbf{y}' = (\mathbf{y}_0, \ldots, \mathbf{y}_1) ,$$

which we are also going to interpret as a row vector. The random variables x_0 , w_i and m_i ($i \ge 0$) are all assumed to be independent and Gaussian with mean zero and variance $\sigma_{(\cdot)}^2$, the subscript being the identifier. The functions h_i and γ_i , i=0,1,..., are the control policies, each one Borel-measurable in its arguments, and leading to second-order random variables u_i and v_i , respectively, the former also satisfying the power constraint (3.1c).

The criterion for comparing different control policies h_i 's and γ_i 's is based on the cost function we wish to optimize, which in this case is taken to be

$$J(h^{N}, \gamma^{N}) = E\left[\sum_{i=0}^{N} (a_{i+1}x_{i+1}^{2} + b_{i}v_{i}^{2})\right]$$
(3.2)

where $a_{i+1} > 0$ and $b_i > 0$ for all i=0,...,N, and the action variables u_i and v_i are related to the policy variables h_i and γ_i via (3.1d)-(3.1e). We thus have Problem P1 below.

Problem P1:

$$\begin{array}{ll} \text{Minimize} & J(h^N, \gamma^N) \\ & & h^N, \gamma^N \end{array}$$

subject to (3.1a) through (3.1e), where $J(h^N, \gamma^N)$ is defined by (3.2).

3.3. Construction of an Equivalent Problem

The gist of this section is as follows: First an equivalent Problem P2 is constructed from P1 which differs only in the form of the cost function. The cost function for P2 is in the form of a sum of the squared differences between state and control variables. In the transformation from P1 to P2 the constraints represented by (3.1a) through (3.1e) are unaltered. In a follow-up step Problem P3 is constructed from Problem P2 such that the

33

structure of the cost function is left unaltered, while the state equations are redefined so as to facilitate subsequent analysis.

These two transformations are presented below as Claims 3.1 and 3.2, respectively.

Claim 3.1: Under the set of constraints represented by (3.1a) through (3.1e), the cost function for Problem P1, defined by (3.2), is identical to

$$J''(n^{N}, \gamma^{N}) = E[\sum_{i=0}^{N} a'_{i}(v_{i}-b'_{i}x_{i})^{2}] + c_{N}$$
(3.3)

where

$$b'_{i} \triangleq k_{i+1} \rho_{i} / (b_{i} + k_{i+1})$$
 (3.4a)

$$\mathbf{a}'_{i} \triangleq \mathbf{b}_{i} + \mathbf{k}_{i+1} \tag{3.4b}$$

$$c_{N} \stackrel{\Delta}{=} k_{o} \sigma_{x_{o}}^{2} + \sum_{i=0}^{N} k_{i+1} \sigma_{m_{i}}^{2}$$
 (3.4c)

and $\{k_i\}$ is a sequence defined recursively by

$$k_{i} = a_{i} + k_{i+1} b_{i} \rho_{i}^{2} / (b_{i} + k_{i+1})$$

$$k_{N+1} = a_{N+1}$$
(3.5)

Proof: This is a standard result in stochastic LQ control with perfect state information (see, e.g., Bertsekas [1987] or Kumar an I Varaiya [1986]), known also as "completing the squares." Note that c_N is a constant (independent of the control sequence $\{v_j\}$), and (3.5) is the so-called discrete-time Riccati equation for this scalar problem.

Claim 3.2: The solution to Problem P2 may be obtained by solving the following equivalent Problem P3.

Minimize h^N, y^N

$$J(h^{N},\gamma^{N})$$

$$\tilde{\mathbf{x}}_{i+1} = \rho_i \tilde{\mathbf{x}}_i + \mathbf{m}_i \tag{3.6a}$$

$$\tilde{\mathbf{v}}_{i} = \boldsymbol{\gamma}_{i}(\mathbf{y}^{\prime}) \tag{3.6b}$$

$$\mathbf{y}_i = \mathbf{u}_i + \mathbf{w}_i \tag{3.6c}$$

$$u_i = h_i(\tilde{x}_i, y^{i-1})$$
 (3.6d)

$$E[u_i^2] \leq P_i^2$$

with

and the u_i 's satisfy

Problem P3:

where

$$J(h^{N},\gamma^{N}) = E[\sum_{i=0}^{N} a'_{i}(\tilde{v}_{i}-b'_{i}\tilde{x}_{i})^{2}] + c_{N} . \qquad (3.7)$$

Proof: The situation is depicted in Figure 3.1. Substituting for
$$x_1$$
 using Equation (3.1a) we get

$$v_1 - b'_1 x_1 = v_1 + b'_1 v_0 - b'_1 (\rho_0 x_0 + m_0)$$
(3.8)

Similarly,

$$\mathbf{v}_{2} - \mathbf{b}'_{2} \mathbf{x}_{2} = \mathbf{v}_{2} + \mathbf{b}'_{2} \mathbf{v}_{1} + \mathbf{b}'_{2} \rho_{1} \mathbf{v}_{0} - \mathbf{b}'_{2} (\rho_{1} (\rho_{0} \mathbf{x}_{0} + \mathbf{m}_{0}) + \mathbf{m}_{1}),$$
(3.9)

$$\mathbf{v}_{i} - \mathbf{b}'_{i} \mathbf{x}_{i} = \mathbf{v}_{i} + \mathbf{b}'_{i} \mathbf{v}_{i-1} + \mathbf{b}'_{i} \rho_{i-1} \mathbf{v}_{i-2} + \dots + \mathbf{b}'_{i} \rho_{i-1} \cdots \rho_{1} \mathbf{v}_{o} - \mathbf{b}'_{i} (\rho_{i-1} (\rho_{i-2} (\dots (\rho_{o} \mathbf{x}_{o} + \mathbf{m}_{o}) \dots) + \mathbf{m}_{i-2}) + \mathbf{m}_{i-1}) .$$
(3.10)

We now define

$$\tilde{x}_{o} = x_{o}$$

 $\tilde{x}_{i+1} = \rho_{i}\tilde{x}_{i} + m_{i} , \quad i=0,1,...,$
(3.11)

and

Ľ.

1223

uu Uu

88

Z

к Х

R

8

2

 $\mathbf{S}_{\mathbf{1}}$

ľ.

3

3

2,

ξġ

۰۴ جرب

(3.6e)

8

d 14



Figure 3.1. Diagrammatic representation of Problem P3.

$$\tilde{\mathbf{v}}_{i} = \mathbf{v}_{i} + b'_{i} \mathbf{v}_{i-1} + b'_{i} \rho_{i-1} \mathbf{v}_{i-2} + \dots + b'_{i} \rho_{i-1} \cdots \rho_{1} \mathbf{v}_{o} \quad .$$
(3.12)

Using these new variables, the cost function (3.3) can be rewritten as

$$J(h^{N}, \tilde{\gamma}^{N}) = E[\sum_{i=0}^{N} a'_{i}(\tilde{v}_{i} - b'_{i}\tilde{x}_{i})^{2}] + c_{N} ,$$

$$\tilde{v}_{i} = \tilde{\gamma}_{i}(y^{i}, v^{i-1}) ,$$

$$(3.13)$$

where the evolution of the \tilde{x}_i 's is determined by (3.11). Since this is a team problem, and for each fixed h^N the resulting stochastic control problem has classical information, minimization of (3.13) over $(h^N, \tilde{\gamma}^N)$ is equivalent to its minimization over (h^N, γ^N) where γ_i has only y^i as its argument.

The relationship between $\mathbf{\tilde{v}}_i$'s and \mathbf{v}_i 's is given by

$$\tilde{\mathbf{v}}^{\mathbf{N}} = \mathbf{v}^{\mathbf{N}}\mathbf{B}^{\mathbf{T}} \tag{3.14}$$

where \mathbf{B}^{T} is the transpose of the following nonsingular lower triangular matrix:

$$\mathbf{B} = \begin{pmatrix} 1 & 0 & 0 & 0 \cdot 0 & 0 \\ b'_1 & 1 & 0 & 0 \cdot 0 & 0 \\ b'_2 \rho_1 & b'_2 & 1 & 0 \cdot 0 & 0 \\ b'_3 \rho_2 \rho_1 & b'_3 \rho_2 & b'_3 & 1 \cdot 0 & 0 \\ \vdots & & & 1 & 0 \\ \vdots & & & & 1 & 0 \\ b'_N \rho_{N-1} .. \rho_1 & b'_N \rho_{N-1} .. \rho_2 & \cdot & \cdot & b'_N & 1 \end{pmatrix}.$$

We may thus write

$$\mathbf{v}^{\mathsf{N}} = \tilde{\mathbf{v}}^{\mathsf{N}} [\mathbf{B}^{\mathsf{T}}]^{-1} ,$$

and we thus see that in order to obtain the optimal solution to P1, we may equivalently solve the problem in terms of \tilde{v}_i and \tilde{x}_i (over h^N, γ^N), which is precisely Problem P3. This then leads to the following equivalence between the solutions of Problems P1 and P3:

Lemma 3.1.

(i) Problem P1 admits a solution if, and only if, Problem P3 does.

(ii) If (h^{N}, γ^{N}) is a solution for P1, then $(h^{N}, \gamma^{N}B^{T})$ solves P3; conversely, if (h^{N}, γ^{N}) solves P3, then $(h^{N}, \gamma^{N}[B^{T}]^{-1})$ is a solution for P1.

X

8

8

Ŷ

8

<u>.</u> X

Š S

Ņ

ļ,

335 (11) (33)

s.

3

3.4. An Auxiliary Problem

In this section we formulate and solve an auxiliary problem which will play an important role in the solution to Problem P3.

Consider the situation depicted in Figure 3.2, the problem being one of finding the signals u_0, \ldots, u_N subject to the power constraints

$$\mathrm{E[u_i^2]} \leqslant \mathrm{P_i^2}$$
 ,

so as to minimize the mean-square error in the estimation of z_N using y_0, y_1, \ldots, y_N . It is

given that z_0 , n_i , w_i , i=0,1,..., are mutually independent Gaussian random variables, each with mean zero and variance indicated by $\sigma_{(\cdot)}^2$, the subscript being the identifier. Following the convention of Information Theory, we let $I(z_N;y^N)$ denote the mutual information between z_N and y^N , and call the supremum of this quantity the capacity of the corresponding system.

We shall first solve this problem for the case when N=1. The proof for arbitrary finite N will then be shown to follow a similar line of reasoning. The case with N=1 is illustrated in Figure 3.3.

Recall from the discussion in Section 1.3 that for any channel the minimum possible distortion D* that can result from its use is related to its capacity by

$$R(D^*) = C .$$

First consider the lower branch of Figure 3.3, redrawn as Figure 3.4. We find D* for this system. Let us suppose that the input is connected to a Gaussian memoryless source with variance $\sigma_{z_1}^2$, in which case we have

$$R(D) = Max(0, \frac{1}{2} \log(\frac{\sigma_{z_1}^2}{D})) . \qquad (3.15)$$

Let z_o^j denote the sequence (z_{o1}, \ldots, z_{oj}) , which we again interpret as a row vector; define n_o^j, v_1^j and z_1^j likewise. We introduce

$$m_{1}^{j} \triangleq E(z_{1}^{j} | z_{0}^{j}) = \frac{(1+\alpha_{1})\sigma_{z_{1}}^{2}}{(1+\alpha_{1})^{2}\sigma_{z_{1}}^{2} + \sigma_{n_{1}}^{2}} z_{0}^{j}$$

and observe the following.

.

)

X

8

X

252

8

いい

50) R.4

(a) The components of $m_1^{\ j}$ are i.i.d. Gaussian variates with zero mean and variance



Figure 3.2. Diagrammatic representation of the auxiliary problem.



Construction

Ģ

COCOL

Figure 3.3. The auxiliary problem with N=1.

38

1.0 2.8

2

8

ديس ريون

88

5

Х. Х.

8

3

\$2: ____

B

Ŋ

2

22 22 22

32

5

Constraint Proposition Procession

ŝ

6.55566

X

Q

3



Figure 3.4. The lower branch of the system depicted in Figure 3.3.

$$\frac{[(1+\alpha_1)\sigma_{z_1}^2]^2}{(1+\alpha_1)^2\sigma_{z_1}^2+\sigma_{n_1}^2}$$
(3.16)

(b)

$$E\{\frac{1}{j}||m_{1}^{j}-z_{1}^{j}||^{2}\} = \frac{\sigma_{z_{1}}^{2}\sigma_{n_{1}}^{2}}{(1+\alpha_{1})^{2}\sigma_{z_{1}}^{2}+\sigma_{n_{1}}^{2}},$$
(3.17)

where $\|\cdot\|$ denotes the Euclidean norm. We also have

$$\lim_{j \to \infty} \inf_{j} \frac{1}{j} \mathbb{E} \| \mathbf{v}_{1}^{j} - \mathbf{m}_{1}^{j} \|^{2} = \frac{(1 + \alpha_{1})^{2} \sigma_{z_{1}}^{4}}{(1 + \alpha_{1})^{2} \sigma_{z_{1}}^{2} + \sigma_{n_{1}}^{2}} e^{-2C_{o}}$$
(3.18)

where

$$C_{o} = \frac{1}{2} \log \left(\frac{P_{o}^{2} + \sigma_{w_{o}}^{2}}{\sigma_{w_{o}}^{2}} \right)$$
(3.19)

Now since

$$\frac{1}{j} E||z_1^{j} - v_1^{j}||^2 = \frac{1}{j} E||z_1^{j} - m_1^{j} + m_1^{j} - v_1^{j}||^2 = \frac{1}{j} E||z_1^{j} - m_1^{j}||^2 + \frac{1}{j} E||m_1^{j} - v_1^{j}||^2 .$$

we have

$$D^{*} = \lim_{j \to \infty} \frac{1}{j} E \|z_{1}^{j} - v_{1}^{j}\|^{2}$$

$$= \frac{\sigma_{z_{1}}^{2} \sigma_{n_{1}}^{2}}{(1 + \alpha_{1})^{2} \sigma_{z_{1}}^{2} + \sigma_{n_{1}}^{2}} + \frac{(1 + \alpha_{1})^{2} \sigma_{z_{1}}^{4}}{((1 + \alpha_{1})^{2} \sigma_{z_{1}}^{2} + \sigma_{n_{1}}^{2})} \frac{\sigma_{w_{o}}^{2}}{(P_{o}^{2} + \sigma_{w_{o}}^{2})}$$
(3.20)

 $R(D^*)$ may now be computed, and we have the following result:

Lemma 3.2: For the system depicted in Figure 3.4, the mutual information $I(z_1;y_0)$ is bounded above by

$$C_{eq} = \frac{1}{2} \log \left[\frac{(P_o^2 + \sigma_{w_o}^2)((1 + \alpha_1)^2 \sigma_{z_1}^2 + \sigma_{n_1}^2)}{\sigma_{n_1}^2 (P_o^2 + \sigma_{w_o}^2) + (1 + \alpha_1)^2 \sigma_{z_1}^2 \sigma_{w_o}^2} \right].$$
(3.21)

Now, consider the system depicted in Figure 3.3. We first note that

$$I(z_1; y_0, y_1) = I(z_1; y_0) + I(z_1; y_1 | y_0) .$$
(3.22)

Now,

$$I(z_{1};y_{1} | y_{0}) = I(z_{1},w;y_{1} | y_{0}) - I(w;y_{1} | z_{1},y_{0})$$
(3.23a)

$$\leq I(z_1, w; y_1 | y_0) \tag{3.23b}$$

$$= I(w;y_1 | y_o)$$
(3.23c)

$$= H(y_1 | y_0) - H(y_1 | w, y_0)$$
(3.23d)

$$= H(y_1 | y_0) - H(y_1 | w)$$
(3.23e)

$$\leq H(y_1) - H(y_1 | w)$$
(3.23f)

$$= I(y_1; w) \tag{3.23g}$$

$$\leq \frac{1}{2} \log(\frac{P_1^2 + \sigma_{w_1}^2}{\sigma_{w_1}^2})$$
 (3.23h)

where $H(\cdot)$ is the entropy and $H(\cdot | \cdot)$ the conditional entropy. Here steps (a), (d) and (g)

follow from the definition of mutual information, steps (c) and (e) are due to the Markov Property, step (b) follows because information is always positive, step (f) is valid because conditioning cannot increase entropy, and the last step holds because, for a fixed variance, the Gaussian random variable has the maximum entropy (Kagan et al. [1973]).

Using (3.21) and (3.23), along with (3.22), we obtain the next result.

Lemma 3.3: For the system depicted in Figure 3.3,

$$I(z_{1};y_{o},y_{1}) \leq \frac{1}{2} \log \left[\frac{(P_{o}^{2} + \sigma_{w_{o}}^{2})((1+\alpha_{1})^{2}\sigma_{z_{1}}^{2} + \sigma_{n_{1}}^{2})}{(\sigma_{n_{1}}^{2}(P_{o}^{2} + \sigma_{w_{o}}^{2}) + (1+\alpha_{1})^{2}\sigma_{z_{1}}^{2}\sigma_{w_{o}}^{2})} \cdot \frac{P_{1}^{2} + \sigma_{w_{1}}^{2}}{\sigma_{w_{1}}^{2}} \right].$$
(3.24)

Using this upper bound on $I(z_1, y_0, y_1)$ we can find a lower bound on the minimum mean-square error achievable when the problem is to estimate z_1 from the observations y_0 and y_1 .

We have '

$$I(z_{1}; y_{0}, y_{1}) \ge I(z_{1}; v_{1}) \ge \frac{1}{2} \log \frac{\sigma_{z_{1}}^{2}}{E[(z_{1} - v_{1})^{2}]}$$
(3.25)

which implies (using Equation (3.24))

$$\frac{1}{2} \log \frac{(P_o^2 + \sigma_{w_o}^2)((1 + \alpha_1)^2 \sigma_{z_1}^2 + \sigma_{n_1}^2)}{(\sigma_{n_1}^2 (P_o^2 + \sigma_{w_o}^2) + (1 + \alpha_1)^2 \sigma_{z_1}^2 \sigma_{w_o}^2)} \cdot \frac{P_1^2 + \sigma_{w_1}^2}{\sigma_{w_1}^2} \ge \frac{1}{2} \log \frac{\sigma_{z_1}^2}{E[(z_1 - v_1)^2]} \quad (3.26)$$

i.e.,

$$E[(z_1 - v_1)^2] \ge \frac{\sigma_{w_1}^2(\sigma_{n_1}^2(P_o^2 + \sigma_{w_o}^2) + (1 + \alpha_1)^2 \sigma_{z_1}^2 \sigma_{w_o}^2)}{(P_o^2 + \sigma_{w_o}^2)(P_1^2 + \sigma_{w_1}^2) \sigma_{z_o}^2}.$$
 (3.27)

Ŕ

We next note that if we use the policy

$$u_{o} = \lambda_{o} z_{o}$$

$$u_{1} = \lambda_{1} (z_{1} - E(z_{1} | y_{o}))$$
(3.28)

with λ_o and λ_i chosen so as to satisfy the power constraints, the minimum mean-square error is indeed achieved and we have

42

Lemma 3.4. The policies given by (3.28) are the policies which minimize the mean-square error in estimating z_1 from the pair (y_0, y_1) .

We now consider the case with arbitrary N, depicted in Figure 3.2. First, consider the problem in the absence of the most recent observation, i.e., the uppermost channel of Figure 3.2 removed. Assuming that the version of the problem with (N-1) channels is already solved, the capacity C_{N-1} (i.e., maximum mutual information between input and output) for the portion of the system within the rectangular box CDEF is known. We can therefore find the minimum achievable distortion for a memoryless Gaussian source with variance $\sigma_{z_{n}}^{2}$ when only the portion within the box ABCD is in use, by computing the conditional estimate of z_N given z_{N-1} , and transmitting this optimally. Using this minimum achievable distortion we can find an upper bound for $I(z_N;y^{N-1})$ as follows (where z_N^j is the sequence z_{N_1}, \ldots, z_{N_j})

$$D^{*} = \lim_{j \to \infty} \frac{1}{j} \frac{E \| z_{N}^{j} - v_{N}^{j} \|^{2}}{\sigma_{z_{N}}^{2} \sigma_{n_{N}}^{2}} + \frac{(1 + \alpha_{N})^{2} \sigma_{z_{N}}^{4}}{(1 + \alpha_{N})^{2} \sigma_{z_{N}}^{2} + \sigma_{n_{N}}^{2}} e^{-2C_{N-1}}$$
(3.29)

i.e.,

 $I(z_{N}; y^{N-1}) \leq \frac{1}{2} \log \left| \frac{(1+\alpha_{N})^{2} \sigma_{z_{N}}^{2} + \sigma_{n_{N}}^{2}}{\sigma_{n_{N}}^{2} + (1+\alpha_{N})^{2} \sigma_{z_{N}}^{2} e^{-2C_{N-1}}} \right|.$ (3.30)

We can use a series of inequalities as in (3.23) to show that

$$I(z_{N};y_{N}|y^{N-1}) \leq \frac{1}{2} \log \frac{P_{N}^{2} + \sigma_{w_{N}}^{2}}{\sigma_{w_{N}}^{2}}.$$
 (3.31)

Â

Now since

 $I(z_{N}; y^{N}) = I(z_{N}; y^{N-1}) + I(z_{N}; y_{N} | y^{N-1})$ (3.32)

we get

$$I(z_{N};y^{N}) \leq \frac{1}{2} \log \left[\frac{(1+\alpha_{N})^{2} \sigma_{z_{N}}^{2} + \sigma_{n_{N}}^{2}}{\sigma_{n_{N}}^{2} + (1+\alpha_{N})^{2} \sigma_{z_{N}}^{2} e^{-2C_{N-1}}} \cdot \frac{P_{N}^{2} + \sigma_{w_{N}}^{2}}{\sigma_{w_{N}}^{2}} \right]$$
(3.33)

and we have the following lemma.

Lemma 3.5. The mutual information $I(z_N; y^N)$ is bounded above by C_N , which is the last step of the recursion:

$$C_{o} = \frac{1}{2} \log(\frac{P_{o}^{2} + \sigma_{w_{o}}^{2}}{\sigma_{w_{o}}^{2}})$$
(3.34a)

 \mathcal{R}

 \overline{C}

Ň

R

Ň

3

3

44

$$C_{i} = \frac{1}{2} \log \left(\frac{(1+\alpha_{i})^{2} \sigma_{z_{1}}^{2} + \sigma_{n_{i}}^{2}}{\sigma_{n_{i}}^{2} + (1+\alpha_{i})^{2} \sigma_{z_{1}}^{2} e^{-2C_{i-1}}} \cdot \frac{P_{i}^{2} + \sigma_{w_{i}}^{2}}{\sigma_{w_{i}}^{2}} \right).$$
(3.34b)

Let Δ_i denote the minimum achievable mean-square error when z_i is estimated using y^i , i.e., (using (3.25))

$$\Delta_{i} \stackrel{\Delta}{=} \sigma_{z_{i}}^{2} e^{-2C_{i}} . \tag{3.35}$$

We therefore have

and for i=1,...,N

$$\Delta_{\rm o} = \frac{\sigma_{z_{\rm o}}^2 \sigma_{w_{\rm o}}^2}{P_{\rm o}^2 + \sigma_{w_{\rm o}}^2}$$
(3.36a)

and for i=1,...,N

$$\Delta_{i} = \frac{\sigma_{w_{i}}^{2}}{P_{i}^{2} + \sigma_{w_{i}}^{2}} \left[\frac{(1 + \alpha_{i})^{2} \sigma_{z_{i}}^{4}}{\sigma_{z_{i-1}}^{4}} \Delta_{i-1} + \frac{\sigma_{z_{i}}^{2}}{\sigma_{z_{i-1}}^{2}} \sigma_{n_{i}}^{2} \right].$$
(3.36b)

We shall next show that this lower bound is tight and may be achieved by using the policies

$$u_{o} = \lambda_{o} z_{o} \tag{3.37a}$$

and for i=1,...,N

$$u_i = \lambda_i (z_i - E(z_i | y^{i-1}))$$
 (3.37b)

Here the $\lambda_i{}^{\prime}s$ are chosen so as to meet the power constraints with equality. Since

$$z_{i-1} = (1+\alpha_i)z_i + n_i$$
 (3.38)

Occession (

HERE AND A

Noncession I

we may equivalently write

$$z_i = \tilde{\rho}_{i-1} z_{i-1} + m_{i-1} \tag{3.39}$$

where $\tilde{\rho}_{i-1}$ is a constant,

$$\tilde{\rho}_{i-1} \triangleq (1+\alpha_i) \frac{\sigma_{z_i}^2}{\sigma_{z_{i-1}}^2} , \qquad (3.40)$$

and m_{i-1} is a Gaussian random variable which is independent of z_{i-1} and has variance

$$\sigma_{m_{l-1}}^{2} = \frac{\sigma_{z_{l}}^{2}}{\sigma_{z_{l-1}}^{2}} \sigma_{n_{l}}^{2}.$$
 (3.41)

With the policies chosen as in (3.37), we have

$$y_{o} = \lambda_{o} z_{o} + w_{o} \tag{3.42a}$$

and for i=1,...,N

$$y_i = \lambda_i (z_i - E(z_i | y^{i-1})) + w_i$$
. (3.42b)

Let Σ_i denote the mean-square error in the estimation of z_i from y^i when the communication strategies are chosen as in (3.37), i.e.,

$$\Sigma_i = E[(z_i - E(z_i | y^i))^2]$$

We then have, using (3.36)

$$\lambda_o^2 = \frac{P_o^2}{\sigma_z^2} \tag{3.43a}$$

 $\lambda_{i}^{2} = \frac{P_{i}^{2}}{(1+\alpha_{i})^{2} \frac{\sigma_{z_{i}}^{4}}{\sigma_{z_{i-1}}^{4}} \sum_{j=1}^{2} \frac{\sigma_{z_{j}}^{2}}{\sigma_{z_{j-1}}^{2}} \sigma_{n_{i}}^{2}}$ (3.43b)

Further, by our specific choice of policy, $(z_i - E(z_i | y^{i-1}))$ is a zero mean Gaussian random variable, and therefore

$$E[(z_{i}-E(z_{i}|y^{i-1})-E[(z_{i}-E(z_{i}|y^{i-1}))|y_{i}])^{2}] = \frac{s_{i}^{2}\sigma_{w_{i}}^{2}}{P_{i}^{2}+\sigma_{w_{i}}^{2}}$$
(3.44)

where

$$s_{i}^{2} \triangleq E[(z_{i}-E(z_{i}|y^{i-1}))^{2}]$$

= $(1+\alpha_{i})^{2} \frac{\sigma_{z_{i}}^{4}}{\sigma_{z_{i-1}}^{4}} \sum_{i=1}^{2} \frac{\sigma_{z_{i}}^{2}}{\sigma_{z_{i-1}}^{2}} \sigma_{n_{i}}^{2}.$ (3.45)

Also, $E(E(z_i|y^{i-1})|y_i)=0$ by our choice of policy, since y_i is independent of y^{i-1} , and therefore the expression on the left-hand side of (3.44) becomes

$$E[(z_i - E(z_i | y^{i-1}) - E(z_i | y_i))^2]$$

=
$$E[(z_i - E(z_i | y^i))^2]$$

=
$$\Sigma_i$$

and we get the recursion (for i=1,...,N)

$$\Sigma_{i} = \frac{\sigma_{w_{i}}^{2}}{P_{i}^{2} + \sigma_{w_{i}}^{2}} \left[\frac{(1 + \alpha_{i})^{2} \sigma_{z_{i}}^{4}}{\sigma_{z_{i-1}}^{4}} \Sigma_{i-1} + \frac{\sigma_{z_{i}}^{2}}{\sigma_{z_{i-1}}^{2}} \sigma_{n_{i}}^{2} \right]$$
(3.46a)

with the initial condition

$$\Sigma_{o} = \frac{\sigma_{w_{o}}^{2} \sigma_{z_{o}}^{2}}{P_{o}^{2} + \sigma_{w_{o}}^{2}}.$$
 (3.46b)

The recursion for the Σ_i 's is therefore identical to the recursion for the Δ 's (given by (3.36)), which denote the minimum mean-square error achievable. This shows that the lower bound on error is indeed tight and thus leads to the following theorem.

Theorem 3.1.

8

8

8

223

ŝ,

Ň

3

. ,

R

Ň

(a) The policies given by (3.37) minimize the mean-square error incurred in estimating z_N from y^N for the system depicted in Figure 3.2, where the λ_i 's are defined by (3.43) using the Σ_i 's defined by (3.46).

(b) The minimum mean-square error is given by the last step of the recursion (3.46) or equivalently by the last step of the recursion (3.36).

3.5. Solutions to Problems P3 and P1

We now return to Problem P3 defined in Section 3.3, where the policies

$$\mathbf{u}_{i} = \mathbf{h}_{i}(\tilde{\mathbf{x}}_{i}, \mathbf{y}^{i-1})$$

and

$$\mathbf{v}_{i} = \boldsymbol{\gamma}_{i}(\mathbf{y}^{1})$$

are to be chosen in order to minimize

$$J = E[\sum_{i=0}^{N} a'_{i} (\tilde{v}_{i} - b'_{i} \tilde{x}_{i})^{2}]$$
(3.47)

under the constraints depicted in Figure 3.1.

We first consider the minimization of the N-th term in the expression for J, which is

$$E[a'_{N}(\tilde{v}_{N}-b'_{N}\tilde{x}_{N})^{2}],$$

the optimization problem being equivalent to minimizing

$$E[a'_{N}b'_{N}^{2}(\tilde{v}'_{N}-\tilde{x}_{N})^{2}],$$

where

$$\tilde{v'}_{N} \equiv \frac{\tilde{v}_{N}}{b'_{N}} ,$$

i.e., the problem is one of forming the best estimate of x_N under the mean-square distortion criterion. We now show that the situations depicted in Figures 3.1 and 3.2 are identical except for nomenclature. To show this equivalence we note that for i=1,...,N

$$z_{i-1} = (1 + \alpha_i) z_i + n_i$$
 (3.48)

which implies

$$z_{i} = (1+\alpha_{i}) \frac{\sigma_{z_{i}}^{2}}{\sigma_{z_{i+1}}^{2}} z_{i-1} + m_{i-1}$$
(3.49)

where the m_i's are zero mean Gaussian random variables each with variance

$$\sigma_{m_{i}}^{2} = \frac{\sigma_{z_{i+1}}^{2}}{\sigma_{z_{i}}^{2}} \sigma_{n_{i+1}}^{2}.$$
 (3.50)

We therefore have

$$\rho_{i-1} = (1+\alpha_i) \frac{\sigma_{z_i}^2}{\sigma_{z_{i-1}}^2} \text{ for } i=1,...,N$$

and by defining $\sigma_{\tilde{x}_o}^2 = \sigma_{z_o}^2$ we can complete the correspondence between the variables \tilde{x}_i 's and z_i 's for i=0,1,...,N.

The solution to the problem of minimizing the mean-square error in estimating \tilde{x}_N from y_0, \ldots, y_N may therefore be obtained as in Section 3.4. Using the techniques of Section 3.4 we get

(a) The minimum mean-square error in estimating x_i using y^i is given by Δ_i , where Δ_i 's satisfy the recursion (for i=1,...,N)

$$\Delta_{i} = \frac{\sigma_{w_{i}}^{2}}{P_{i}^{2} + \sigma_{w_{i}}^{2}} \left(\rho_{i-1}^{2} \Delta_{i-1} + \sigma_{m_{i-1}}^{2}\right)$$
(3.51a)

with the initial condition

8.31 V 1

8

22

Х Х

8

N:C

8

N. M

Ż

5

Υ.

5

5.7

$$\Delta_{\rm o} = \frac{\sigma_{\tilde{x}_{\rm o}}^2 \sigma_{\rm w_{\rm o}}^2}{P_{\rm o}^2 + \sigma_{\rm w_{\rm o}}^2} \,. \tag{3.51b}$$

(b) The optimal encoding strategies are

$$\mathbf{u}_{o} = \mathbf{h}^{*}_{o}(\tilde{\mathbf{x}}_{o}) = \lambda_{o}\tilde{\mathbf{x}}_{o}$$
(3.52a)

and for i=1,...,N

$$u_{i} = h_{i}^{*}(\tilde{x}^{i}, y^{i-1}) = \lambda_{i}(\tilde{x}_{i} - E(\tilde{x}_{i}|y^{i-1}))$$
(3.52b)

where the λ_i 's satisfy the recursion (for i=1,...,N)

$$\lambda_{i}^{2} = \frac{P_{i}^{2}}{\rho_{i-1}^{2} \Delta_{i-1} + \sigma_{m_{i-1}}^{2}}$$
(3.53a)

with the initial condition

$$\lambda_{o}^{2} = \frac{P_{o}^{2}}{\sigma_{\tilde{x}_{o}}^{2}}$$
(3.53b)

(the Δ_i 's being as defined by (3.51)).

(c) The optimal choice for the $\boldsymbol{\tilde{\gamma}}_i$'s is

$$\tilde{\mathbf{v}}_{i} = \mathbf{b}'_{i} \mathbf{E}(\tilde{\mathbf{x}}_{i} | \mathbf{y}^{i}) \tag{3.54}$$

where $E(\tilde{x}_i|y^i)$ satisfy the recursion (for i=1,...,N)

$$E(\tilde{x}_{i}|y^{i}) = \rho_{i-1}E(\tilde{x}_{i-1}|y^{i-1}) + \frac{P_{i}}{P_{i}^{2} + \sigma_{w_{i}}^{2}} (\rho_{i-1}^{2}\Delta_{i-1} + \sigma_{m_{i-1}}^{2})^{\frac{1}{2}}y_{i}$$
(3.55a)

with the initial condition

$$E(\tilde{x}_{o}|y^{o}) = \frac{P_{o}\sigma_{\tilde{x}_{o}}y_{o}}{P_{o}^{2} + \sigma_{\tilde{x}_{o}}^{2}}$$
(3.55b)

(the Δ_i 's being as defined in (3.51)).

We finally note that the policies which minimize the mean-square error in the estimation of \tilde{x}_i given y^i for i=0, N-1, are identical to the corresponding policies used in the estimation of \tilde{x}_N given y^N , and we therefore have the following theorem.

Theorem 3.2.

(a) The optimum policies h_i and $\hat{\gamma}_i$ for Problem P3 are given by (3.52) and (3.54), respectively, using the λ_i 's and Δ_i 's as defined by (3.53) and (3.51), respectively.

(b) The minimum value of the cost function for Problem P3 is

$$J^{*} = \sum_{i=0}^{N} a'_{i} {b'_{i}}^{2} \Delta_{i} + c_{N}$$

7

Now we turn to the original Problem P1 formulated in Section 3.2. Taking the difference

$$\frac{\tilde{\mathbf{v}}_{i}}{\mathbf{b}'_{i}} - \rho_{i-1} \frac{\tilde{\mathbf{v}}_{i-1}}{\mathbf{b}'_{i-1}}$$
(3.56)

and using (3.54), (3.55) and (3.12), we find that

$$\frac{\mathbf{v}_{i}}{\mathbf{b}'_{i}} = (\rho_{i-1} - \mathbf{b}'_{i-1}) \frac{\mathbf{v}_{i-1}}{\mathbf{b}'_{i-1}} + \frac{P_{i}}{P_{i}^{2} + \sigma_{\mathbf{w}_{i}}^{2}} (\rho_{i-1}^{2} \Delta_{i-1} + \sigma_{\mathbf{m}_{i-1}}^{2})^{\frac{1}{2}} \mathbf{y}_{i}$$
(3.57)

which implies that the optimal control policies for the original problem are

$$\mathbf{v}^{*}_{i} = \boldsymbol{\gamma}^{*}_{i}(\mathbf{y}^{i}) = \mathbf{b}'_{i} \mathbf{E}(\mathbf{x}_{i} | \mathbf{y}^{i}) \triangleq \mathbf{b}'_{i} \hat{\mathbf{x}}_{i}$$
(3.58)

where $\hat{x}_i \triangleq E(x_i | y^i)$ satisfy the recursion (for i=1,...,N)

$$\hat{\mathbf{x}}_{i} = (\rho_{i-1} - b'_{i-1})\hat{\mathbf{x}}_{i-1} + \frac{P_{i}}{P_{i}^{2} + \sigma_{\mathbf{w}_{i}}^{2}} (\rho_{i-1}^{2} \Delta_{i-1} + \sigma_{\mathbf{m}_{i-1}}^{2})^{\frac{1}{2}} \mathbf{y}_{i}$$
(3.59a)

with the initial condition

$$\hat{x}_{o} = \frac{P_{o}\sigma_{x_{o}}}{P_{o}^{2} + \sigma_{x_{o}}^{2}} y_{o}, \qquad (3.59b)$$

the Δ_i 's being as defined in (3.51).

We therefore have the following theorem.

Theorem 3.3.

(a) The optimum policies $\{h_i^*\}$ and $\{\gamma_i^*\}$ for Problem P1 are given by (3.52) and (3.58), respectively, using the λ_i 's and Δ_i 's as defined by (3.53) and (3.51), respectively.

(b) The minimum value of the cost function for Problem P1 is

$$J^* = \sum_{i=0}^{N} a'_{i} b'_{i}^2 \Delta_i + c_N$$

<u>3</u> 33

E E

ž

80

8

2

¢۲ با

23 28

S

, ,

С.

An Illustration

Consider the case with N=2, the stochastic system model being given as

$$x_1 = x_0 + m_0 - v_0$$

 $x_2 = x_1 + m_1 - v_1$
 $x_3 = x_2 + m_2 - v_2$

and the objective being to minimize

$$J(h^{2},\gamma^{2}) = \sum_{i=0}^{2} (x_{i+1}^{2} + v_{i}^{2}),$$

subject to the constraints of Equations (3.1b) through (3.1e), it being given that

$$\sigma_{\mathbf{x}_0}^2 = \sigma_{\mathbf{w}_1}^2 = \sigma_{\mathbf{m}_1}^2 = P_i^2 = 1.0$$
.

Using Claim 3.1, (Section 3.2), we get the equivalent cost functional

$$\sum_{i=0}^{2} a'_{i} (v_{i} - b'_{i} x_{i})^{2} + c_{2}$$

where

53

$$a'_0 = 13/5$$
, $a'_1 = 5/2$, $a'_2 = 2$,
 $b'_0 = 8/13$, $b'_1 = 3/5$, $b'_2 = 1/2$,

and

 $c_2 = 613/130$.

Using Theorem 3.3 above, we obtain the optimal strategies for the problem:

 $u_{0}^{*} = \lambda_{0} x_{0}$ $u_{1}^{*} = \lambda_{1} (x_{1} - E(x_{1} | y^{0}))$ $u_{2}^{*} = \lambda_{2} (x_{2} - E(x_{2} | y^{1}))$

where

$$\lambda_0^2 = 1$$
, $\lambda_1^2 = 2/3$ and $\lambda_2^2 = 4/7$

and

 $v_0^* = 8/3 \hat{x}_0$ $v_1^* = 3/5 \hat{x}_1$ $v_2^* = 1/2 \hat{x}_2$

where

$$\hat{x}_0 = 1/2 y_0$$

$$\hat{x}_1 = 5/26 y_0 + 3/34 y_1$$

$$\hat{x}_2 = 1/3 y_0 + 3/2 y_1 + 7/16 y_2.$$

$$E(x, |y^0|) = 7/26 y_0$$

and

$$E(x_2 | y') = 1/13 y_0 + 3/10 y_1$$
.

Further, the optimal cost is 6.1852.

3.6. The Soft Constraint Version

ي الاستان المالية المالية الم

We consider again the problem depicted in Figure 3.1, but now with the power constraints

54

$$E[u_i^2] \leq P_i^2$$

removed, i.e., the optimum power levels are also to be determined via the underlying optimization problem. Such a formulation is useful in situations where more "costly" measurements that contain more useful or reliable information may be used. It may be possible to transmit a larger power at the encoder (at additional cost) in order to further decrease the mean-square error at the decoder, this tradeoff being reflected by the cost criterion. Mathematically, we may represent this as a power constraint which is "implied" or "soft", appearing as an additional term in the cost functional, which now becomes

$$J(h^{N},\gamma^{N}) = E\left[\sum_{i=0}^{N} (a'_{i}(v_{i}-b'_{i}x_{i})^{2} + q_{i}u_{i}^{2})\right].$$
(3.60)

We shall obtain the solution to the soft constraint version by using the solution to the hard constraint Problem P1 found in Section 3.5. Let J_p denote the infimum of J under the hard power constraints,

$$J_{p} \stackrel{\Delta}{=} \inf_{\substack{h^{N}: \gamma^{N}: E[h_{1}^{2}] = P_{1}^{2}, \hat{n}^{p}}} J(h^{N}, \gamma^{N})$$
(3.61)

We then have the following series of equalities and inequalities:

$$J_{P} = \sum_{i=0}^{N} q_{i}P_{i}^{2} + \inf_{E[h_{i}^{2}]=P_{i}^{2}} E\left[\sum_{i=0}^{N} (a'_{i}(v_{i}-b'_{i}x_{i})^{2})\right]$$

$$\geq \sum_{i=0}^{N} q_{i}P_{i}^{2} + \inf_{E[h_{i}^{2}] \leq P_{i}^{2}} E\left[\sum_{i=0}^{N} (a'_{i}(v_{i}-b'_{i}x_{i})^{2})\right]$$

$$= \sum_{i=0}^{N} q_{i}P_{i}^{2} + \sum_{i=0}^{N} a'_{i}b'_{i}^{2}\Delta_{i}$$

$$\geq \min_{P_{i}^{2} \geq 0} \left[\sum_{i=0}^{N} q_{i}P_{i}^{2} + a'_{i}b'_{i}^{2}\Delta_{i}\right]$$

$$= \sum_{i=0}^{N} q_{i}P_{i}^{*}^{2} + a'b'_{i}^{2}\Delta_{i}$$
(3.62)

Crock Field Charles and Charles and

where the Δ_i 's are defined in (3.51) and Δ^* 's are defined recursively likewise, with P_i replaced by P^*_i , i.e., for i=1,...,N

$$\Delta^{*}_{i} = \frac{\sigma_{w_{1}}^{2}}{P^{*}_{i}^{2} + \sigma_{w_{1}}^{2}} \left(\rho_{i-1}^{2} \Delta^{*}_{i-1} + \sigma_{m_{i-1}}^{2}\right)$$
(3.63a)

with the initial condition

Š

28

8

X

Ű.

Ň

8

Ģ

У. У.

82

8

Ś

Ż

$$\Delta *_{o} = \frac{\sigma_{w_{o}}^{2} \sigma_{x_{o}}^{2}}{P *_{o}^{2} + \sigma_{w_{o}}^{2}}.$$
 (3.63b)

In order to find the optimal power levels $(P_{i}^{*2}s)$, we can solve the following problem

$$\underset{P_{o},\ldots,P_{N}}{\text{Min}}\sum_{i=0}^{N} q_{i}P_{i}^{2} + a'_{i}b'_{i}^{2}\Delta_{i}$$
(3.64)

which is a nonlinear optimal control problem, the solution to which is given by the following dynamic program (with $\rho_{-1} \triangleq 1$ and $\sigma_{m_{-1}}^2 \triangleq 0$), where $W_i(\Delta)$ is the "optimum cost to go" given that the system is at state Δ at stage i

$$W_{i}(\Delta) = M_{in} \left[q_{i} P_{i}^{2} + a'_{i} b'_{i}^{2} \frac{\sigma_{w_{i}}^{2}}{P_{i}^{2} + \sigma_{w_{i}}^{2}} (\rho_{i-1}^{2} \Delta + \sigma_{m_{i-1}}^{2}) + W_{i+1} (\frac{\sigma_{w_{i}}^{2}}{P_{i}^{2} + \sigma_{w_{i}}^{2}} (\rho_{i-1}^{2} \Delta + \sigma_{m_{i-1}}^{2})) \right]$$
(3.65)

and the optimal value of the cost is

$$\underset{P_{0}^{2},\ldots,P_{N}^{2}}{\text{Min}} J(P_{0}^{2},\ldots,P_{N}^{2}) = W_{0}(\sigma_{x_{0}}^{2}).$$
(3.66)

We next show that a solution to the above problem always exists. If we define

$$f(\Delta, P_i^2) \triangleq q_i P_i^2 + \frac{a'_i b'_i^2 \sigma_{w_i}^2}{P_i^2 + \sigma_{w_i}^2} \left(\rho_{i-1}^2 \Delta + \sigma_{m_{i-1}}^2\right) + W_{i+1} \left(\frac{\sigma_{w_i}^2 (\rho_{i-1}^2 \Delta + \sigma_{m_{i-1}}^2)}{P_i^2 + \sigma_{w_i}^2}\right)$$
(3.67)

then

$$W_{i}(\Delta) = \operatorname{Min}_{P_{i}^{2}} f(\Delta, P_{i}^{2}) .$$

Note that W_i is a continuous function of its argument if W_{i+1} is, since the continuity of W_{i+1} implies continuity of f. From the continuity of W_{N+1} , (which was defined to be zero), the continuity of W_i follows for all i. We also note that as $P_i^2 \rightarrow \infty$, $f(\Delta, P_i^2) \rightarrow \infty$ also, and since $P_i^2 \ge 0$, the search for P_i^* can be confined to a compact set over which a continuous function always admits a minimum.

The dynamic program (3.65) can therefore be solved, yielding values for

$$P_{0}^{*}, P_{1}^{*}, \ldots, P_{N}^{*},$$

and we have the following theorem.

$$\underset{h^{N},\gamma^{N}}{\text{Minimize } J(h^{N},\gamma^{N})},$$

subject to (3.1a) through (3.1e) where $J(h^N, \gamma^N)$ is defined by Equation (3.60).

(a) The optimum policies $\{h_i^*\}$ and $\{\gamma_i^*\}$ for this problem are given by (3.52) and (3.58), respectively, using λ_i 's and Δ_i 's as defined by Equations (3.51) and (3.53), with the solution to the dynamic program (3.65) providing the optimum power levels, i.e.,

$$P_i^2 = P_i^* \text{ for } i=0,...,N$$
.

(b) The optimum cost is given by

$$W_{o}(\sigma_{x_{o}}^{2}) = \sum_{i=0}^{N} q_{i}P_{i}^{*2} + a'_{i}b'_{i}^{2}\Delta_{i}^{*},$$

with the $\Delta *_i$'s being defined by (3.63).

An Illustration

The optimal power levels depend critically on the power penalties (q_i) . If for the problem stated at the beginning of this section we assume N=1 and the following parameter values

$$\sigma_{x_o}^2 = 1.0, \ \sigma_{w_o}^2 = 1.0, \ \sigma_{w_1}^2 = 1.0, \ \sigma_{m_o}^2 = 1.0$$

$$q_o = 2.0, \ q_1 = 4.0, \ \rho_o = 0.5, \ a'_o = 1.0$$

$$b'_o = 1.0, \ a'_1 = 2.0 \ \text{and} \ b'_1 = 1.0,$$

then the optimal value of the cost is 3.5, which is attained by $P_o^2 = P_1^2 = 0.0$. If the power penalty q_o is changed to 0.25 with all other parameters remaining the same, we can achieve an optimal cost of 2.9747, which is attained by using $P_o^2 = 1.4495$ and $P_1^2 = 0.0$.

If the power penalty q_1 is also changed to 0.25, then the optimal cost is further reduced to 1.9968, which is attained by $P_0^2 = 1.1609$ and $P_1^2 = 1.9876$. It is notable that the optimal solution satisfies a threshold property, and the number of channels in use depends on the relative magnitudes of the weighting terms.

3.7. The Infinite Horizon Problem

Notation

<u>5550</u> • 5555

Let x, denote the realization of a first-order Markov process:

$$x_{i+1} = \rho x_i + m_i$$
. (3.68)

Here x_0 , m_0 , m_1 ,... are zero mean Gaussian random variables:

$$\begin{aligned} \mathbf{x}_{0} &\sim \mathrm{N}(0, \sigma_{\mathrm{x}}^{2}) \\ \mathbf{m}_{\mathrm{i}} &\sim \mathrm{N}(0, \sigma_{\mathrm{m}}^{2}) \;. \end{aligned}$$

The measurement y_i is a noise corrupted version of the control u_i :

$$\mathbf{y}_{i} = \mathbf{u}_{i} + \mathbf{w}_{i} \tag{3.69}$$

where w_0 , w_1 ,... are zero mean Gaussian random variables:

$$w_i \sim N(0, \sigma_w^2)$$

We are concerned with obtaining the optimal solution to Problem P^{∞} below.

Problem P^{°°}

Minimize
$$J(h^{\infty}, \gamma^{\infty}) = E[\sum_{i=0}^{\infty} (qu_i^2 + a(v_i - x_i)^2)\beta^i]$$

where a, q are given positive constants, β is the given discount factor ($0 < \beta < 1$) and

8 60 (H 88 8 8 55 55 Š. Ω × X 3 C N R N 3 -1 X ν٦. 5

$$\mathbf{v}_{i} = \boldsymbol{\gamma}_{i}(\mathbf{y}^{i}) \,. \tag{3.70b}$$

We treat the infinite horizon problem as a limit of the finite horizon case with horizon length N, as $N \rightarrow \infty$. This may be done provided that the discounted cost remains bounded and the optimum policy sequence converges to a well-defined limit.

The truncated version of the finite horizon problem, with horizon length N, is given as Problem P^N below:

Problem P^N

.

, , ,

Minimize
$$J_N(h^N, \gamma^N) = E[\sum_{i=0}^N (qu_i^2 + a(v_i - x_i)^2)\beta^i]$$

subject to (3.68) through (3.70).

For notational convenience, let $q'_i \triangleq q\beta^i$ and $a'_i \triangleq a\beta^i$.

We first consider the following hard constraint version of Problem P^{N} .

Problem PH^N

$$\underset{h^{N},\gamma^{N}}{\text{Minimize E}} E[\sum_{i=0}^{N} a'_{i} (v_{i} - x_{i})^{2}]$$

subject to

 $E[u_i^2] \leq P_i^2$

ß

۲۲ ۲۲ ۵۲

8

3

N.

ß

8

.H

 \tilde{x}

 \mathcal{X}

Š.

22

۲ ۱

į,

 $\langle \cdot \rangle$

S

60

under the constraint (3.68) through (3.70).

The following Lemma now follows directly from the analysis in the preceding sections.

Lemma 3.6: (a) Problem PH^N admits an optimal solution which is linear in the measurements and is given as follows:

$$u_{i}^{*} = h_{i}^{*}(x_{i}, y^{i-1}) = \lambda_{i}(x_{i} - E(x_{i} | y^{i-1}))$$

$$v_{i}^{*} = \gamma_{i}^{*}(y^{i}) = \hat{x}_{i}$$

where

$$\lambda_0^2 = \frac{P_0^2}{\sigma_w^2},$$
 (3.71a)

$$\lambda_{i}^{2} = \frac{P_{i}^{2}}{\rho^{2} \Delta_{i-1} + \sigma_{m}^{2}} \text{ for } i=1,...,N, \qquad (3.71b)$$

the
$$\Delta_i$$
's satisfy

$$\Delta_{0} = \frac{\sigma_{x}^{2} \sigma_{w}^{2}}{P_{o}^{2} + \sigma_{w}^{2}}, \qquad (3.72a)$$

$$\Delta_{i} = \frac{\sigma_{w}^{2}}{P_{i}^{2} + \sigma_{w}^{2}} \left(\rho^{2} \Delta_{i-1} + \sigma_{m}^{2}\right) \text{ for } i=1,...,N.$$
 (3.72b)

and $\mathbf{\hat{x}}_i = E(\mathbf{x}_i | \mathbf{y}^i)$ satisfies the recursion

$$\hat{x}_{0} = \frac{P_{0}\sigma_{x}}{P_{0}^{2} + \sigma_{x}^{2}} y_{0},$$

$$\hat{x}_{i} = \rho \hat{x}_{i-1} + \frac{P_{i}}{P_{i}^{2} + \sigma_{w}^{2}} (\rho^{2} \Delta_{i} + \sigma_{m}^{2})^{\frac{1}{2}} y_{i}.$$

(b) The minimum value of the cost is

$$J^* = \sum_{i=0}^{N} a'_i \Delta_i .$$

Now let J_p denote the infimum of $J_N(h^N, \gamma^N)$ under the hard power constraints, i.e.,

$$J_{\mathbf{P}} \stackrel{\Delta}{=} \inf_{\substack{\mathbf{h}^{N}, \boldsymbol{\gamma}^{N}, E[\mathbf{h}_{1}^{2}] = P_{1}^{2}}} J_{\mathbf{N}}(\mathbf{h}^{N}, \boldsymbol{\gamma}^{N}).$$

Using a sequence of equalities and inequalities as in (3.62), we have

$$J_{p} = \sum_{i=0}^{N} (qP_{i}^{2*} + a\Delta_{i}^{*})\beta^{i}$$

where the Δ_i 's are as defined in (3.72), and Δ_i^* 's are defined likewise, with P_i^2 replaced by P_i^{2*} .

The next task is to find the optimum power levels $\{P_i^{2^*}s\}$, which is done via the following deterministic optimal control problem:

The Deterministic N-stage Problem

$$\underset{P_0^2,\ldots,P_N^2}{\text{Minimize}} \sum_{i=0}^{N} (qP_i^2 + a\Delta_i)\beta^i \qquad (3.73)$$

subject to (3.72a) and (3.72b).

A CARLES AND A CARLE

Notation: For each positive scalar Δ , let $\{W_k(\Delta)\}_{k=0}^{N+1}$ be defined recursively by

$$W_{N+1}(\Delta) = 0 \tag{3.74a}$$

$$W_{k}(\Delta) = \inf_{P^{2}} \left[qP^{2} + \frac{a\sigma_{w}^{2}}{P^{2} + \sigma_{w}^{2}} (\rho^{2}\Delta + \sigma_{m}^{2}) + \beta W_{k+1} \left[\frac{\sigma_{w}^{2}}{P^{2} + \sigma_{w}^{2}} (\rho^{2}\Delta + \sigma_{m}^{2}) \right] \text{ for } k = N, ..., 0. \quad (3.74b)$$

Let $P_k^2 = P_k^2(\Delta)$ be a minimizing solution of the right-hand side of (3.74b), whenever it exists, and let $\{\Delta^*_k\}_{k=0}^N$ be the trajectory sequence defined recursively by

$$\Delta *_{0} = \left(\frac{\sigma_{w}^{2} \sigma_{x}^{2}}{P_{0}^{2} (\sigma_{x}^{2}) + \sigma_{w}^{2}}\right)$$
(3.75a)

$$\Delta *_{k} = \frac{\sigma_{w}^{2}}{P_{k}^{2}(\Delta *_{k-1}) + \sigma_{w}^{2}} \left(\rho^{2} \Delta *_{k-1} + \sigma_{m}^{2}\right). \qquad (3.75b)$$

Finally, let

$$P_0^{2*} = P_0^2(\sigma_x^2)$$
 (3.76a)

$$P_{k}^{2*} = P_{k}^{2}(\Delta *_{k-1}).$$
(3.76b)

Proposition 3.1:

- (i) The minimization problem (3.74b) admits a solution for each positive Δ .
- (ii) The control problem (3.73) admits a solution $\{P_k^{2*}\}_{k=0}^N$ which is given by (3.76) and the corresponding optimal trajectory is generated by (3.75).
- (iii) The minimum value for the optimal control problem is $W_0(\sigma_x^2)$.

The proof of the proposition above follows from the following lemmata.

Lemma 3.7: The value of the optimal control problem (3.73) is $J^* = W_0(\sigma_x^2)$ where $W_0(\cdot)$ is obtained through the recursive equations

$$W_{N+1}(\Delta) = 0 \tag{3.77a}$$

 $W_{k}(\Delta) = \inf_{p^{2}} \left[q P^{2} + \frac{a \sigma_{w}^{2}}{P^{2} + \sigma_{m}^{2}} (\rho^{2} \Delta + \sigma_{m}^{2}) + \beta W_{k+1} (\frac{\sigma_{w}^{2}}{P^{2} + \sigma_{m}^{2}} (\rho^{2} \Delta + \sigma_{m}^{2})) \right] \text{ for } k \leq N. \quad (3.77b)$

Furthermore, if the right-hand side of (3.77b) admits a solution sequence $\{P_k^2\}$, $k \leq N$, then

 $P_0^2(\sigma_x^2)$

and

$$P_k^2(\Delta *_{k-1})$$
 for $1 \le k \le N$

provide the optimal solution, where $\Delta *_k$ is generated by (3.75).

Proof: This follows from a standard dynamic programming argument.

Lemma 3.8: For every $\Delta > 0$, there exists a solution to the right-hand side of (3.74b).

Proof: Define $f(\Delta, P^2)$ by

$$f_{k}(\Delta, P^{2}) = qP^{2} + \frac{a\sigma_{w}^{2}}{P^{2} + \sigma_{w}^{2}} + W_{k+1}(\frac{\sigma_{w}^{2}}{P^{2} + \sigma_{w}^{2}}(\rho^{2}\Delta + \sigma_{m}^{2}))$$
(3.78)

so that

$$W_k(\Delta) = \inf_{P^2} f_k(\Delta, P^2).$$

Note that W_k , which is positive for all k, is a continuous function of its argument if W_{k+1} is, since the continuity of W_{k+1} implies continuity of f_k . From the continuity of W_{N+1} (which is defined to be zero) the continuity of W_k follows for all k. We then note that as $P^2 \rightarrow \infty$, so does $f(\Delta, P^2)$, and since $P^2 \ge 0$, the search for an optimum may be confined to a compact set over which a continuous function always admits a minimum.

13

ŝ

The following lemmata will be used in the construction of the solution for the infinite horizon problem.

Lemma 3.9: $W_k(\Delta)$ is strictly increasing for decreasing k, for all $\Delta > 0$, i.e., $W_k(\Delta) > W_{k+1}(\Delta) \ \forall k \leq N.$

Proof: Clearly the lemma is true for k=N, since $W_{N+1} \triangleq 0$ and $W_N(\Delta)$ is necessarily larger than zero for all Δ . We now note the following sequence of equalities and inequalities:

$$\begin{split} & W_{k}(\Delta) - W_{k+1}(\Delta) \\ &= \underset{P^{2}>0}{\text{Min}}[qP^{2} + \frac{a\sigma_{w}^{2}}{p^{2} + \sigma_{w}^{2}} (\rho^{2}\Delta + \sigma_{m}^{2}) + \beta W_{k+1}(\frac{\sigma_{2}^{2}}{P^{2} + \sigma_{w}^{2}} (\rho^{2}\Delta + \sigma_{m}^{2}))] \\ &- \underset{P^{2}>0}{\text{Min}}[qP^{2} + \frac{a\sigma_{w}^{2}}{p^{2} + \sigma_{w}^{2}} (\rho^{2}\Delta + \sigma_{m}^{2}) + \beta W_{k+2}(\frac{\sigma_{w}^{2}}{P^{2} + \sigma_{w}^{2}} (\rho^{2}\Delta + \sigma_{m}^{2}))] \\ &\geqslant q\hat{P}^{2} + \frac{a\sigma_{w}^{2}}{\hat{P}^{2} + \sigma_{w}^{2}} (\rho^{2}\Delta + \sigma_{m}^{2}) + \beta W_{k+1}(\frac{\sigma_{w}^{2}}{\hat{P}^{2} + \sigma_{w}^{2}} (\rho^{2}\Delta + \sigma_{m}^{2}))] \\ &- q\hat{P}^{2} + \frac{a\sigma_{w}^{2}}{\hat{P}^{2} + \sigma_{w}^{2}} (\rho^{2}\Delta + \sigma_{m}^{2}) - \beta W_{k+2}(\frac{\sigma_{w}^{2}}{\hat{P}^{2} + \sigma_{w}^{2}} (\rho^{2}\Delta + \sigma_{m}^{2}))] \\ &= \beta (W_{k+1} - W_{k+2}) \cdot (\frac{\sigma_{w}^{2}(\rho^{2}\Delta + \sigma_{m}^{2})}{\hat{P}^{2} + \sigma_{w}^{2}}) \end{split}$$

where $\boldsymbol{\hat{P}}^2$ is chosen as the argument of the first minimization. (In case of nonunique solutions, any one of the minimizing solutions may be chosen).

Thus if $W_{k+1}(\Delta)$ is larger than $W_{k+2}(\Delta)$, then $W_k(\Delta)$ is larger than $W_{k+1}(\Delta)$. Since $W_N(\Delta)$ is known to be larger than $W_{N+1}(\Delta)$, the proof is complete.

Lemma 3.10: $W_k(\Delta)$ is an increasing function of $\Delta \{W_k(\Delta) \uparrow \Delta\}$ for all $k \leq N$.

Proof: We prove this by induction. First consider the case with k=N. We have

$$W_{N}(\Delta) = \operatorname{Min}[qP^{2} + \frac{a\sigma_{w}^{2}}{P^{2} + \sigma_{w}^{2}} (\rho^{2}\Delta + \sigma_{m}^{2})].$$

65

There are only two possible cases, either $P^{2*} = 0$, or $P^{2*} > 0$. If $P^{2*} = 0$, then $W_N(\Delta) = a(\rho^2 \Delta + \sigma_m^2)$. If $P^{2*} > 0$, which requires

$$q - \frac{a\sigma_{w}^{2}(\rho^{2}\Delta + \sigma_{m}^{2})}{(p^{2*} + \sigma_{w}^{2})^{2}} = 0$$

i.e.,

$$P^{2*} = \left[\frac{a\sigma_{w}^{2}(\rho^{2}\Delta + \sigma_{m}^{2})}{q}\right]^{\frac{1}{2}} - \sigma_{w}^{2}$$
(3.79)

and we get

$$W_{N}(\Delta) = 2a^{\frac{1}{2}}q^{\frac{1}{2}}\sigma_{w}(\rho^{2}\Delta + \sigma_{m}^{2})^{\frac{1}{2}} - q\sigma_{w}^{2}.$$
 (3.80)

Thus the lemma is true for k=N.

Now, if
$$W_{k+1}(\Delta) \uparrow \Delta$$
, then for each P^2

$$W_{k+1}\left(\frac{a\sigma_{w}^{2}}{P^{2}+\sigma_{w}^{2}}\left(\rho^{2}\Delta+\sigma_{m}^{2}\right)\right)\uparrow\Delta,$$

since

$$\frac{a\sigma_{w}^{2}}{P^{2}+\sigma_{w}^{2}}(\rho^{2}\Delta+\sigma_{m}^{2})\uparrow\Delta.$$

Also,

$$(qP^{2} + \frac{a\sigma_{w}^{2}(\rho^{2}\Delta + \sigma_{m}^{2})}{P^{2} + \sigma_{w}^{2}}) \uparrow \Delta,$$

and thus both terms in the expression to be minimized to obtain $W_K(\Delta)$ are increasing in Δ

22220 E2000 2220 E2000 E2000 E2000
66

for all P^2 . Therefore, $W_k(\Delta) \uparrow \Delta$.

Lemma 3.11: For each $\Delta > 0$, $W_k(\Delta)$ is bounded above for all k, by an affine function of Δ , i.e.,

$$0 < W_{\mathbf{k}}(\Delta) \leq \Omega_{1,\mathbf{k}}\Delta + \Omega_{2,\mathbf{k}}.$$

Proof: The proof is by induction, using the observation that since $W_k(\Delta)$ is given by the minimum over P^2 , an upper bound is given by the value that the expression to be minimized attains when P^2 is fixed arbitrarily at zero.

Thus $W_N(\Delta) \leq a(\rho^2 \Delta + \sigma_m^2)$ and we may choose $\Omega_{1,N} = a\rho^2, \Omega_{2,N} = a\sigma_m^2$.

Now consider the following sequence of equalities and inequalities

$$\begin{split} W_{k}(\Delta) &= \operatorname{Min}[qP^{2} + \frac{a\sigma_{w}^{2}}{P^{2} + \sigma_{w}^{2}} \left(\rho^{2}\Delta + \sigma_{m}^{2}\right) + \beta W_{k+1}\left(\frac{\sigma_{w}^{2}}{P^{2} + \sigma_{w}^{2}} \left(\rho^{2}\Delta + \sigma_{m}^{2}\right)\right)] \\ &\leq \operatorname{Min}[qP^{2} + \frac{a\sigma_{w}^{2}}{P^{2} + \sigma_{w}^{2}} \left(\rho^{2}\Delta + \sigma_{m}^{2}\right) + \beta W_{k+1}(\rho^{2}\Delta + \sigma_{m}^{2})] \\ &\leq a(\rho^{2}\Delta + \sigma_{m}^{2}) + \beta W_{k+1}(\rho^{2}\Delta + \sigma_{m}^{2}) \\ &\leq a(\rho^{2}\Delta + \sigma_{m}^{2}) + \beta(\Omega_{1,k+1}(\rho^{2}\Delta + \sigma_{m}^{2}) + \Omega_{2,k+1}) \\ &= (a\rho^{2} + \beta\Omega_{1,k+1}\rho^{2})\Delta + a\sigma_{m}^{2} + \beta\Omega_{1,k+1}\sigma_{m}^{2} + \beta\Omega_{2,k+1} \\ &\triangleq \Omega_{1,k}\Delta + \Omega_{2,k} \,. \end{split}$$

Therefore, the lemma is proved with the sequences $\{\Omega_{1,k}\}$ and $\{\Omega_{2,k}\}$ defined recursively by

 $\Omega_{1,N} = a\rho^2 , \quad \Omega_{2,N} = a\sigma_m^2$ $\Omega_{1,k} = a\rho^2 + \beta\rho^2 \Omega_{1,k+1}$ $\Omega_{2,k} = a\sigma_m^2 + \beta\sigma_m^2 \Omega_{1,k+1} + \beta\Omega_{2,k+1} .$

We now return to the study of the infinite horizon problem. Note that since the optimal policy for the stochastic control problem is linear, the stationary limiting policy is given by

$$h_{n}^{*}(x_{n}, y^{n-1}) = \lambda^{*}(x_{n} - E(x_{n} | y^{n-1}))$$
(3.81)

where

$$\lambda^{*2} = \frac{p^{2*}}{\rho^2 \Delta^{*+} \sigma_m^2}, \qquad (3.82)$$

with P^{2*} and $\Delta *$ being obtained through the stationary solution of the optimum control problem as $N \rightarrow \infty$.

For each N, denote the solution given by Proposition 3.1. (ii) by $\{P_k^{2^*}\}^N$, k < N. We then expect that

$$P^{2*} = \lim_{N \to \infty} \{P_k^{2*}\}^N$$

for every finite k.

To establish the existence of this limit, we recall that $W_k(\Delta)$ is strictly increasing for decreasing k<N (Lemma 3.9) and further that it is bounded above by an affine function (Lemma 3.11). This last property follows since both Ω_{1k} and Ω_{2k} are bounded in retrograde time,

<u>}</u>>

.

68

$$\Omega_{1k} < (\frac{a\rho^2}{1-\beta\rho^2}) \text{ and } \Omega_{2,k} < \frac{a\sigma_m^2}{(1-\beta)(1-\rho^2\beta)}.$$
(3.83)

Hence, Limit $W_k(\Delta) = W(\Delta)$ where the limiting function satisfies

$$W(\Delta) = Min_{P^{2}} \left[qP^{2} + \frac{a\sigma_{w}^{2}(\rho^{2}\Delta + \sigma_{m}^{2})}{P^{2} + \sigma_{w}^{2}} + \beta W \left[\frac{\sigma_{w}^{2}}{P^{2} + \sigma_{w}^{2}} (\rho^{2}\Delta + \sigma_{m}^{2}) \right] \right].$$
(3.84)

Denote the minimizing solution here by $P^2(\Delta)$. Now note that since $\rho^2 < 1$, Equation (3.75b) describes a stable system with P_k^2 replaced by P^2 , and hence $\Delta_k \rightarrow \Delta^*$ where Δ^* solves

$$\Delta^{*} = \frac{\sigma_{w}^{2}}{P^{2}(\Delta^{*}) + \sigma_{w}^{2}} \left(\rho^{2} \Delta^{*} + \sigma_{m}^{2}\right).$$
(3.85)

Let

$$P^{2*} \equiv P^2(\Delta^*) \,. \tag{3.86}$$

Then we have the following solution to the infinite horizon problem:

Theorem 3.5: With $N \rightarrow \infty$ in (3.73), the stochastic control problem in consideration admits the optimal stationary policies,

$$h_{n}^{*} = \lambda^{*}(x_{n} - E(x_{n} | y^{n-1}))$$

for n sufficiently large, where

$$\lambda^{*2} = \frac{P^{2*}}{\rho^2 \Delta^{*+} \sigma_m^2}$$

with Δ^* and P^{2*} given by (3.85) and (3.86).

 \square

To numerically compute the optimal stationary policies, we start with $\Delta_0=0$ and run the following algorithm:

Algorithm A:

(1) Compute

$$(P_{k}^{2})^{*} = \arg \min_{p^{2} \ge 0} \min \left[q P^{2} + \frac{a \sigma_{w}^{2} (\rho^{2} \Delta_{k} + \sigma_{m}^{2})}{p^{2} + \sigma_{w}^{2}} + \frac{\beta (q P^{2} + a \Delta_{k})}{(1 - \beta)} \right]$$

(2) Compute the new value, Δ_{k+1} , by

$$\Delta_{\mathbf{k}+1} = \frac{\sigma_{\mathbf{w}}^2}{(\mathsf{P}_{\mathbf{k}}^2)^* + \sigma_{\mathbf{w}}^2} \ (\rho^2 \Delta_{\mathbf{k}} + \sigma_{\mathbf{m}}^2)$$

(3) Go to step (1), and iterate.

The optimal cost is then given by

W(
$$\Delta^*$$
) = (q(P²)* + $\frac{a\sigma_w^2(\rho^2\Delta^* + \sigma_m^2)}{(P^2)^* + \sigma_w^2})/(1 - \beta)$.

We next show that Algorithm A always converges. We first note that $(P_k^2)^*$, found from step (1) of the algorithm, satisfies

$$(P_{k}^{2})^{*} = Max\{0, (\frac{a^{\frac{1}{2}}(1-\beta)^{\frac{1}{2}}}{q^{\frac{1}{2}}} \sigma_{w}(\rho^{2}\Delta_{k}+\sigma_{m}^{2})^{\frac{1}{2}}-\sigma_{w}^{2})\}$$

which implies that if $\Delta_{k+1} \ge \Delta_k$, then $(P_{k+1}^2)^* \ge (P_k^2)^*$.

Now, given $\Delta_k > \Delta_{k-1}$, we have

c

8

3

8

3) 2)

80

• •

S

Ñ

5

33 33

10



Therefore, since $\Delta_1 > \Delta_0$, it follows that the Δ_k 's form a monotone increasing sequence. Further, to show that the Δ_k 's are bounded above, we consider the sequence

$$\Gamma_0 = 0$$

$$\Gamma_{k+1} = \rho^2 \Gamma_k + \sigma_m^2,$$

and note that if $\Gamma_k \ge \Delta_k$, we have

$$\Gamma_{k+1} = \rho^{2} \Gamma_{k} + \sigma_{m}^{2}$$

$$\geqslant \rho^{2} \Delta_{k} + \sigma_{m}^{2}$$

$$\geqslant \frac{\sigma_{w}^{2}}{(P_{k}^{2})^{*} + \sigma_{w}^{2}} (\rho^{2} \Delta_{k} + \sigma_{m}^{2})$$

$$= \Delta_{k+1},$$

i.e., if $\Gamma_k \ge \Delta_k$, then $\Gamma_{k+1} \ge \Delta_{k+1}$. But $\Gamma_0 = \Delta_0 (\equiv 0)$, and the sequence Γ_k is bounded above by

$$\frac{\sigma_{\rm m}^2}{(1-\rho^2)}$$

Therefore, the monotone sequence Δ_k is also bounded above, and the convergence of the algorithm follows.

2 X 20 Š **%** R ۶¢ 8 N N \hat{S} ŝ $\left\{ \cdot \right\}$. 70

1

Tables 3.1 through 3.4 provide a few of these convergence results for given parameter values.

TABLE 3.1. CONVERGENCE RESULTS FOR THE PROBLEM WITH $a = \sigma_v^2 = \sigma_m^2 = 1.0$, $\rho^2 = \beta = q = 0.5$.

Iter. #	Δ	(<i>P</i> ²)*
0	0.0	0.0
1	1.0	0.22474
2	1.22474	0.26979
3	1.26979	0.27863
4	1.27863	0.28035
5	1.28035	0.26069
6	1.28069	0.28076
7	1.28076	0.28077
8	1.28077	0.28077
9	1.28077	0.28077

TABLE 3.2. CONVERGENCE RESULTS FOR THE PROBLEM WITH $a = \sigma_m^2 = 1.0$. $\rho^2 = \beta = 0.5$, $q = \sigma_w^2 = 0.1$.

Iter. #	Ł	(P ²)*
0	0.0	0.60710
1	0.14142	0.63167
2	0.14633	0.63251
3	0.14650	0.63254
4	0.14650	0.63254

TABLE 3.3.	CONVERGENCE	RESULTS FOR	THE	PROBLEM	WITH	$a = \sigma_m^2 = 1.0,$
	$\rho^2 = \beta = 0.5, q = 0$	$0.1, \sigma_{w}^{2} = 2.0.$				

Iter. #	Δ	(<i>P</i> ²)*
0	0.0	1.16227
1	0.63245	1.62798
2	0.72559	1.69161
3	0.73832	1.70021
4	0.74004	1.70138
5	0.74027	1.70153
6	0.74030	1.70155
7	0.74031	1.70156
8	0.74031	1.70156

TABLE 3.4. CONVERGENCE RESULTS FOR $\rho^2 = \sigma_m^2 = \beta = 0.5$, $a = \sigma_w^2 = 1.0$, q = 0.1.

l

(**P**²)* Iter. # Δ 0.0 0.58113 0 0.81399 0.31622 -1 2 0.36279 0.84580 0.36916 0.85010 3 0.37002 0.85069 4 0.37013 0.85076 5 6 0.37015 0.85078 7 0.37015 0.85078

THE

PROBLEM W

WITH

8

8

١Ľ

\$

Ř

 $\tilde{\zeta}$

8

S.

*

8

Kr

 $\hat{\vec{x}}$

ः स्ट D.

We finally note that an infinite horizon version of the originally formulated Problem P1, can be solved via an equivalent problem of the form P^{∞} .

Consider the problem of minimizing $J(h^{\infty}, \gamma^{\infty})$ where

$$J(h^{\infty},\gamma^{\infty}) = E[\sum_{i=0}^{\infty} \beta^{i}(qu_{i}^{2} + a\beta x_{i+1}^{2} + bv_{i}^{2})]$$

subject to (3.1b), (3.1d) and (3.1e), where

 $\mathbf{x}_{i+1} = \rho \mathbf{x}_i + \mathbf{m}_i - \mathbf{v}_i \,.$

Using (3.4) and (3.5), we find that the cost can be rewritten as

$$J = \sum_{i=0}^{\infty} q\beta^{i} u_{i}^{2} + \sum_{i=0}^{\infty} a'_{i} (\tilde{v}_{i} - b'_{i} \tilde{x}_{i})^{2},$$

where

$$a'_i \rightarrow \beta^i(b+k\beta) \equiv a'\beta^i$$

and

$$b'_i \rightarrow \frac{k\beta\rho}{a'} \equiv b'$$

with the k found by solving for the positive root of the equation:

$$(k-a)(k\beta+b) = kb\beta\rho^2$$
.

i.e., the k satisfies

$$\mathbf{k} = \frac{1}{2\beta} \left(\sqrt{(\mathbf{b} - \mathbf{b}\beta\rho^2 - \mathbf{a}\beta)^2 + 4\mathbf{a}\mathbf{b}\beta} - (\mathbf{b} - \mathbf{b}\beta\rho^2 - \mathbf{a}\beta) \right).$$

Thus, an infinite horizon version of the originally formulated problem, with discounted cost, may be solved by solving a problem of the form P^{∞} , with a replaced by $a'(b')^2$.

In this chapter we have studied the problem of simultaneous communication and control for first-order ARMA models with feedback. These are stochastic team problems, where the design of the measurement strategy itself is a part of the problem. As such, they are difficult to analyze because of the nonclassical nature of the information structure. For cases with hard power constraints on the measurement strategies, we have shown that the optimum measurement policy is to linearly amplify the innovation at each stage, to the maximum permissible power level. For the cases with soft power constraints the structure of the solution is similar; however, now the optimal design of the power levels is also a part of the problem. These optimum power levels may be found via dynamic programming. We have then studied some infinite horizon stochastic team problems involving a first-order ARMA model, established the existence of optimal stationary policies for these problems, provided an algorithm that always converges to the optimal solution, and also provided some numerical examples illustrating the calculation of these stationary policies.

CHAPTER 4

SIMULTANEOUS COMMUNICATION AND CONTROL: GENERAL ARMA MODELS WITH FEEDBACK

4.1. Introduction

K

ř.

In this chapter we study the problem of simultaneously designing communication and control strategies for problems involving ARMA models of orders higher than one. In Section 4.2 we formulate the general problem and then the transformed problem consisting of squared differences. In Section 4.3 we study one of the simplest such problems involving a second-order ARMA model, and show that the strategies which are optimal over the linear class may be outperformed by appropriately chosen nonlinear strategies. In Section 4.4 we consider optimality over the affine class, and show that within this class the optimal policy consists of transmitting only the current innovation, multiplied by a gain factor. In Section 4.5 we study the linear solutions for the second-order ARMA model; this illustrates the methodology and concepts for the more complex problem involving the general model which is treated in Section 4.6. The concluding remarks in Section 4.7 then end this chapter.

4.2. Problem Formulation

In Chapter 3 we had studied a stochastic dynamic system involving a first-order ARMA model, with the current state directly correlated only with the immediately preceding state. In case we allow this correlation to extend to j previous stages, we obtain a jth-order ARMA model. Accordingly, let us suppose that the stochastic system is specified by the following set of equations:

75

$$\mathbf{x}_{i+1} = \sum_{k=0}^{j-1} \rho_{i+1,i-k} \mathbf{x}_{i-k} + \mathbf{m}_i - \mathbf{v}_i$$
(4.1)

and

$$\mathbf{y}_i = \mathbf{u}_i + \mathbf{w}_i \tag{4.2}$$

along with

$$a_i = h_i(x_i, y^{i-1})$$
 (4.3)

$$\mathbf{v}_{i} = \boldsymbol{\gamma}_{i}(\mathbf{y}^{1}) \,. \tag{4.4}$$

Here (4.1) and (4.2) are the state and measurement equations respectively, the random variables x_0 , w_i and m_i ($i \ge 0$) are assumed to be independent, zero-mean and Gaussian with variances $\sigma_{(\cdot)}^2$, (the subscript being the identifier), and $\rho_{k,j} \equiv 0$ for j < 0. The functions h_i and γ_i , $i \ge 0$, are the communication and control policies, respectively, each Borel measurable in its arguments, and leading to second-order random variables u_i and v_i , respectively.

As seen in Chapter 3, we may formulate a *hard* constraint version of the problem, by restricting the communication policies to satisfy

$$\mathsf{E}[\mathsf{u}_{i}^{2}] \leqslant \mathsf{P}_{i}^{2} \tag{4.5}$$

or the soft constraint version may be formulated, using the additional term

$$\mathsf{E}[\sum_{i=1}^{n} \mathsf{q}_i \mathsf{u}_i^2] \tag{+.6}$$

in the cost functional, implying that a tradeoff between higher signalling costs and lower estimation costs is permissible.

These soft and hard constraint versions are given below as Problems PS° and PH°, respectively.

No Contracto Contracto

CCCCC 4

Problem PS°

 $\underset{h^{N},\gamma^{N}}{\text{Minimize } E[\sum_{i=0}^{N} q_{i}u_{i}^{2} + c_{i+1}x_{i+1}^{2} + d_{i}v_{i}^{2}]}$

subject to (4.1) through (4.4).

Problem PH°

Minimize
$$E[\sum_{i=0}^{N} c_{i+1}x_{i+1}^{2} + d_{i}v_{i}^{2}]$$

subject to (4.1) through (4.5).

Using completion of squares and a redefinition of the v_i 's, (as in the case of the first-order ARMA model) we may obtain the equivalent problems PS and PH below:

Problem PS

Minimize E[
$$\sum_{i=0}^{N} q_i u_i^2 + a_i (\tilde{v}_i - \sum_{k=0}^{j-1} b_{i,i-k} \tilde{x}_{i-k})^2$$
]

subject to (4.2) through (4.4), with x_i replaced by \tilde{x}_i and v_i replaced by \tilde{v}_i .

Problem PH

$$\underset{\substack{h^{N},\gamma^{N}}}{\text{Minimize }} E[a_{i}(\tilde{v}_{i} - \sum_{k=0}^{j-1} b_{i,i-k} \tilde{x}_{i-k})^{2}]$$

subject to (4.2) through (4.5), with x_i replaced by \tilde{x}_i and v_i replaced by \tilde{v}_i . For these transformed problems, the \tilde{x}_i 's are given by the following recursion:

$$\tilde{\mathbf{x}}_{i+1} = \sum_{k=0}^{j-1} \rho_{i+1,i-k} \tilde{\mathbf{x}}_{i-k} + \mathbf{m}_i$$
(4.7)

where $\tilde{x}_0 = x_0$ is given, $\rho_{k,j} = 0$ for all j < 0 and $b_{k,j} = 0$ for j < 0. Furthermore, the precise (recursive) expressions for the a_i 's are given in Appendix A, where the details of the justification for this reformulation can also be found.

We now turn to analyzing these reformulated stochastic team problems.

4.3. Nonoptimality of Linear Laws

In this section we show that for one of the simplest team problems of the type above, involving an ARMA model of order 2, the optimum linear solution may be outperformed by an appropriately chosen nonlinear policy.

We first restrict our attention to the following stochastic team Problem P_a , of which a schematic representation is provided in Figure 4.1.



Figure 4.1. Schematics for Problem P_a.

Problem Pa

 $\underset{\substack{\gamma_1,\gamma_2,\delta}}{\text{Minimize }E[(x-v)^2]}$

where

$\mathbf{x}_1 = \mathbf{x} + \mathbf{n}_1$	(4.8)
$\mathbf{x}_1 = \mathbf{x} + \mathbf{n}_1$	(4.8)

$$\mathbf{u}_1 = \boldsymbol{\gamma}_1(\mathbf{x}) \tag{4.9}$$

$$y_1 = u_1 + w_1$$
 (4.10)

$$u_2 = \gamma_2(x_1, y_1)$$
 (4.11)

$$y_2 = u_2 + w_2$$
 (4.12)

and

$$\mathbf{v} = \delta(\mathbf{y}_1, \mathbf{y}_2) \tag{4.13}$$

subject to the hard power constraints:

$$\mathbf{E[u_1^2] \leqslant P_1^2} \tag{4.14a}$$

$$E[u_2^2] \leq P_2^2$$
. (4.14b)

Note that if the problem involved estimating x_1 at the decoder (instead of x), then we would have had the two-stage version of a problem involving a first-order ARMA model as studied in Chapter 3, for which the optimal solutions have been shown to be linear.

We now show that Problem P_a above does not, in general, admit an optimal linear solution. This is done by constructing an instance of the problem where the optimal linear strategies are outperformed by appropriately chosen nonlinear strategies.

In order to see why one might suspect nonoptimality of affine laws, consider the above problem with $\sigma_{w_2}^2 = 0$. We then have Problem P'_a below which is represented

schematically in Figure 4.2, and for which the hard power constraint on u_2 is immaterial since there is no noise to combat.

Problem P'

Minimize $E[(x-v)^2]$

subject to (4.8) through (4.11), and (4.13), along with the restriction

 $y_2 = u_2$. (4.15)

Note that since

 $\mathbf{v} = \delta(\mathbf{y}_1, \mathbf{y}_2) \tag{4.16}$

where

66566

 $y_2 = u_2 = \gamma_2(x_1, y_1)$



Figure 4.2. Schematics for Problem P'a.

we may equivalently write

$$\mathbf{v} = \delta(\mathbf{y}_1, \mathbf{y}_2(\mathbf{x}_1, \mathbf{y}_1))$$
 (4.17a)

$$=\delta'(y_1,x_1),$$
 (4.17b)

and thus Problem P'_{a} may equivalently be represented as in Figure 4.3 below.

We thus obtain a problem of simultaneously designing encoding and decoding policies with side information at the decoder, for which nonlinear strategies that outperform the optimal linear strategies do exist (see Appendix B).

Since linear policies are not optimal for Problem P_a with $\sigma_{w_2}^2 = 0$, they may continue to be nonoptimal for small enough values of $\sigma_{w_2}^2$. We show next that this is precisely the case. In particular, if we consider the optimal linear design for Problem P_a , using

$$u_1 = \gamma_1(\mathbf{x}) = \lambda_1 \mathbf{x}$$

$$u_2 = \gamma_2(\mathbf{x}_1, \mathbf{y}_1) = \lambda_2(\mathbf{x}_1 - \mathbf{E}(\mathbf{x}_1 + \mathbf{y}_1))$$

where λ_1 and λ_2 are chosen to meet the hard power constraints with equality, (this being the optimal choice in the affine class, as to be shown later), we have



Figure 4.3. Equivalent representation for Problem P'_a .

$$E(x_1 | y_1) = \frac{\lambda_1 \sigma_x^2}{P_1^2 + \sigma_{w_1}^2} y_1, \qquad (4.18)$$

Ŋ

_•)

N N

Ę.

· · ·

Ŷ

.

i,

and

$$\mathbf{x}_{1} - \mathbf{E}(\mathbf{x}_{1} | \mathbf{y}_{1}) = \mathbf{n}_{1} + \frac{\sigma_{\mathbf{w}_{1}}^{2}}{\mathbf{P}_{1}^{2} + \sigma_{\mathbf{w}_{1}}^{2}} \mathbf{x} - \frac{\lambda_{1} \sigma_{\mathbf{x}}^{2}}{\mathbf{P}_{1}^{2} + \sigma_{\mathbf{w}_{1}}^{2}} \mathbf{w}_{1}$$
(4.19)

which implies that

$$u_{2} = \lambda_{2} \left[\frac{\sigma_{w_{1}}^{2}}{P_{1}^{2} + \sigma_{w_{1}}^{2}} x - \frac{\lambda_{1} \sigma_{x}^{2}}{P_{1}^{2} + \sigma_{w_{1}}^{2}} w + n_{1} \right]$$

with

$$\lambda_2^2 = \frac{P_2^2(P_1^2 + \sigma_{w_1}^2)}{\sigma_x^2 \sigma_{w_1}^2 + \sigma_{n_1}^2(P_1^2 + \sigma_{w_1}^2)} .$$

The mean square error in estimating x from the simultaneous observation of \boldsymbol{y}_1 and \boldsymbol{y}_2 then is

$$E[(x - E(x | y_1, y_2))^2] = \frac{\sigma_x^2 \sigma_{w_1}^2}{(P_1^2 + \sigma_{w_1}^2)(P_2^2 + \sigma_{w_2}^2)} \left[\sigma_{w_2}^2 + \frac{P_2^2 \sigma_{n_1}^2}{(\sigma_{n_1}^2 + \sigma_x^2 \sigma_{w_1}^2/(P_1^2 + \sigma_{w_1}^2))} \right].$$
(4.20)

Considering the situation with $\sigma_x^2 = 100.0$, $\sigma_{n_1}^2 = 0.99$, $\sigma_{w_1}^2 = 1.0$, $\sigma_{w_2}^2 = 0.01$, $P_1^2 = 85.0423$ and $P_2^2 = 100.99$, we find that the optimal linear policy yields a cost of 0.53467.

We next consider the design

ないでしていたのでした。

 $\gamma_1(x) = x + \epsilon \operatorname{sgn} x$ $\gamma_2(x_1) = x_1$

and

$$\delta(y_1, y_2) = \begin{cases} (y_1 + y_2 - \epsilon)/2 & \text{if } y_2 \ge 0 \\ \\ y_1 + y_2 + \epsilon)/2 & \text{if } y_2 < 0 \end{cases}$$

(letting $\epsilon = -1.0$ we obtain $E[u_1^2] = 85.0423$ and $E(u_2^2) = 100.99$).

Now

$$y_2 = x + n_1 + w_2$$
$$\equiv x + w_3$$

where

$$w_3 \sim N(0,1)$$
.

If we calculate the mean square error under the above policy by an analysis similar to that used in Appendix B, we find that the nonlinear policy yields a cost of 0.53172, and hence is superior to the optimal linear policy.

We now return to the problem of showing nonoptimality of linear laws for at least some instances of higher-order ARMA models. Consider the following second-order model with feedback, illustrated in Figure 4.4. We have

$$x_{2} = \rho_{21}x_{1} + \rho_{20}x_{0} + m_{1}$$
$$x_{1} = \rho_{10}x_{0} + m_{0}$$

and x_0 , m_0 and m_1 are given independent, zero-mean, Gaussian random variables.



Figure 4.4. Schematics for a second-order model with feedback.

The problem is to minimize $E[(v_2-x_2)^2]$ under the schematics of Figure 4.4, with

 $v_2 = \gamma_2(y_0, y_1, y_2)$.

Let us now suppose that $\sigma_{w_2}^2$ is arbitrarily large, essentially making the third channel redundant, and therefore

$$E(x_2 | y_0, y_1, y_2) = E(x_2 | y_0, y_1).$$

Further suppose that

$$\rho_{21} = 0$$
, $\rho_{20} = 1$, $\sigma_{m_2}^2 = 0$, $\rho_{10} = 1$,

which imply

$$\begin{aligned} \mathbf{x}_2 &= \mathbf{x}_0 \\ \mathbf{x}_1 &= \mathbf{x}_0 + \mathbf{m}_1 \end{aligned}$$

and we obtain the problem depicted in Figure 4.5. We thus obtain a problem of the type

ىتك ¥¥

X

<u>A</u>

 \tilde{M}

B

.*)



Figure 4.5. The second-order ARMA model under the given restrictions.

 P_{a} , discussed earlier in this section, for which there are instances when linear strategies are not optimal. Hence we see that there are instances of the general problem (described by second-order ARMA processes) for which optimal linear designs are not optimal in the general class of policies.

4.4. Optimality over the Affine Class

In the preceding section we have shown that for ARMA models of order higher than one, the strategies which simultaneously provide optimum communication and control are not necessarily linear. In this section we show that if we confine the design to the affine class, then the optimal communication strategies use a linear transformation on the innovation process.

Theorem 4.1. Consider the general formulation of the Problem PS given in Section 4.2, but with h_i resticted to the class

$$\mathbf{u}_i = \mathbf{h}_i(\tilde{\mathbf{x}}_i, \mathbf{y}^{i-1}) = \mathbf{L}_i(\tilde{\mathbf{x}}_i, \mathbf{y}^{i-1})$$

where L_1 is a general affine mapping. Then one may, without loss of generality, be

Ľ

e:

85

2

ŝ

Ŵ

Ż

.

1

confined to optimizing over the class of transmitter policies which satisfy the structural restriction

$$u_i = \lambda_i (\tilde{x}_i - E(\tilde{x}_i | y^{i-1})).$$
 (4.21)

Thus it is sufficient to optimize over the class of policies which use a linear transformation on the innovation in $\boldsymbol{\tilde{x}}_i.$

Proof. Note that over the affine class we may write

$$\mathbf{u}_i = \mathbf{L}_i(\tilde{\mathbf{x}}_i, \mathbf{y}^{i-1}) = \tilde{\mathbf{u}}_i + \mathbf{p}_i$$

with

 $\tilde{\mathbf{u}}_{i} = \lambda_{i}(\tilde{\mathbf{x}}_{i} - \mathbf{E}(\tilde{\mathbf{x}}_{i} | \mathbf{y}^{i-1}))$

and

$$\mathbf{p}_{i} = \mathbf{L}'_{i}(\mathbf{y}^{i-1})$$

where L'_i is an arbitrary affine mapping, and \tilde{u}_i and p_i are uncorrelated. Thus, we have

$$\begin{split} \underset{\gamma_{N}}{\text{Min } J(h^{N}, \gamma^{N})} &= E\left[\sum_{i=0}^{N} (q_{i} u_{i}^{2} + a_{i} (E(\sum_{k=0}^{j-1} b_{i,i-k} \tilde{x}_{i-k} | y^{i}) - \sum_{k=0}^{j-1} b_{i,i-k} \tilde{x}_{i-k})^{2})\right] \\ &= E\left[\sum_{i=0}^{N} (q_{i} \tilde{u}_{i}^{2} + q_{i} p_{i}^{2} + a_{i} (E(\sum_{k=0}^{j-1} b_{i,i-k} \tilde{x}_{i-k} | y^{i}) - \sum_{k=0}^{j-1} b_{i,i-k} \tilde{x}_{i-k})^{2})\right] \\ &\geq E\left[\sum_{i=0}^{N} q_{i} \tilde{u}_{i}^{2} + a_{i} (E(\sum_{k=0}^{j-1} b_{i,i-k} \tilde{x}_{i-k} | y^{i}) - \sum_{k=0}^{j-1} b_{i,i-k} \tilde{x}_{i-k})^{2})\right]. \end{split}$$

We now note that the sigma field generated by \tilde{y}^i is the same as the sigma field generated by yⁱ where

$$\tilde{\mathbf{y}}_i = \tilde{\mathbf{u}}_i + \mathbf{w}_i$$
$$= \mathbf{y}_i - \mathbf{p}_i$$

since p_i is y^{i-1} measurable.

Hence,

$$\underset{\gamma_{N}}{\text{Min } J(h^{N}, \gamma^{N}) \ge E\left|\sum_{i=0}^{N} (q_{i}\tilde{u}_{i}^{2} + a_{i}(E(\sum_{k=0}^{j-1} b_{i,i-k}\tilde{x}_{i-k} | \tilde{y}^{i}) - \sum_{k=0}^{j-1} b_{i,i-k}\tilde{x}_{i-k})^{2})\right|$$

and therefore the cost functional may be optimized under the structural constraint (4.21).

4.5. Optimal Linear Strategies for Second-Order ARMA Models

We now concern ourselves with the design of communication and control strategies which simultaneously optimize over the linear class. In order to facilitate an understanding of the concepts and methodology, we restrict ourselves in this section to a study of second-order ARMA models. The general ARMA model will be studied in the next section.

(i) The problem with Hard Power Constraints.

 $Problem PH^2$

$$\underset{h^{N},\gamma^{N}}{\text{Minimize } E[\sum_{i=0}^{N} a_{i}(v_{i}-b_{i,i}\tilde{x}_{i}-b_{i,i-1}\tilde{x}_{i-1})^{2}]}$$

where

$$u_{i} = h_{i}(\tilde{x}_{i}, y^{i-1})$$
$$\tilde{v}_{i} = \gamma_{i}(y^{i})$$
$$E[u_{i}^{2}] = P_{i}^{2}$$

and the x_i's are generated via

**0\C\03\0e3\060\020\0300000035

$$\begin{split} \tilde{\mathbf{x}}_{1} &= \rho_{10} \tilde{\mathbf{x}}_{0} + \mathbf{m}_{0} \\ \tilde{\mathbf{x}}_{2} &= \rho_{21} \tilde{\mathbf{x}}_{1} + \rho_{20} \tilde{\mathbf{x}}_{0} + \mathbf{m}_{1} \\ \vdots \\ \tilde{\mathbf{x}}_{i+1} &= \rho_{i+1,1} \tilde{\mathbf{x}}_{i} + \rho_{i+1,i-1} \tilde{\mathbf{x}}_{i-1} + \mathbf{m}_{i} \\ \vdots \\ \tilde{\mathbf{x}}_{N+1} &= \rho_{N+1,N} \tilde{\mathbf{x}}_{N} + \rho_{N+1,N-1} \tilde{\mathbf{x}}_{N-1} + \mathbf{m}_{N} \,. \end{split}$$

3

s X

3

СХ.

535 303 433

.

4

PSYSSESSING

• 62252555 • 155252555 •

D

ŝ

Notation

Let

$$\Delta_i^j \equiv E[(\tilde{x}_i - E(\tilde{x}_i | y^j))^2]$$
(4.22a)

$$\Delta_{i\mathbf{k}}^{j} \equiv E[(\tilde{\mathbf{x}}_{i} - E(\tilde{\mathbf{x}}_{i} | \mathbf{y}^{j}))(\tilde{\mathbf{x}}_{\mathbf{k}} - E(\tilde{\mathbf{x}}_{k} | \mathbf{y}^{j}))]. \qquad (4.22b)$$

Theorem 4.2. The encoding and decoding policies for problem PH^2 , which are optimal over the linear class, are given by

$$u_{i} = \gamma_{i}(\tilde{x}_{i}, y^{i-1}) = \lambda_{i}(\tilde{x}_{i} - E(\tilde{x}_{i} + y^{i-1}))$$
(4.23)

$$\tilde{\mathbf{v}}_{i} = \delta_{i}(\mathbf{y}^{i}) = \mathbf{b}_{i,i} \mathbf{E}(\tilde{\mathbf{x}}_{i} | \mathbf{y}^{i}) + \mathbf{b}_{i,i-1} \mathbf{E}(\tilde{\mathbf{x}}_{i-1} | \mathbf{y}^{i})$$
 (4.24)

and the optimal cost is

$$\sum_{i=0}^{N} a_{i}(b_{i,i}^{2}\Delta_{i}^{i} + b_{i,i-1}^{2}\Delta_{i-1}^{i} + 2b_{i,i}b_{i,i-1}\Delta_{i,i-1}^{i})$$
(4.25)

where

$$E[\tilde{x}_{i} + y^{i}] = E[\tilde{x}_{i} + y^{i-1}] + E[\tilde{x}_{i} + y_{i}]$$
(4.26)

$$E[\tilde{x}_{i} | y^{i-1}] = \rho_{i,i-1} E[\tilde{x}_{i-1} | y^{i-1}] + \rho_{i,i-2} E[\tilde{x}_{i-2} | y^{i-1}]$$
(4.27)

$$E[\tilde{x}_{i-1} | y^{i}] = E[\tilde{x}_{i-1} | y^{i-1}] + E[\tilde{x}_{i-1} | y_{i}]$$
(4.28)

$$E[\tilde{x}_{i} | y_{i}] = \frac{\lambda_{i} \Delta_{i}^{-1}}{P_{i}^{2} + \sigma_{w_{i}}^{2}} y_{i}$$
(4.29)

Uparation Uparatic Uparatics

1.1

89

$$E[\tilde{x}_{i-1} | y_i] = \frac{\lambda_i (\rho_{i,i-1} \Delta_{i-1}^{i-1} + \rho_{i,i-2} \Delta_{i-1,i-2}^{i-1})}{P_i^2 + \sigma_{w_i}^2} y_i$$
(4.30)

where

 $\lambda_i^2 = \frac{P_i^2}{\Delta_i^{i-1}}$ (4.31)

and the $\Delta {}^{\prime} s$ are generated recursively by

$$\Delta_{i}^{i} = \Delta_{i}^{i-1} \frac{\sigma_{w_{i}}^{2}}{P_{i}^{2} + \sigma_{w_{i}}^{2}}$$
(4.32)

$$\Delta_{i}^{i-1} = \rho_{i,i-1}^{2} \Delta_{i-1}^{i-1} + \rho_{i,i-2}^{2} \Delta_{i-2}^{i-1} + 2\rho_{i,i-1}\rho_{i,i-2} \Delta_{i-1,i-2}^{i-1} + \sigma_{m_{i-1}}^{2}$$
(4.33)

$$\Delta_{i-1}^{i} = \Delta_{i-1}^{i-1} - \frac{P_i^2}{(P_i^2 + \sigma_{w_i}^2)} \left(\rho_{i,i-1} \Delta_{i-1}^{i-1} + \rho_{i,i-2} \Delta_{i-1,i-2}^{i-1}\right)^2$$
(4.34)

$$\Delta_{i,i-1}^{i} = (\rho_{i,i-1}\Delta_{i-1}^{i-1} + \rho_{i,i-2}\Delta_{i-1,i-2}^{i-1}) \frac{\sigma_{w_{i}}^{2}}{P_{i}^{2} + \sigma_{w_{i}}^{2}}$$
(4.35)

with the initial condition

$$\Delta_0^{-1} = \sigma_{x_0}^2; \Delta_{0,-1}^0 = 0 \text{ and } \Delta_{-1}^0 = 0.$$

X

Proof. Expressions (4.23) and (4.24) are immediate (since we know that the optimum u_i is linear in the innovation and the conditional expectation minimizes the mean square error), and (4.25) follows from the definition of Δ . Expressions (4.26) and (4.28) are due to the fact that y_i is independent of y^{i+1} and (4.27) follows from the definition of \tilde{x}_i .

 $\mathbf{y}_i = \lambda_i (\tilde{\mathbf{x}}_i - \mathbf{E}(\tilde{\mathbf{x}}_i | \mathbf{y}^{i-1})) + \mathbf{w}_i$

and

Now,

$$E(\tilde{\mathbf{x}}_i | \mathbf{y}_i) = E(\tilde{\mathbf{x}}_i - E(\tilde{\mathbf{x}}_i | \mathbf{y}^{i-1}) | \mathbf{y}_i)$$

hence we have (4.29), and (4.31) follows from the hard power constraint on $E[u_i^2]$.

To obtain (4.30) note that for any two zero-mean Gaussian random variables z_1 and z_2 , we may write

$$z_1 = \frac{E[z_1 z_2]}{E[z_2^2]} z_2 + n$$

where n is zero-mean, Gaussian, and independent of z_2 .

Thus we have

$$y_{i} = \lambda_{i} \rho_{i,i-1} (\tilde{x}_{i-1} - E(\tilde{x}_{i-1} | y^{i-1})) + \lambda_{i} \rho_{i,i-2} (\tilde{x}_{i-2} - E(\tilde{x}_{i-2} | y^{i-1})) + \lambda_{i} m_{i-1} + w_{i}$$

and writing

$$(\tilde{\mathbf{x}}_{i-2} - \mathbf{E}[\tilde{\mathbf{x}}_{i-2} | \mathbf{y}^{i-1}]) = \frac{\Delta_{i-1,i-2}^{i-1}}{\Delta_{i-1}^{i-1}} (\tilde{\mathbf{x}}_{i-1} - \mathbf{E}(\tilde{\mathbf{x}}_{i-1} | \mathbf{y}^{i-1})) + n,$$

(with n independent of $(\boldsymbol{\tilde{x}}_{i-1}{-}E(\boldsymbol{\tilde{x}}_{i-1}{+}\boldsymbol{y}^{i-1})))$ we have

$$\mathbf{y}_{i} = \lambda_{i} (\rho_{i,i-1} + \rho_{i,i-2} \frac{\Delta_{i-1,i-2}^{i-1}}{\Delta_{i-1}^{i-1}}) (\tilde{\mathbf{x}}_{i-1} - \mathbf{E}(\tilde{\mathbf{x}}_{i-1} | \mathbf{y}^{i-1})) + \mathbf{n}'$$

and therefore

ANT VERY CONTRACT AND ADDRESS

يور مور

$$\mathbf{E}[\mathbf{\tilde{x}}_{i-1} | \mathbf{y}_i] = \mathbf{E}[\mathbf{\tilde{x}}_{i-1} - \mathbf{E}(\mathbf{\tilde{x}}_{i-1} | \mathbf{y}^{i-1}) | \mathbf{y}_i]$$

is given by (4.30).

What remains to establish now, is the validity of the recursions for the Δ 's.

We have

$$\Delta_i^i = E[(\tilde{x}_i - E(\tilde{x}_i | y^i))^2]$$

= $E[(\tilde{x}_i - E(\tilde{x}_i | y^{i-1}) - E(\tilde{x}_i | y_i))^2]$
= $\Delta_i^{i-1} - E[E(\tilde{x}_i | y_i)^2]$

D.

and using (4.29), (4.32) follows. Further,

$$\Delta_i^{i-1} = E[(\tilde{x}_i - E(\tilde{x}_i | y^{i-1}))^2]$$

and using the definition of $\boldsymbol{\tilde{x}}_i$ we have (4.33).

Now

$$\Delta_{i-1}^{i} = E[(\tilde{x}_{i-1} - E(\tilde{x}_{i-1} | y^{i}))^{2}]$$

= $E[(\tilde{x}_{i-1} - E(\tilde{x}_{i-1} | y^{i-1}) - E(\tilde{x}_{i-1} | y_{i}))^{2}]$
= $\Delta_{i-1}^{i-1} - E[E(\tilde{x}_{i-1} | y_{i})^{2}]$
= $\Delta_{i-1}^{i-1} - \frac{P_{i}^{2}}{(P_{i}^{2} + \sigma_{w_{i}}^{2})} \frac{(\rho_{i,i-1}\Delta_{i-1}^{i-1} + \rho_{i,i-2}\Delta_{i-1,i-2}^{i-1})^{2}}{\Delta_{i}^{i-1}}$

i.e., we have (4.34). Finally,

$$\begin{aligned} \Delta_{i,i-1}^{i} &= E[(\tilde{x}_{i} - E(\tilde{x}_{i} | y'))(\tilde{x}_{i-1} - E(\tilde{x}_{i-1} | y'))] \\ &= E[(\tilde{x}_{i} - E(\tilde{x}_{i} | y^{i-1}) - E(\tilde{x}_{i} | y_{i}))(\tilde{x}_{i-1} - E(\tilde{x}_{i-1} | y^{i-1}) - E(\tilde{x}_{i-1} | y_{i}))] \\ &= \Delta_{i,i-1}^{i-1} - E[E(\tilde{x}_{i} | y_{i})E(\tilde{x}_{i-1} | y_{i})] \end{aligned}$$

and

$$\Delta_{i,i-1}^{i-1} = E[(\tilde{x}_i - E(\tilde{x}_i | y^{i-1}))(\tilde{x}_{i-1} - E(\tilde{x}_{i-1} | y^{i-1}))]$$
$$= \rho_{i,i-1} \Delta_{i-1}^{i-1} + \rho_{i,i-2} \Delta_{i-1,i-2}^{i-1},$$

T

92

$$E[E(\tilde{x}_{i} | y_{i}) E(\tilde{x}_{i-1} | y_{i})] = \frac{\lambda_{i}^{2} \Delta_{i}^{i-1} (\rho_{i,i-1} \Delta_{i-1}^{i-1} + \rho_{i,i-2} \Delta_{i-1,i-2}^{i-1})}{P_{i}^{2} + \sigma_{w_{i}}^{2}}$$

$$= \frac{P_i^2}{P_i^2 + \sigma_{w_i}^2} (\rho_{i,i-1} \Delta_{i-1}^{i-1} + \rho_{i,i-2} \Delta_{i-1,i-2}^{i-1})$$

implies (4.35).

The initial conditions for the above hard constraint version are immediate, and it is a straightforward task to verify that using the given recursions, all expressions required for the solution to PH^2 can be generated.

(ii) The Soft Constraint Version

It follows from the solution to the hard constraint version (as in the case of the first-order ARMA model) that Problem PS^2 may be solved via the following nonlinear optimal control problem.

 $Problem NL^2$

$$\begin{array}{l} \text{Minimize} \sum\limits_{P_0^2, \dots, P_N^2}^N (q_i P_i^2 + a_i b_{i,i}^2 \Delta_i^i + a_i b_{i,i-1}^2 \Delta_{i-1}^i + 2a_i b_{i,i} b_{i,i-1} \Delta_{i,i-1}^i) \end{array}$$

where the Δ 's are generated recursively by (4.32) through (4.35).

The solution to Problem NL^2 is given by the following dynamic program:

$$W_{N+1}(\Delta_1;\Delta_2;\Delta_3) = 0$$
$$W_i(\Delta_1;\Delta_2;\Delta_3)$$

along with

$$\begin{split} &= \operatorname{Min}[q_{i}P_{i}^{2} + a_{i}b_{i,i}^{2}(\frac{\Delta_{1}\sigma_{w_{i}}^{2}}{P_{i}^{2} + \sigma_{w_{i}}^{2}}) \\ &+ a_{i}b_{i,i-1}^{2}(\Delta_{2} - \frac{P_{i}^{2}}{(P_{i}^{2} + \sigma_{w_{i}}^{2})}\frac{(\rho_{i,i-1}\Delta_{2} + \rho_{i,i-2}\Delta_{3})^{2}}{(P_{i}^{2} + \sigma_{w_{i}}^{2})}) \\ &+ 2a_{i}b_{i,i}b_{i,i-1}(\frac{\sigma_{w_{i}}^{2}}{P_{i}^{2} + \sigma_{w_{i}}^{2}})(\rho_{i,i-1}\Delta_{2} + \rho_{i,i-2}\Delta_{3}) \\ &+ W_{i+1}(\rho_{i+1,i}^{2}\frac{\sigma_{w_{i}}^{2}}{P_{i}^{2} + \sigma_{w_{i}}^{2}}\Delta_{1} + \rho_{i+1,i-1}^{2}(\Delta_{2} - \frac{P_{i}^{2}}{(P_{i}^{2} + \sigma_{w_{i}}^{2})\Delta_{1}}(\rho_{i,i-1}\Delta_{2} + \rho_{i,i-2}\Delta_{3})^{2}) \\ &+ 2\rho_{i+1,i}\rho_{i+1,i-1}(\rho_{i,i-1}\Delta_{2} + \rho_{i,i-2}\Delta_{3})\frac{\sigma_{w_{i}}^{2}}{P_{i}^{2} + \sigma_{w_{i}}^{2}}; \\ &+ 2\rho_{i+1,i}\rho_{i+1,i-1}(\rho_{i,i-1}\Delta_{2} + \rho_{i,i-2}\Delta_{3})\frac{\sigma_{w_{i}}^{2}}{P_{i}^{2} + \sigma_{w_{i}}^{2}}; \\ &\frac{\Delta_{1}\sigma_{w_{i}}^{2}}{P_{i}^{2} + \sigma_{w_{i}}^{2}}; \frac{(\rho_{i,i-1}\Delta_{2} + \rho_{i,i-2}\Delta_{3})\sigma_{w_{i}}^{2}}{P_{i}^{2} + \sigma_{w_{i}}^{2}}). \end{split}$$

Further, the optimal value of the cost under linear policies is $W_0(\sigma_{x_0}^2;0;0)$.

4.6. Optimal Linear Strategies for the General ARMA Models

(i) The Transformed Problem with Hard Power Constraints

Problem PH^J

8

8

8

Š

3

Š

2

Minimize
$$E\left[\sum_{i=0}^{N} a_i (\tilde{v}_i - \sum_{k=0}^{j-1} b_{i,i-k} \tilde{x}_{i-k})^2\right]$$

where

$$u_{i} = h_{i}(\tilde{x}_{i}, y^{i-1})$$
$$\tilde{v}_{i} = \gamma_{i}(y^{i})$$
$$E[u_{i}^{2}] = P_{i}^{2}$$
$$y_{i} = u_{i} + w_{i}$$

and

Theorem 4.3. The encoding and decoding policies for Problem PH^{i} , which are optimal over the linear class, are given by

 $\tilde{\mathbf{x}}_{i} = \sum_{k=1}^{j} \rho_{i,i-k} \tilde{\mathbf{x}}_{i-k} + \mathbf{m}_{i-1} \ .$

$$u_i = \gamma_i(\tilde{x}_i, y^{i-1}) = \lambda_i(\tilde{x}_i - E(\tilde{x}_i | y^{i-1}))$$
 (4.36)

$$\tilde{\mathbf{v}}_{i} = \delta_{i}(\mathbf{y}^{i}) = \sum_{k=0}^{j-1} b_{i,i-k} E[\tilde{\mathbf{x}}_{i-k} | \mathbf{y}^{i}]$$
(4.37)

and the optimal cost is

$$J^* = \sum_{i=0}^{N} a_i \left(\sum_{k=0}^{j-1} \sum_{m=0}^{j-1} b_{i,i-k} b_{i,i-m} \Delta_{i-k,i-m}^i \right)$$
(4.38)

where

$$E[\tilde{x}_{j} | y^{i}] = E[\tilde{x}_{j} | y^{i-1}] + E[\tilde{x}_{j} | y_{i}]$$
(4.39)

$$E[\tilde{x}_{i} | y_{i}] = \frac{\lambda_{i} \Delta_{i}^{i-1}}{P_{i}^{2} + \sigma_{w}^{2}} y_{i}$$
(4.40)

$$\lambda_{i} \sum_{k=1}^{j} \rho_{i,i-k} \Delta_{i-k,m}^{i-1}$$

$$(4.41)$$

$$E[\tilde{x}_{m} | y_{i}] = \frac{k=1}{P_{i}^{2} + \sigma_{w_{i}}^{2}} y_{i}$$
(4.41)

$$E[\tilde{x}_{i} | y^{i-1}] = \sum_{k=1}^{J} \rho_{i,i-k} E[\tilde{x}_{i-k} | y^{i-1}]$$
(4.42)

$$\lambda_i^2 = \frac{P_i^2}{\Delta_i^{i-1}}$$
(4.43)

and the Δ 's are generated recursively by

and the second second

8

8

Š

N.

XX

Ŷ,

чт С

8

N.

 $\hat{\boldsymbol{\lambda}}$

5

5 4 A A A

R

$$\Delta_{i}^{i} = \frac{\Delta_{i}^{i-1} \sigma_{w_{i}}^{2}}{P_{i}^{2} + \sigma_{w}^{2}}$$
(4.44)

$$\Delta_{i}^{i-1} = \sigma_{m_{i-1}}^{2} + \sum_{k=1}^{j} \sum_{m=1}^{j} \rho_{i,i-k} \rho_{i,i-m} \Delta_{i-k,i-m}^{i-1}$$
(4.45)

$$\Delta_{\rm m}^{\rm i} = \Delta_{\rm m}^{\rm i-1} - \frac{{\rm P}_{\rm i}^{\rm 2}}{({\rm P}_{\rm i}^{\rm 2} + \sigma_{{\rm w}_{\rm i}}^{\rm 2})\Delta_{\rm i}^{\rm i-1}} (\sum_{\rm k=1}^{\rm j} \rho_{\rm i,i-k} \Delta_{\rm i-k,m}^{\rm i-1})^{\rm 2}; (\rm i-j \le m \le i-1)$$
(4.46)

$$\Delta_{n,m}^{i} = \Delta_{n,m}^{i-1} - \frac{P_{i}^{2}}{(P_{i}^{2} + \sigma_{w_{i}}^{2})} \left(\sum_{k=1}^{j} \rho_{i,i-k} \Delta_{i-k,n}^{i-1}\right) \left(\sum_{k=1}^{j} \rho_{i,i-k} \Delta_{i-k,m}^{i-1}\right)$$
(4.47)

$$\Delta_{n}^{i-1} = \sum_{k=1}^{J} \sum_{p=1}^{J} \rho_{n,n-k} \rho_{n,n-p} \Delta_{n-k,n-p}^{i-1} + \sigma_{m_{n-1}}^{2} (i-j \le n \le i-1)$$
(4.48)

$$\Delta_{n,m}^{i-1} = \sum_{k=1}^{j} \rho_{n,n-k} \Delta_{n-k,m}^{i-1} (i-j \le m < n \le i-1)$$
(4.49)

with $\Delta_0^{-1} = \sigma_{x_0}^2$; $\Delta_{k,m}^{-1} = 0$ for either k < 0 or m < 0.

Proof. The proof follows in a manner similar to that for the case j=2 discussed in the previous section.

(ii) The Transformed Problem with Soft Power Constraints

It follows from the solution to the problem with hard power constraints that Problem PS^j can be solved via the following nonlinear optimal control problem:

$$\underset{P_{0}^{2},\ldots,P_{N}^{2}}{\text{Minimize}} \sum_{i=0}^{N} q_{i}P_{i}^{2} + a_{i}(\sum_{k=0}^{j-1} \sum_{m=0}^{j-1} b_{i,i-k}b_{i,i-m}\Delta_{i-k,i-m}^{i})$$

where the Δ 's are defined via (4.44) through (4.49).

A SECTION OF CONTRACT OF CONTRACT.

The solution to the above optimal control problem may be obtained via the following dynamic program.

$$w_{N+1} = 0$$

$$W_{i}(\Delta_{i,i}^{i-1}, \dots, \Delta_{i,i-(j-1)}^{i-1}, \Delta_{i-1,i-1}^{i-1}, \dots, \Delta_{i-1,i-(j-1)}^{i-1}, \dots, \Delta_{i-(j-1),i-(j-1)}^{i-1}))$$

$$= M_{in} \left[q_{i} P_{i}^{2} + a_{i} (\sum_{k=0}^{j-1} \sum_{p=0}^{j-1} b_{i,i-k} b_{i,i-p} (\Delta_{i-k,i-p}^{i-1} - \frac{P_{i}^{2}}{\Delta_{i}^{i-1}(P_{i}^{2} + \sigma_{w_{1}}^{2})} (\sum_{n=1}^{j} \rho_{i,i-n} \Delta_{i-n,i-k}^{i-1}) (\sum_{m=1}^{j} \rho_{i,i-m} \Delta_{i-m,i-p}^{i-1}) \right]$$

$$+ W_{i+1} (\Delta_{i,i}^{i}, \dots, \Delta_{i,i-(j-1)}^{i}, \Delta_{i-1}^{i}, \dots, \Delta_{i-1,i-(j-1)}^{i}, \dots, \Delta_{i-(j-1),i-(j-1)}^{i}) \right]$$

where the relationship between the Δ 's is as given in (4.44) through (4.49). The optimal value of the cost is $W_0(\sigma_{x_0}^2, 0, 0, ..., 0)$.

(iii) Solutions to the Original Problem

The optimal linear solutions to the original problems PH° and PS° can be generated once the solutions to the transformed problems are available. We indicate how this may be done for the hard constraint version. For the problem with soft power constraints, the procedure is identical, except that we first need to find the optimal power levels by solving the nonlinear optimal control problem via dynamic programming.

Noting that since v_i is y^i measurable, we have

$$\mathbf{x}_{i} - \mathbf{E}(\mathbf{x}_{i} | \mathbf{y}^{i-1}) = \tilde{\mathbf{x}}_{i} - \mathbf{E}(\tilde{\mathbf{x}}_{i} | \mathbf{y}^{i-1})$$

and hence the optimal u_i 's are given by

$$u^{*}_{i} = \gamma^{*}_{i}(x_{i}, y^{i-1}) = \lambda_{i}(x_{i} - E(x_{i} | y^{i-1}))$$
,

with the λ 's being the same as those for the transformed version. To obtain the optimum v_i 's, note that

$$v_{i}^{*} = \delta_{i}^{*}(y^{i}) = \sum_{k=0}^{j-1} b_{i,i-k} E(x_{i} | y^{i}),$$

and recalling that (see Appendix A)

$$\mathbf{x}_{i-\mathbf{k}} = \tilde{\mathbf{x}}_{i-\mathbf{k}} - \overline{\mathbf{v}}_{i-\mathbf{k}-1} ,$$

we have

$$\mathbf{v}^{*}_{i} = \sum_{k=0}^{j-1} \mathbf{b}_{i,i-k} \mathbf{E}(\tilde{\mathbf{x}}_{i-k} | \mathbf{y}^{i}) - \sum_{k=0}^{j-1} \overline{\mathbf{v}}_{i-k-1},$$

where the \overline{v} 's are defined in terms of the preceding v*'s as in Appendix A and the expected values of the \hat{x} 's are given by (4.39) through (4.42).

4.7. Conclusion

In this chapter we have studied the problem of simultaneously designing communication strategies and control policies for problems involving ARMA models of orders higher than one. For one of the simplest such problems, involving an ARMA model of order 2, we have shown that the optimum linear strategy may be outperformed by an appropriately chosen nonlinear policy. This is done by relating the problem involving the second-order ARMA model, to a problem involving transmission through a Gaussian channel with noisy side information at the decoder. It has further been shown that over the affine class, the optimal strategy consists of transmitting the innovation at each stage.

2000

The solutions which are optimal over the affine class have been studied for the hard power constraint version as well as for the soft power constraint version of the general ARMA model.

1352

8

8

5.3

ĸы

8

EJ,

88

R

N^a

8

75

23

8

5ð

¥.?

THE DECENTRALIZED TWO-PERSON TEAM WITH MULTIPLE INFORMATION CHANNELS

5.1. Introduction

5

8

X X

88

8

X A

Ŋ

Z

23

In this chapter we consider decentralized, two-person stochastic team problems, where the action of one agent is transmitted to the other agent through a number of noisy channels simultaneously. This is a generalized version of the decentralized two-person team discussed in Chapter 2, where only a single communication channel was allowed between the two agents. • <u>2002.00</u> • <u>2002.00</u> • <u>2000.00</u> • <u>2000.00</u> • <u>2000.00</u>

522222 • FAA1122 •

In Section 5.2 we formalize the problems to be analyzed in this chapter. In Section 5.3 we consider the situation where the channel noises are independent of the input variable to be transmitted, with the observation of the first agent being noise corrupted in general. For this class we show that the optimal strategies are linear and may be found through a related parameter optimization problem. In Section 5.5 we analyze problems where the channel noise is correlated with the input variable to be transmitted and find that the strategies which are optimal over the linear class may be outperformed by non-linear strategies (except for a very restrictive subclass), even when the first agent observes an uncorrupted version of the input. The concluding remarks of Section 5.4 then end this chapter.

5.2. Problem Formulation

We have noted in Chapter 2 that if the first agent observes a noise corrupted version of the variable to be transmitted, and the channel noise is also correlated with this variable, then the linear strategies for the general stochastic team problem may be outper-

3 Х Ал Œ. · W 8 3 . . Ŗ č, Ň Ě T N N ${\boldsymbol{\Sigma}}^{1}$ Ś.

5

100

formed by nonlinear strategies. In view of this result, we shall restrict ourselves to the following two classes of problems with multiple information channels:

(i) The channel noises are all independent of the input x, while the first agent observes a garbled version of the input x.

(ii) The first agent observes the input x directly, and the channel noises are allowed to be correlated with the input x.

These situations are depicted in Figures 5.1 and 5.2, respectively. We consider the quadratic cost functional $J(\gamma_0, \gamma_1)$ where

$$J(\gamma_0, \gamma_1) = E[k_0 u_0^2 + s_0 u_0 x + s u_1^2 + s_1 u_1 x]$$
$$u_0 = \gamma_0(\cdot), u_1 = \gamma_1(\cdot),$$

and the following two classes of problems are thus identified:

Problem PI

$$\underset{\gamma_{0},\gamma_{1}}{\text{Minimize } J(\gamma_{0},\gamma_{1})}$$

where

$$\mathbf{y} = \begin{pmatrix} \lambda_1 \mathbf{u}_0 + \mathbf{w}_1 \\ \vdots \\ \lambda_n \mathbf{u}_0 + \mathbf{w}_n \end{pmatrix}$$

$$u_0 = \gamma_0(z)$$
$$u_1 = \gamma_1(y)$$
$$z = x + y$$

and



rei Dianesses Disenses (Derezzen) Diandesse (Disense

Figure 5.1. The decentralized two-person team, with channel noises independent of each other and of the input.

ě

Ş

5

52

N.

ζĊ,

 λ_{n}

0.000



correlated with the input.
Problem P2

 $\underset{\gamma_{0},\gamma_{1}}{\text{Minimize } J(\gamma_{0},\gamma_{1})}$

where

 $y = \begin{pmatrix} b_1 x + \lambda_1 u_0 + w_1 \\ b_2 x + \lambda_2 u_0 + w_2 \\ \vdots \\ b_n x + \lambda_n u_0 + w_n \end{pmatrix}$

 $u_0 = \gamma_0(x)$ $u_1 = \gamma_1(y)$

with x, v, w_1, \ldots, w_n being zero mean Gaussian random variables, independent of one another, having variances denoted by $\sigma_{(\cdot)}^2$, and with λ_i 's being given positive constants.

We now turn to analyzing the above classes of problems.

5.3. Instances with Optimal Linear Solutions

In this section we analyze Problem PI formulated in the previous section and show that the optimal strategies are linear. This problem has been studied in Bansal and Basar [1987b] and the analysis is given here for completeness. Towards obtaining the solution to PI we define

$$\mathbf{m} = \mathbf{E}[\mathbf{x} \mid \mathbf{z}],$$

make observations (i) through (iv) as in the case of the single channel problem (Section 2.3 (b)) and obtain the following equivalent problem:

$$\underset{\gamma'_{0},\gamma_{1}}{\text{Minimize } J'(\gamma'_{0},\gamma_{1})}$$

where

$$J'(\gamma'_{0},\gamma_{1}) = E[k_{0}u'_{0}^{2} + s_{0}u'_{0}m + \frac{s_{1}^{2}}{4s}(u_{1}-m)^{2} + c]$$

with

 $u'_{0} = \gamma'_{0}(m)$ $u_{1} = \gamma_{1}(y)$

and c is a constant independent of u'_0 and u_1 .

Let J_p denote the infimum of J under the constraint $E[u_o^{2}] = P^2$, i.e.,

$$J_{p} = \inf_{\substack{\gamma_{o}, \gamma_{1}; E[\gamma_{o}^{2}] = P^{2}}} J(\gamma_{o}, \gamma_{1}) \quad .$$
(5.1)

0000000

1000001141

We first have

$$J_{p} \ge k_{o}P^{2} + \inf_{E[\gamma_{o}^{2}]=P^{2}} \sup_{E[\gamma_{o}^{2}]=P^{2}} E[u_{0}m] + \inf_{E[\gamma_{o}^{2}]=P^{2}} E[u_{1}-m]^{2} + c.$$
(5.2)

By Cauchy-Schwartz inequality, we know that

$$\inf_{\mathbf{E}[\mathbf{u}_{o}^{2}]=\mathbf{P}^{2}} \mathbf{s}_{o} \mathbf{E}[\mathbf{u}_{o}^{\prime}\mathbf{m}] = -|\mathbf{s}_{o}| \mathbf{P}\boldsymbol{\sigma}_{\mathbf{m}} .$$

$$(5.3)$$

We now consider the optimization problem $\inf_{E[\gamma_0^{-2}]=P^2} E(u_1-m)^2$. Since m, \overline{y} and u_1 form a

Markov chain, we have

$$I\{m;u_1\} \leq I\{m;\overline{y}\} \tag{5.4}$$

where $I\{a;b\}$ denotes the mutual information of two random variables a and b.

Also, we have the inequality

$$I\{m;u_1\} \ge \frac{1}{2} \log \frac{\sigma_m^2}{E\{(u_1 - m)^2\}}$$
(5.5)

(Wyner [1970]), and the equality

$$I\{m;\overline{y}\} = H\{\overline{y}\} - H\{\overline{y} \mid m\}$$
(5.6)

where $H\{a\}$ is the entropy of the random variable a, and $H\{a \mid b\}$ is the conditional entropy of the random variable a given b.

The correlation matrix for vector \overline{y} , with $E[u'_o^2] = P^2$, is

$$\mathsf{E}_{\overline{y}} = \begin{bmatrix} \lambda_1^2 \mathbf{P}^2 + \sigma_{\mathbf{w}_1}^2 & \cdots & \lambda_1 \lambda_n \mathbf{P}^2 \\ \\ \lambda_n \lambda_1 \mathbf{P}^2 & \cdots & \lambda_n^2 \mathbf{P}^2 + \sigma_{\mathbf{w}_n}^2 \end{bmatrix}$$

which has the determinant

$$|\mathbf{C}_{\bar{\mathbf{y}}}| = \prod_{i=1}^{n} \sigma_{\mathbf{w}_{i}}^{2} (1 + P^{2} \sum_{i} \frac{\lambda_{i}^{2}}{\sigma_{\mathbf{w}_{i}}^{2}}) .$$
(5.7)

Since for any n-variate random vector with a fixed mean and a fixed covariance matrix, the maximum entropy is attained by a normal distribution (see, for example, Kagan et al. [1973]), we have

$$H\{\overline{y}\} \leq \frac{1}{2} \log(2\pi e)^{n} | \mathbf{t}_{\overline{z}}| = \frac{1}{2} \log(2\pi e)^{n} \prod_{i=1}^{n} \sigma_{w_{i}}^{2} (1+P^{2}\lambda)$$
(5.8)

where $\lambda \leq \sum_{i} \frac{\lambda_{i}^{2}}{\sigma_{w_{i}}^{2}}$.

Using (5.8) in (5.6) we get

, ° i 1 i

RX X

88

 $\overline{\mathcal{X}}$

¥,

Ϋ́,

X

$$I\{m;\overline{y}\} \leq \frac{1}{2} \log(1 + P^2 \sum_{i} \frac{\lambda_i^2}{\sigma_{w_i}^2})$$
(5.9)

where we have made use of the expression

$$H\{\overline{y} \mid m\} = \frac{1}{2} \log(2\pi e)^n \prod_{i=1}^n \sigma_{w_i}^2,$$

since for fixed γ_0 , the vector \overline{y} given m consists of n independent Gaussian random variables.

Using (5.5) and (5.9) in (5.4) we get

$$E\{(u_{1}-m)^{2}\} \ge \frac{\sigma_{m}^{2}}{1+P^{2}\lambda} .$$
(5.10)
$$E[\gamma_{0}^{2}]=P^{2} \qquad 1+P^{2}\lambda$$

It follows from (5.2) (using (5.3) and (5.10)) that

$$J_{p} \ge k_{o}P^{2} - |s_{o}|P\sigma_{m} + \frac{\sigma_{m}^{2}}{(1+P^{2}\lambda)^{2}} + c$$

$$\ge Min[k_{o}P^{2} - |s_{o}|P\sigma_{m} + \frac{\sigma_{m}^{2}}{1+P^{2}\lambda} + c]$$

$$= k_{o}P^{*2} - |s_{o}|P^{*}\sigma_{m} + \frac{\sigma_{m}^{2}}{1+P^{*2}\lambda} + c$$
(5.11)

where

$$P^* = \operatorname{Arg} \operatorname{Min}_{P} [k_0 P^2 - |s_0| P \sigma_m + \frac{\sigma_m^2}{1 + P^2 \lambda}] > 0.$$

Note that P* necessarily exists, since at P=0 the function to be minimized is decreasing and as $P \rightarrow \infty$, $J_p \rightarrow \infty$, implying that the search can be confined to a closed bounded region of \mathbb{R}^1 , over which a continuous function always admits a minimum. Taking derivatives, we find

105

that the value $P=P^*$ which attains this minimum satisfies

$$(2k_{o}P^{*} - |s_{o}|\sigma_{m})(P^{*} + \frac{1}{\lambda})^{2} = \frac{2P^{*}\sigma_{m}^{2}}{\lambda} .$$
 (5.12)

We now have

$$J_{opt} \stackrel{\Delta}{=} \inf_{\substack{\gamma_o, \gamma_1 \\ \gamma_o, \gamma_1}} J(\gamma_o, \gamma_1) = \inf_{\substack{P \ge 0}} J_p$$

$$\geq k_o P^{*2} - (|s_o| \sigma_m) P^* + \frac{\sigma_m^2}{1 + P^{*2} \lambda} + c$$
(5.13)

which gives us a lower bound for the infimum of J. The final task is to note that this lower bound is tight and is attained by linear strategies. We thus have Theorem 5.1 below.

Theorem 5.1.

(i) The stochastic team problem P1 formulated in Section 5.2 admits an optimal solution which is linear in the observation variables, and is given by

$$\gamma_{0}^{*}(z) = \beta_{z}^{*}z$$

$$\gamma_{1}^{*}(y) = -\frac{s_{1}}{2s} E[x | y]$$

$$= -\frac{s_{1}}{2s} \frac{\sigma_{x}^{2}}{\sigma_{x}^{2} + \sigma_{y}^{2}} \sum_{i=1}^{n} \frac{\lambda_{i}\beta^{*}}{\sigma_{w}^{2}[(1/(\sigma_{x}^{2} + \sigma_{y}^{2})) + \beta^{*}\lambda]} y$$

where β^* is given by the solution to the following parameter optimization problem:

$$\beta^{*} = \arg \min_{\beta} \left[k_{0}\beta^{2}(\sigma_{x}^{2} + \sigma_{y}^{2}) + s_{0}\beta\sigma_{x}^{2} + \frac{s_{1}^{2}}{4s} \cdot \frac{\sigma_{x}^{4}}{(\sigma_{x}^{2} + \sigma_{y}^{2})(1 + \beta^{2}\lambda(\sigma_{x}^{2} + \sigma_{y}^{2}))} \right]$$

a solution to which may always be found as discussed earlier.

107

(ii) The optimal value of the cost is

X

8

8

8

8

8

N.

8

2

 $\widetilde{\mathcal{X}}$

10

$$J_{opt} = k_0 \beta^{*2} (\sigma_x^2 + \sigma_v^2) + s_0 \beta^* \sigma_x^2 + \frac{s_1^2 \sigma_x^4}{4s(\sigma_x^2 + \sigma_v^2)(1 + \beta^{*2}\lambda(\sigma_x^2 + \sigma_v^2))} + \frac{s_1^2}{4s} \frac{\sigma_x^2 \sigma_v^2}{\sigma_x^2 + \sigma_v^2} - \frac{s_1^2}{4s} \sigma_x^2.$$

5.4. Nonoptimality of Linear Strategies

In this section we analyze problems where the channel noises are correlated with the input variable x which is to be transmitted through the channel. We show that except for a very restrictive subclass of such problems, the strategies which are optimal over the linear class may be outperformed by appropriately chosen nonlinear strategies.

Since λ_i 's are assumed to be nonzero (otherwise the channel is redundant and may be removed) the observation y for Problem P2 is equivalent to the following observation y',

$$\mathbf{y}' = \begin{pmatrix} (\mathbf{b}_1/\lambda_1) \mathbf{x} + \mathbf{u}_0 + (\mathbf{w}_1/\lambda_1) \\ \vdots \\ (\mathbf{b}_n/\lambda_n) \mathbf{x} + \mathbf{u}_0 + (\mathbf{w}_n/\lambda_n) \end{pmatrix}.$$

Lemma 5.1. If $b_1/\lambda_1 = b_2/\lambda_2 = \cdots = b_n/\lambda_n$, then the search for optimal strategies for Problem P2 may be confined to the linear class.

Proof. Under the conditions of the lemma we may define

$$\mathbf{u'_0} = \mathbf{u_0} + \frac{\mathbf{b_1}}{\lambda_1} \mathbf{x}$$

to obtain the equivalent stochastic team problem below:

Minimize
$$E[k_0 u'_0^2 + s'_0 u'_0 x + su_1^2 + s_1 u_1 x]$$

 $y'_0 y_1$

 $u'_0 = \gamma'_0(x)$ $u_1 = \gamma_1(y)$

where

and

y =	$\lambda_1 u_0 + w_1$
	•
	•
	J •
	$\lambda_n u'_0 + w_n$

with

$$\mathbf{s'}_0 = \mathbf{s}_0 - 2\mathbf{k}_0 \frac{\mathbf{b}_1}{\lambda_1}$$

We thus obtain a special case of Problem P1 with $\sigma_v^2 = 0$, for which the optimality of linear strategies has been established in Section 5.3.

We now return to the general Problem P2 without the restriction of equality on the b_i/λ_i 's, and show that, for this class, nonlinear strategies may outperform the optimal linear strategies.

In the following we shall, for notational convenience, assume that $\lambda_i = 1$ for all i. This does not introduce any loss of generality since a redefinition of the b_i 's and the noise variances yield the original problem.

It is sufficient to establish nonoptimality of linear laws for a special two-channel case, which we study next.

Problem $P2^2$

Minimize
$$J_2(\gamma_0, \gamma_1) = E[k_0 u_0^2 + (u_1 - x)^2]$$

 $u_0 = \gamma_0(x)$ $u_1 = \gamma_1(y)$

where

and

$$\mathbf{y} = \begin{pmatrix} \mathbf{y}_1 \\ \mathbf{y}_2 \end{pmatrix} = \begin{pmatrix} \mathbf{u}_0 + \mathbf{b}\mathbf{x} + \mathbf{w}_1 \\ \mathbf{u}_0 - \mathbf{b}\mathbf{x} + \mathbf{w}_2 \end{pmatrix}$$

with b>0 and $\sigma_{w_1}^2 = \sigma_{w_2}^2$.

Lemma 5.2. Problem $P2^2$ is equivalent to the soft constraint version of the Gaussian Channel Problem with noisy side information at the decoder, studied in Appendix B.

Proof. We need to note that the vector observation y is equivalent to observing

$$\begin{pmatrix} \frac{y_1 + y_2}{2} \\ \frac{y_1 - y_2}{2} \\ \frac{y_1 - y_2}{2} \end{pmatrix} = \begin{pmatrix} u_0 + \frac{w_1 + w_2}{2} \\ bx + \frac{w_1 - w_2}{2} \\ \frac{w_1 - w_2}{2} \end{pmatrix}$$

and the lemma is established.

D

Property Property

XXXXXXXXX

PALITICICAL V

Since linear strategies are not optimal for the problem with noisy side information, in view of Lemma 5.2 the linear strategies are not optimal for Problem $P2^2$.

2:X d-3 8 Š **1** R ĝ. 10 8 15 S S Č. Š ų, 3 57

E

Problem P2^{2G}

Minimize
$$J_2(\gamma_0, \gamma_1) = E[k_0 u_0^2 + s_0 u_0 x + (u_1 - x)^2]$$

where

 $u_0 = \gamma_0(x)$ $u_1 = \gamma_1(y)$

and

$\mathbf{y} = \begin{pmatrix} \mathbf{u}_0 + \mathbf{b}_1 \mathbf{x} + \mathbf{w}_1 \\ \\ \mathbf{u}_0 + \mathbf{b}_2 \mathbf{x} + \mathbf{w}_2 \end{pmatrix}$

with $b_1 > b_2$.

We let

to obtain the equivalent cost functional

$$J'_{2}(\gamma'_{0},\gamma_{1}) = E[k_{0}u'_{0}^{2} + s'_{0}u'_{0}x + (u_{1}-x)^{2}]$$

where

 $u'_0 = \gamma'_0(x)$ $u_1 = \gamma_1(y)$

and

 $u'_0 = u_0 + \frac{b_1 + b_2}{2} x$

 $\mathbf{y} = \begin{pmatrix} \mathbf{u}_0 + \mathbf{b}\mathbf{x} + \mathbf{w}_1 \\ \mathbf{u}_0 - \mathbf{b}\mathbf{x} + \mathbf{w}_2 \end{pmatrix}$

with

8

X

<u>88</u>

Ŕ

8

Ż

33

X

 Σ

 $b = \frac{b_1 - b_2}{2}$

and

$$s'_0 = s_0 - k_0(b_1 + b_2)$$
.

The transformed problem may now be viewed as one involving noisy side information at the decoder (cf. Lemma 5.2), and the nonoptimality of linear strategies follows.

5.5. Conclusion

In this chapter we have studied decentralized two-person stochastic team problems with multiple communication channels. For the case when all channel noises are independent of the input variable to be transmitted, we have established the optimality of linear strategies. However, when the channel noise is correlated with the input variable we find that linear strategies are in general not optimal for the case with multiple channels, even when the first agent observes an uncorrupted version of the variable to be transmitted.

The problems in this chapter may be viewed as those involving encoding and transmitting over vector channels with Gaussian noises (under generalized fidelity criteria), where the source output may be distorted prior to encoding or the channel noises may be correlated with the source output.

The classical information transmission problem, viewed as a team problem, generally assumes that one can directly encode the variable to be recovered at the receiving end. A

83 Ś UD. 8 32 8 3 N Ä 2 S S Ţ Å 8

77

112

model which allows distortion prior to transmission was first considered by Dobrushin and Tysbakov [1962]. If we suppose that the message to be transmitted is a temperature read by a digital thermometer or the pixel levels of an image, then the message would not be an exact copy of the object of interest but a noise-bearing variant of it, and the analysis in this chapter would then be applicable.

We also find that for the case where the channel noise is uncorrelated with the encoder input, the nature of the solution is such that it may be considered as first extracting the message from its noisy version under a mean square criterion and then transmitting this extracted message. This is in accordance with the scheme reported for the hard power constraint version with n-1 by Wolf and Ziv [1970], as well as with the "disconnection principle" introduced for finite alphabets by Witsenhausen [1980].

We should also note that channels of the form considered in this chapter have been called multipath channels in the literature (Ovseevich and Pinsker [1958]). In such channels, even though there is a single transmitter, the reception is as though a number of channels are operative in parallel. Pinsker [1972] also gives expressions and estimates for the quantity of information contained in observations with respect to an estimated parameter for a fixed and random number of observations. For the special case where each of the λ_i 's are assumed to be unity, our result on maximum information between m and \overline{y} (eqn. (3.9)) corresponds to the expression derived in Pinsker [1972], when the number of observations is fixed, and the variables are all Gaussian. Pinsker's approach, however, is based on sufficient statistics, whereas here we have employed known results on entropy maximization.

CHAPTER 6

DECENTRALIZED MULTIPERSON TEAMS

6.1. Introduction

X

5 . 1 .

X

8

88

X

ß

8

R

8

8

10

In this chapter we formulate and analyze some decentralized, multiperson, stochastic team problems which are generalizations of the two-person teams studied in Chapter 2. It is assumed that a (possibly noisy) version of a Gaussian random variable, available at one location, is used to generate either a single communication signal or a set of communication signals, which may then be transmitted via noise-corrupted channels to either a single location or to a set of locations.

In Section 6.2 we formulate the problems to be analyzed in this chapter. In Section 6.3 we analyze problems with a single transmitting agent, (synonymously, encoder) and multiple receiving agents (synonymously, decoders). In Section 6.4 we analyze problems with multiple transmitting agents as well as multiple receiving agents. In Section 6.5 we consider problems with multiple transmitting agents and a single receiving agent. The concluding remarks in Section 6.6 then end this chapter.

6.2. Problem Formulation

In this section we formulate three classes of problems involving decentralized multiperson teams, where communication between agents is permitted only via noisecorrupted channels.

First, consider the situation depicted in Figure 6.1, where the measurement variable u_0 is generated based on a (possibly noisy) observation of a Gaussian random variable x, and is then transmitted over a number of noisy channels to various locations. We allow the possibility of controlling the sign of transmission over each of the channels

individually, and the channel noises may, in general, be correlated with the input variable x. All agents are assumed to cooperate in minimizing a quadratic cost functional, and we formally have Problem P1 below:





Problem P1

Minimize
$$J_1(\gamma_0;\gamma_{11},...,\gamma_{1N};s_1,...,s_N) = E[k_0u_0^2 + r_0u_0x + \sum_{i=1}^{N} p_iu_{1i}^2 + q_iu_{1i}x]$$

where

$$u_0 = \gamma_0(z)$$
$$u_{1i} = \gamma_{1i}(y_i)$$

with

 \hat{J}

3.5

91

J.V

ŝ

3

 $\tilde{\lambda}_{j}$

99 22

8

Зð

Š

 \overline{Y}

-> :{

М Х

 \tilde{X}

X

Ś

5

C.C.C.C.C.C.

X

8

3

X

Ř

26 20

ŝ

3

Ř

<u>г</u>. У

z = x + v $y_i = s_i u_0 + b_i x + w_i.$

Here k_0, p_1, \ldots, p_N are given positive constants, r_0 , q_i 's and b_i 's are arbitrary constants and x, v, w_1, \ldots, w_N are independent Gaussian random variables, with zero mean and prescribed variances. The s_i 's (i=1,...,N) are design variables, constrained to be either +1 or -1; these control the sign of transmission over each of the channels individually.

Note that Problem P1 involves the design of only a single communicating agent, and it is only the sign of transmission on different channels that may be controlled. In case we have complete freedom of designing the communication variable for each of the channels individually, we obtain the situation depicted in Figure 6.2, which leads to Problem P2 below:



Figure 6.2. The decentralized stochastic team with multiple transmitting as well as multiple receiving agents.

Problem P2

 $\underset{\gamma_{01}, \dots, \gamma_{0N}; \gamma_{11}, \dots, \gamma_{1N}}{\text{Minimize}} J_2(\gamma_{01}, \dots, \gamma_{0N}; \gamma_{11}, \dots, \gamma_{1N}) = E[\sum_{i=1}^{N} k_i u_{0i}^2 + r_i u_{0i} x + p_i u_{1i}^2 + q_i u_{1i} x]$

where

 $u_{0i} = \gamma_{0i}(z)$ $u_{1i} = \gamma_{1i}(y_i)$

with

z = x + v $y_i = u_{0i} + b_i + w_i.$

We finally consider the situation shown in Figure 6.3, where multiple communication agents are permitted and noise corrupted versions of all their outputs are made available at a single location. This formally leads to Problem P3 below.



Figure 6.3. The decentralized stochastic team with multiple transmitting agents and a single receiving agent.

Problem P3

Minimize
$$J_3(\gamma_{01}, ..., \gamma_{0N}; \gamma_1) = E[\sum_{i=1}^{N} (k_i u_{0i}^2 + r_i u_{0i} x) + p_1 u_1^2 + q_1 u_1 x]$$

where

 $u_{0i} = \gamma_{0i}(z)$ $u_1 = \gamma_1(y_1, \dots, y_N)$

with

$$z = x + v$$

$$y_i = u_{0i} + b_i x + w_i.$$

The three classes of problems identified in this chapter may all be viewed as generalizations of the decentralized two-person team problem studied in Chapter 2, where it was established that linear strategies are optimal if either one of the following two conditions holds: (a) the channel noise is uncorrelated with the input, or (b) an uncorrupted version of x is available at the encoder. Therefore, in order to study how the results for the two-agent problem carry over to the multiple-agent problems, we shall restrict ourselves to the following two conditions for each of the Problems P1, P2 and P3:

Case (a): $\sigma_v^2 = 0$, b_i's are arbitrary, and

Case (b): $\sigma_v^2 \neq 0$, $b_i = 0$ for all i=1,..., N.

In the following three sections we respectively analyze Problems P1, P2 and P3. under conditions (a) and (b) above.

4.9. A.S. 448. 11

6.3. The Problem with a Single Encoder and Multiple Decoders

In this section we study Problem P1 identified in Section 6.2. By completing squares, we find that the cost function J_1 may be rewritten as

$$J_{1}(\gamma_{0};\gamma_{11},...,\gamma_{1N}) = E[k_{0}u_{0}^{2} + r_{0}u_{0}x + \sum_{i=1}^{N} p_{i}(u_{1i} + \frac{q_{i}}{2p_{i}}x)^{2} - \sum_{i=1}^{N} \frac{q_{i}^{2}}{4p_{i}}\sigma_{x}^{2}]$$
(6.1)

Case (a): $\sigma_v^2 = 0$, b's are arbitrary.

If $\sigma_v^2 = 0$, we may define

$$u_{0i} = s_i u_0 + b_i x$$
, (6.2)

and then consider the problem of minimizing

$$E[(u_{1i} + \frac{q_i}{2p_i} x)^2]$$
 (6.3)

subject to

$$u_{1i} = \gamma_{1i}(u_{0i} + w_i)$$
. (6.4)

Note that for every fixed $E[u_0^2] = P^2$, we have

$$E[u_{0i}^{2}] = P^{2} + b_{i}^{2} \sigma_{x}^{2} + E[2s_{i}b_{i}u_{0}x]$$
(6.5)

$$\leq P^2 + b_i^2 \sigma_x^2 + 2 + b_i P \sigma_x + .$$

Since

$$\mathbf{y}_{i} = \mathbf{u}_{0i} + \mathbf{w}_{i} \tag{6.6}$$

<u>7</u>7

i Si 55 8 525 Š ŝ Į.

8

Ŗ

2

K.

119

it follows using (2.10) and (2.11) that

$$I(x;y_{i}) \leq \frac{1}{2} \log \left(\frac{P^{2} + b_{i}^{2} \sigma_{x}^{2} + 2 |b_{i} P \sigma_{x}| + \sigma_{w_{i}}^{2}}{\sigma_{w_{i}}^{2}} \right)$$
(6.7)

and further, using (2.9), we obtain that

$$E[(u_{1i} + \frac{q_i}{2p_i}x)^2] \ge \frac{q_i^2}{4p_i^2} \cdot \frac{\sigma_x^2 \sigma_{w_i}^2}{((|P| + |b_i| \sigma_x)^2 + \sigma_{w_i}^2)}.$$
 (6.8)

Let J_{1P} denote the minimum of J_1 under the hard power constraint $E[u_0^2] = P^2$. We

then have (with
$$K \Delta - \sum_{i=1}^{N} (q_i^2/4p_i)\sigma_x^2)$$

$$J_{1P} \ge k_0 P^2 + \inf_{E[u_0^2] = P^2} E[r_0 u_0 x] + \inf_{E[u_0^2] = P^2} E[\sum_{i=1}^N p_i (u_{1i} + \frac{q_i}{2p_i} x)^2] + K$$

= $k_0 P^2 - |r_0 P \sigma_x| + \sum_{i=1}^N \frac{q_i^2}{4p_i} \cdot \frac{\sigma_x^2 \sigma_{w_i}^2}{(|P| + |b_i| \sigma_x)^2 + \sigma_{w_i}^2} + K$

$$\geq \min_{\mathbf{P}} \left| k_0 \mathbf{P}^2 - |\mathbf{r}_0 \mathbf{P} \sigma_{\mathbf{x}}| + \sum_{i=1}^{N} \frac{\mathbf{q}_i^2}{4p_i} \cdot \frac{\sigma_{\mathbf{x}}^2 \sigma_{\mathbf{w}_i}^2}{(|\mathbf{P}| + |\mathbf{b}_i| \sigma_{\mathbf{x}})^2 + \sigma_{\mathbf{w}_i}^2} + K \right|$$
(6.9)

$$= k_0 P^{*2} - |r_0 P^* \sigma_x| + \sum_{i=1}^{N} \frac{q_i^2}{4p_i} \cdot \frac{\sigma_x^2 \sigma_{w_i}^2}{(|P^*| + |b_i| \sigma_x)^2 + \sigma_{w_i}^2} + K.$$

We thus obtain a lower bound on the optimal cost, and the next task is to note that this lower bound is tight and is achieved by using the linear policies:

$$u_{0}^{*} = \gamma_{0}^{*}(x) = -\operatorname{sgn} r_{0} \cdot \frac{|P^{*}|}{\sigma_{x}} x$$
 (6.10a)

$$u_{1i}^{*} = \gamma_{1i}^{*}(y_{i}) = -\frac{q_{i}}{2p_{i}} \frac{s_{i}(|P^{*}| + |b_{i}|\sigma_{x})\sigma_{x}}{((|P^{*}| + |b_{i}|\sigma_{x})^{2} + \sigma_{w_{i}}^{2})} y_{i}$$
(6.10b)

where

$$s_i = \operatorname{sgn} b_i . \tag{6.10c}$$

 $\hat{\mathbf{M}}$

ģ

50

٩7

3

R

S.

22

8

8

Ø

8

Q,

έł β

Ę,

13. 20

-2

We therefore have Theorem 6.1 below.

Theorem 6.1.

(i) The stochastic team problem P1 with $\sigma_v^2 = 0$ admits an optimal solution which is linear in the observation variables, and is given by (6.10), where P* is given by the solution to the following parameter optimization problem,

$$P^* = \arg\min_{P} [k_0 P^2 + r_0 P \sigma_x + \sum_{i=1}^{N} \frac{q_i^2}{4p_i} \frac{\sigma_x^2 \sigma_{w_i}^2}{((|P| + |b_i| \sigma_x)^2 + \sigma_{w_i}^2)}]$$

which always admits a solution.

(ii) The optimal value of the cost is

$$J_{1}^{*} = k_{0}P^{*2} - r_{0}P^{*}\sigma_{x} + \sum_{i=1}^{N} \frac{q_{i}^{2}}{4p_{i}} \frac{\sigma_{x}^{2}\sigma_{w_{i}}^{2}}{((|P^{*}| + |b_{i}|\sigma_{x})^{2} + \sigma_{w_{i}}^{2})} - \sum_{i=1}^{N} \frac{q_{i}^{2}}{4p_{i}} \sigma_{x}^{2}$$

Case (b): $b_i = 0$ for all i, σ_v^2 is arbitrary.

8

88

X

8

8

5

Ϋ́ς

X

3

ŝ

λ.

Š

If b_i 's are zero for all i, then all channel noises are independent of the input x. We now define

$m \triangleq E(x | z)$

and use observations (ii) through (iv) of Section (2.3) to obtain the equivalent problem below:

$$\begin{array}{l} \text{Minimize}_{\gamma'_{0};\gamma'_{11},\ldots,\gamma'_{1N}} \mathbb{E}[k_{0}{u'_{0}}^{2} + r_{0}{u'_{0}}m + \sum_{i=1}^{N} \frac{q_{i}^{2}}{4p_{i}}(u'_{1i} - m)^{2} + K] \\ u'_{0} = \gamma'_{0}(m) \\ u'_{1i} = \gamma'_{1i}(y_{i}) \end{array}$$
(6.11)

and K is a constant independent of u'_{0} , u'_{11} , ..., u'_{1N} . We thus obtain a special case of the problem discussed in part (a) of this section, and the optimality of linear strategies follows. Note that the s_i's in this case may be chosen arbitrarily, since the b_i's are all zero; hence in the statement of the following theorem we have taken $s_i = +1$ for all i, without any loss of generality.

Theorem 6.2.

(i) The stochastic team problem P1, with $b_i = 0$ for all i, admits an optimal solution which is linear in the observation variables and is given by

$$u_0^* = \lambda^* z \tag{6.12a}$$

$$u_{1i}^{*} = -\frac{q_{i}}{2p_{i}} \left(\frac{\lambda^{*} \sigma_{x}^{2}}{\lambda^{*2} (\sigma_{x}^{2} + \sigma_{y}^{2}) + \sigma_{w_{i}}^{2}} \right) y_{i}$$
(6.12b)

Ś

Ŷ

32

30

5

ğ

8

Ъ,

X

Ę

00 2

, S

Ç,

ų,

where λ^* is given by the solution to the following parameter optimization problem:

$$\lambda^* = \arg\min_{\lambda} [k_0 \lambda^2 (\sigma_x^2 + \sigma_v^2) + r_0 \lambda \sigma_x^2 + \sum_{mi=1}^{N} \frac{q_i^2}{4p_i} \frac{\sigma_x^2 (\lambda^2 \sigma_v^2 + \sigma_{w_i}^2)}{(\lambda^2 (\sigma_x^2 + \sigma_v^2) + \sigma_{w_i}^2)}]$$

a solution to which always exists (Remark 2.1).

(ii) The optimal value for the cost is

$$k_{0}\lambda^{*2}(\sigma_{x}^{2}+\sigma_{v}^{2})+r_{0}\lambda^{*}\sigma_{x}^{2}+\sum_{i=1}^{N}\frac{q_{i}^{2}}{4p_{i}}\frac{\sigma_{x}^{2}(\lambda^{*2}\sigma_{v}^{2}+\sigma_{w_{i}}^{2})}{(\lambda^{*2}(\sigma_{x}^{2}+\sigma_{v}^{2})+\sigma_{w_{i}}^{2})}-\sum_{i=1}^{N}\frac{q_{i}^{2}}{4p_{i}}\sigma_{x}^{2}.$$

6.4. The Problem with Multiple Encoders and Multiple Decoders

In this section we analyze Problem P2 identified in Section 6.2. First, by completing squares, we find that the cost functional J_2 may be rewritten as

$$J_{2}(\gamma_{01}, \dots, \gamma_{0N}; \gamma_{11}, \dots, \gamma_{1N}) = E[\sum_{i=1}^{N} (k_{i} u_{0i}^{2} + r_{i} u_{0i} x + p_{i} (u_{1i} + \frac{q_{i}}{2p_{i}} x)^{2} - \frac{q_{i}^{2}}{4p_{i}} \sigma_{x}^{2}). \quad (6.13)$$

Case (a): $\sigma_v^2 = 0$, b_i 's are arbitrary.

The analysis is similar to that in Section 2.3. We first define

$$u'_{0i} = u_{0i} + b_i x$$
 (6.14)

to obtain the equivalent cost functional

$$J'_{2}(\gamma'_{01}, \ldots, \gamma'_{0N}; \gamma'_{11}, \ldots, \gamma'_{1N}) = E[\sum_{i=1}^{N} k_{i} u'_{0i}^{2} + r'_{i} u'_{0i} x + \frac{q_{i}^{2}}{4p_{i}} (u'_{1i} - x)^{2} + K]$$

,

123

where

$$u'_{0i} = \gamma'_{0i}(x)$$
 (6.15a)

$$u'_{1i} = \gamma'_{1i}(u'_{0i} + w_i)$$
 (6.15b)

$$\mathbf{r}'_i = \mathbf{r}_i - 2\mathbf{k}_i \mathbf{b}_i \tag{6.15c}$$

and K is a constant independent of $u'_{01},..,u'_{0N},u'_{11},\ldots,u'_{1N}$.

Let J'_{2P} denote the minimum of $J'(\gamma'_0,\gamma'_1)$ under the hard power constraints

$$E[u'_{0i}^{2}] = P_{i}^{2}.$$
 (6.16)

We then have

$$J'_{2P} \ge \sum_{i=1}^{N} (k_i P_i^2 + \inf_{E[u'_{0}]_{2}^{i} = P_i^2} E[r'_{i}u'_{0i}x] + \inf_{E[u'_{0}]_{2}^{i} = P_i^2} E[\frac{q_i^2}{4p_i}(u'_{1i} - x)^2]) + K$$

$$= \sum_{i=1}^{N} (k_i P_i^2 - |r'_i| P_i \sigma_x + \frac{q_i^2}{4p_i} \frac{\sigma_x^2 \sigma_{w_i}^2}{(P_i^2 + \sigma_{w_i}^2)}) + K$$

$$\ge \min_{P_i \ge 0} \sum_{i=1}^{N} (k_i P_i^2 - |r'_i| P_i \sigma_x + \frac{q_i^2}{4p_i} \frac{\sigma_x^2 \sigma_{w_i}^2}{(P_i^2 + \sigma_{w_i}^2)}) + K]$$

$$= \sum_{i=1}^{N} (k_i P_i^{*2} - |r'_i| P_i^{*} \sigma_x + \frac{q_i^2}{4p_i} \frac{\sigma_x^2 \sigma_{w_i}^2}{(P_i^{*2} + \sigma_{w_i}^2)}) + K.$$
(6.17)

Noting that this bound is tight and is attained by linear policies, we have the following theorem:

Theorem 6.3.

•

CONTRACT SECTOR OF SECTOR SECTOR

(i) The stochastic team problem P2 with $\sigma_v^2 = 0$ admits an optimal solution which is linear in the observation variables and is given by

$$u^{*}_{0i} = \gamma^{*}_{0i}(x) = \lambda^{*}_{i}x$$
 (6.18a)

$$u_{1i}^{*} = \gamma_{1i}^{*}(y_{i}) = -\frac{q_{i}}{2p_{i}} \left[\frac{(\lambda_{i}^{*} + b_{i})\sigma_{x}^{2}}{(\lambda_{i}^{*} + b_{i})^{2}\sigma_{x}^{2} + \sigma_{w_{i}}^{2}} \right] y_{i}$$
(6.18b)

where

 $\lambda_{i}^{*} = \left(\frac{P_{i}^{*}}{\sigma_{x}} - b_{i}\right)$ (6.19)

and the P^*_{i} 's are found by solving the following parameter optimization problem:

$$P_{i}^{*} = \arg\min_{P_{i}} \left(k_{i} P_{i}^{2} + r'_{i} P_{i} \sigma_{x}^{2} + \frac{q_{i}^{2}}{4p_{i}} - \frac{\sigma_{x}^{2} \sigma_{w_{i}}^{2}}{(P_{i}^{2} + \sigma_{w_{i}}^{2})}\right)$$
(6.20)

a solution to which always exists (Remark 2.1).

(ii) The optimal value of the cost is

$$J_{2}^{*} = \sum_{i=1}^{N} (k_{i} P_{i}^{*2} - r_{i}' P_{i}^{*} \sigma_{x} + \frac{q_{i}^{2}}{4p_{i}} \frac{\sigma_{x}^{2} \sigma_{w_{i}}^{2}}{(P_{i}^{*2} + \sigma_{w_{i}}^{2})}) + K$$

where

$$K = \sum_{i=1}^{N} (-k_i b_i^2 + r_i b_i - \frac{q_i^2}{4p_i}) \sigma_x^2.$$

থ্য চ Case (b): $b_i = 0$ for all i, σ_v^2 is arbitrary.

If b_i 's are all zero, then the channel noises are all independent of the input x. We define

$m \triangleq E(x | z)$

and use observations (ii) through (iv) of Section 2.3 to obtain the equivalent problem below:

$$\underset{\gamma'_{01}, \dots, \gamma'_{0N}, \gamma'_{11}, \dots, \gamma'_{1N}}{\text{Minimize}} E[\sum_{i=1}^{N} (k_i u'_{0i}^2 + r_i u'_{0i}m + \frac{q_i^2}{4p_i} (u'_{1i} - m)^2) + K'$$

with

$$u'_{0i} = \gamma'_{0i}(m)$$
$$u'_{1i} = \gamma'_{1i}(y_i)$$

and K' is a constant independent of $u'_{01}, \ldots, u'_{0N}, u'_{11}, \ldots, u'_{1N}$. We thus obtain a special case of the problem discussed in part (a) of this section, and the optimality of linear strategies follows.

Theorem 6.4.

(i) The stochastic team problem P2, with $b_i = 0$ for all i, admits an optimal solution which is linear in the observation variable and is given by

$$u_{0i}^{*}(z) = \lambda_{i}^{*} z \qquad (6.21a)$$

$$u_{1i}^{*}(y_{i}) = -\frac{q_{i}}{2p_{i}} \left[\frac{\lambda_{i}^{*}\sigma_{x}^{2}}{\lambda_{i}^{*}(\sigma_{x}^{2} + \sigma_{y}^{2}) + \sigma_{w_{i}}^{2}} \right] y_{i}$$
(6.21b)

04054040404040404040404C46

where λ_{i}^{*} is given by the solution to the following parameter optimization problem:

A REACCOLOR A MANAGEMENTE A BEES

126

$$\Lambda^{*}_{i} = \arg \min_{\lambda_{i}} [k_{i}\lambda_{i}^{2}(\sigma_{x}^{2} + \sigma_{v}^{2}) + r_{i}\lambda_{i}\sigma_{x}^{2} + \frac{q_{i}^{2}}{4p_{i}} \frac{\sigma_{x}^{2}(\lambda_{i}^{2}\sigma_{v}^{2} + \sigma_{v_{i}}^{2})}{(\lambda_{i}^{2}(\sigma_{x}^{2} + \sigma_{v}^{2}) + \sigma_{w_{i}}^{2})}]$$

a solution to which always exists (Remark 2.1).

(ii) The optimal value of the cost is

$$\sum_{i=1}^{N} (k_i \lambda_i^* (\sigma_x^2 + \sigma_v^2) + r_i \lambda_i^* \sigma_x^2 + \frac{q_i^2}{4p_i} \frac{\sigma_x^2 (\lambda_i^* \sigma_v^2 + \sigma_{w_i}^2)}{(\lambda_i^* (\sigma_x^2 + \sigma_v^2) + \sigma_{w_i}^2)} - \frac{q_i^2}{4p_i} \sigma_x^2).$$

6.5. The Problem with Multiple Encoders and a Single Decoder

In this section we study Problem P3 identified in Section 6.2. We find that linear strategies are in general nonoptimal for this class of problems even under the restrictions that all channel noises are independent of the input and the encoder observes an uncorrupted version of the input.

In order to see why linear strategies are not optimal, we consider the special case with N=2 and $b_1=b_2=\sigma_v^2=0$, and impose the additional constraints

$$\mathrm{E}[\mathrm{u}_{1}^{2}] \leqslant \mathrm{P}_{1}^{2} \tag{6.22a}$$

$$E[u_2^2] \le P_2^2$$
. (6.22b)

We thus have a problem of constructing real-time encoding and decoding strategies for a communication system with hard power constraints where the source dimension is one and the channel dimension is two. For this problem it is known that linear strategies are not optimal (Shannon [1949], Wozencraft and Jacobs [1965]).

Our task now is to show that it is possible to construct problems with soft power constraints, for which linear strategies are not optimal. This may be done by an analysis

similar to that used in Appendix B. In particular, if we let $k_1 = k_2 = 100.0/(186.0423)^2$, $\sigma_x^2 = 100.0$, $\sigma_{w_1}^2 = \sigma_{w_2}^2 = 1.0$, then the optimum strategies which are optimal over the linear class yield a cost of 1.07213, whereas nonlinear strategies of the form

$$u_{01} = \gamma_{01}(x) = x$$
 (6.23a)

XXXXI • XXXXI • XXXXI • XXXXI • XXXXI

$$u_{02} = \gamma_{02}(x) = x - \text{sgn } x$$
 (6.23b)

$$u_{1} = \begin{cases} (y_{1}+y_{2}+1)/2 & \text{if } y_{1} \ge 0 \\ \\ (y_{1}+y_{2}-1)/2 & \text{if } y_{1} < 0 \end{cases}$$
(6.24)

yield a cost of 1.06634.

X

2

Ř

NA.

ίγ.

3

8

¥

Ř

We now consider Problem P3 under the additional restriction that the policies lie within the linear class. Performance bounds and optimum linear coding schemes for discrete-time multichannel communication systems have been investigated by Basar, Sankur and Abut [1980]. They have observed that when the source dimension (n) is less than the channel dimension (m), then the performance of real-time linear encoders is fairly close to the optimum achievable performance found via channel capacity. This does not hold for the case where n > m, where the performance is found to improve significantly by allowing nonlinearities in the encoding policies.

For stochastic team problems of the kind that are of interest here, we have n=1 and m=N. Thus, in view of the results reported by Basar et al. [1980], it is of interest to study optimality over the class of linear encoders, where the loss in performance may be offset by the gain associated with avoiding the complexity of more general encoders. Optimum linear encoding and decoding strategies for the two-stage stochastic team prob-

2 r an

77

F.5 C V

3

<u>`</u>

ŝ

ನ್ನ

128

lem have been studied in Basar [1980], where a rigorous solution to the general problem has been provided. Here we are interested in a special case of that general problem, i.e., the case where the source dimension is one, but with the difference that there are additional product terms in the cost functional, the encoder input is possibly noise-corrupted, and the channel noises may be correlated with the input. We now study optimality over the linear class for this case.

Under the restriction that the encoders use linear policies, we have

$$u_{0i} = \lambda_i (x+v) \tag{6.25}$$

which implies that the observation vector \overline{y} is given by

$$\bar{\mathbf{y}} = \begin{bmatrix} \lambda_1(\mathbf{x}+\mathbf{v})+\mathbf{b}_1\mathbf{x}+\mathbf{w}_1 \\ \lambda_2(\mathbf{x}+\mathbf{v})+\mathbf{b}_2\mathbf{x}+\mathbf{w}_2 \\ \vdots \\ \lambda_N(\mathbf{x}+\mathbf{v})+\mathbf{b}_N\mathbf{x}+\mathbf{w}_N \end{bmatrix}.$$
(6.26)

The covariance of x and \overline{y} is given by

$$\operatorname{Cov}(\mathbf{x}, \overline{\mathbf{y}}) = \begin{bmatrix} \boldsymbol{\Sigma}_{\mathbf{x}\mathbf{x}} & \boldsymbol{\Sigma}_{\mathbf{x}\overline{\mathbf{y}}} \\ & & \\ \boldsymbol{\Sigma}_{\mathbf{x}\overline{\mathbf{y}}}^{\mathrm{T}} & \boldsymbol{\Sigma}_{\overline{\mathbf{y}}\overline{\mathbf{y}}} \end{bmatrix}$$
(6.27)

where

$$\sum_{xx} = \sigma_x^2 \tag{6.28}$$

Sara soco Change to a set

eren Unississic Unionensi Unionene

5

$$\Sigma_{\overline{y}\overline{y}} = \begin{bmatrix} (\lambda_{1}+b_{1})^{2}\sigma_{x}^{2}+\lambda_{1}^{2}\sigma_{v}^{2}+\sigma_{w_{1}}^{2} & (\lambda_{1}+b_{1})(\lambda_{N}+b_{N})\sigma_{x}^{2}+\lambda_{1}\lambda_{N}\sigma_{v}^{2} \\ \vdots & \vdots \\ (\lambda_{1}+b_{1})(\lambda_{N}+b_{N})\sigma_{x}^{2}+\lambda_{1}\lambda_{N}\sigma_{v}^{2} & (\lambda_{N}+b_{N})^{2}\sigma_{x}^{2}+\lambda_{N}^{2}\sigma_{v}^{2}+\sigma_{w_{N}}^{2} \end{bmatrix}$$
(6.29)

and

$$\sum_{\mathbf{x}\tilde{\mathbf{y}}} = [(\lambda_1 + b_1)\sigma_{\mathbf{x}}^2, \dots, (\lambda_N + b_N)\sigma_{\mathbf{x}}^2]$$
(6.30)

We now have

$$\mathbf{E}(\mathbf{x} \mid \overline{\mathbf{y}}) = (\sum_{\mathbf{x}\overline{\mathbf{y}}} \sum_{\overline{\mathbf{y}\overline{\mathbf{y}}}}^{-1})\overline{\mathbf{y}}$$
(6.31)

and the error in estimating x from the vector observation \overline{y} is given by

$$\mathbf{e}(\lambda_1,\ldots,\lambda_N) = \sum_{\mathbf{x}\mathbf{x}} - \sum_{\mathbf{x}\overline{\mathbf{y}}} \sum_{\overline{\mathbf{y}}\overline{\mathbf{y}}}^{-1} \sum_{\mathbf{x}\overline{\mathbf{y}}}^{\mathrm{T}} .$$
(6.32)

The final task is to optimize over the linear coefficients $(\lambda_1, \ldots, \lambda_N)$, and noting that the cost function for P3 may be rewritten as

$$J_{3}(\gamma_{01}, \dots, \gamma_{0N}; \gamma_{1}) = E[\sum_{i=1}^{N} k_{i} u_{0i}^{2} + r_{i} u_{0i} x + p_{1} (u_{1} + \frac{q_{1}}{2p_{1}} x)^{2} - \frac{q_{1}^{2}}{4p_{1}} \sigma_{x}^{2}], \quad (6.33)$$

we formally have Theorem 6.5 below:

Theorem 6.5.

(i) The optimum linear strategies for Problem P3 are given by

$$u^*_{0i} = \gamma^*_{0i}(z) = \lambda^* z$$
 (6.34a)

$$\mathbf{u}_{1}^{*} = \gamma_{1}^{*}(\overline{\mathbf{y}}) = -\frac{\mathbf{q}_{1}}{2\mathbf{p}_{1}} (\sum_{\mathbf{x}\overline{\mathbf{y}}} \sum_{\overline{\mathbf{y}}\overline{\mathbf{y}}}^{-1}) \overline{\mathbf{y}}$$
(6.34b)

v

С.) С.)

 $\sum_{i=1}^{n}$

 \mathbf{x}

Ŋ

 \overline{C}

23

S

8

ý **E**

χ χ

130

where $\sum_{x\bar{y}}$ and $\sum_{\bar{y}\bar{y}}$ are defined by (6.29) and (6.30), with λ_i replaced by λ_i^* , where the λ^* 's are given by the solution to the following parameter optimization problem:

$$(\lambda^*_1, \ldots, \lambda^*_N) = \arg \min_{\substack{(\lambda_1, \ldots, \lambda_N)}} \left[\sum_{i=1}^N k_i \lambda_i^2 (\sigma_x^2 + \sigma_v^2) + r_i \lambda_i \sigma_x^2 + \frac{q_1^2}{4p_1} e(\lambda_1, \ldots, \lambda_N) \right]$$
(6.35)

where $e(\lambda_1, \ldots, \lambda_N)$ is defined by (6.32). (The above parameter optimization problem always admits a solution.)

(ii) The optimal value of the cost under linear policies is

$$J_{3}^{*} = E[\sum_{i=1}^{N} k_{i} \lambda_{i}^{*} (\sigma_{x}^{2} + \sigma_{y}^{2}) + r_{i} \lambda_{i}^{*} \sigma_{x}^{2} + \frac{q_{1}^{2}}{4p_{1}} (\sigma_{x}^{2} - e(\lambda_{1}^{*}, \dots, \lambda_{N}^{*})).$$

One special case is of particular interest, that with $\sigma_v^2=0$; $b_1=b_2=..=b_N=0$. $r_1 = r_2 = ... = r_N = 0$. In this case we get

$$e(\lambda_{1}, \dots, \lambda_{N}) = \frac{\sigma_{x}^{2}}{1 + \lambda_{1}^{2} \frac{\sigma_{x}^{2}}{\sigma_{w_{1}}^{2}} + \dots + \lambda_{N}^{2} \frac{\sigma_{x}^{2}}{\sigma_{w_{N}}^{2}}}$$
(6.36)

which implies that the optimum linear coefficients $(\lambda_1^*, \ldots, \lambda_N^*)$ may be found by solving the following parameter optimization problem:

Minimize
$$J(P_1^2, ..., P_N^2)$$

 $(P_1^2, ..., P_N^2)$

where

$$J(P_1^2, \dots, P_N^2) \equiv \sum_{i=1}^{N} k_i P_i^2 + \frac{q_1^2}{4p_1} \frac{\sigma_x^2}{1 + \sum_{i=1}^{N} (P_i^2 / \sigma_{w_i}^2)}$$
(6.37)

and

$$P_i^2 \equiv \lambda_i^2 \sigma_x^2.$$
 (6.38)

The optimum coefficients from this parameter optimization problem may be found by applying Theorem 2.2 of Başar [1980] as follows. We order the channels such that

$$q_1 \sigma_{w_1}^2 \leq q_2 \sigma_{w_2}^2 \leq \dots \leq q_N \sigma_{w_N}^2.$$
 (6.39)

The derivative of $J(P_1^{\,2},\,\ldots,\,P_N^{\,2})$ with respect to $P_i^{\,2}$ is

$$k_{i} = \frac{q_{1}^{2}}{4p_{1}} \frac{\sigma_{x}^{2} / \sigma_{w_{i}}^{2}}{\sum_{i=1}^{N} (1 + \sum_{i=1}^{N} P_{i}^{2} / \sigma_{w_{i}}^{2})}, \qquad (6.40)$$

and clearly all first derivatives cannot be zero unless the product $k_i \sigma_{w_i}^2$ is equal for all i.

If this product is indeed equal for all i, then we may arbitrarily choose one channel over which to transmit the information, and all first-order necessary conditions (which are also sufficient) are satisfied.

On the other hand, if the product $k_i \sigma_{w_i}^2$ is not equal for all i, we can choose any channel from the set for which this product is the smallest (this set may be a singleton). For the remaining channels with larger $k_i \sigma_{w_i}^2$ the first derivative is increasing at zero, implying that zero is indeed optimum.

Under the arrangement (6.39), it is therefore sufficient to choose P_1^2 possibly nonzero while all P_i^2 for i=2,...,N may be restricted to be zero, and we get

arg min
$$J(P_1^2, \dots, P_N^2) = (P_1^{*2}, \dots, P_N^{*2}) = (P_1^{*2}, \dots, P_N^{*2})$$

where

OF.

132

$$P_{1}^{*2} = sqrt(Max\{0, \frac{\sigma_{x}\sigma_{w_{1}}}{k_{1}^{*2}} - \sigma_{w_{1}}^{2}\})$$
(6.41)

and

$$P_{i}^{*2} = 0$$
 for all $i=2,...,N$.

We therefore have the following corollary to Theorem 6.5.

Corollary 6.1.

Consider Problem P3, under the additional restrictions that

$$\sigma_v^2 = 0; b_1 = b_2 = ..= b_N = 0; r_1 = r_2 = ..= r_N = 0.$$

If we assume that the channels are ordered such that (6.39) holds, then the optimal linear strategies are given by

$$u_{01}^{*} = \gamma_{01}^{*}(x) = \lambda_{1}^{*}x$$
$$u_{0i}^{*} = \gamma_{0i}^{*}(x) = 0 \text{ for all } i=2,...,N$$

and

Ŕ

$$u_{1}^{*} = \gamma_{1}^{*}(\bar{y}) = -\frac{q_{1}}{2p_{1}} \cdot \frac{\lambda_{1}^{*}\sigma_{x}^{2}}{\lambda_{1}^{*2}\sigma_{x}^{2} + \sigma_{w_{1}}^{2}} y_{1}$$

where

$$\lambda *_{1}^{2} = \frac{1}{\sigma_{x}^{2}} P *_{1}^{2} ,$$

and P_{1}^{*2} is defined by (6.41).

(ii) The optimal value of the cost is

 $k_{1}P_{1}^{*2} + \frac{q_{1}^{2}}{4p_{1}} \frac{\sigma_{x}^{2}\sigma_{w_{1}}^{2}}{(P_{1}^{*2} + \sigma_{w_{1}}^{2})} - \frac{q_{1}^{2}}{4p_{1}}\sigma_{x}^{2}.$

6.6. Conclusion

In this chapter we have formulated and analyzed three classes of decentralized, multiperson, stochastic team problems, where communication between agents is permitted only via noise-corrupted channels.

For problems involving a single transmitting agent and multiple receiving agents we have shown that the optimum strategies are linear when either the encoder observes an uncorrupted version of the variable to be transmitted, or when all channel noises are independent of the input.

For problems involving multiple transmitting agents and multiple receiving agents we again find that in case the encoders observe an uncorrupted version of the variable to be transmitted, or if all channel noises are independent of the input, then the optimum strategies are linear. However, for the simplest classes of problems involving multiple transmitting agents and a single receiving agent, we find that the strategies which are optimal over the linear class may be outperformed by appropriately chosen nonlinear policies, and for this case we have provided strategies which are optimal within the linear class.

CHAPTER 7

PROBLEMS WITH INCOMPLETE CHANNEL DESCRIPTION

7.1. Introduction

1000000

A STATE OF A DESCRIPTION OF A DESCRIPTIO

In this chapter, we expand on the framework developed so far, by allowing incomplete statistical description of the channel used to transmit measurements between the decentralized agents. We still operate under nonclassical information patterns and consider a number of cases depending on whether there are "hard" power constraints or "soft" power constraints on some of the decision variables and/or soft costs on communication. We obtain minimax rules in all cases, some being saddle points and others not, the techniques of derivation being very much case dependent. These are all important prototype problems which could be considered essential building blocks in multistage, distributed decision making under nonclassical information and with partial statistical description.

In particular, we assume that the variable transmitted through the channel is corrupted not only by an independent Gaussian noise of given variance, but also by an unknown channel noise which is only known to satisfy certain power constraints. Further, this unknown noise is allowed to be correlated with either the input or the output of the encoder. (Recall that in Chapter 2 we had allowed part of the channel noise to vary *linearly* with the input).

We seek optimal solutions under a worst case analysis. We may therefore consider the unknown channel noise as being controlled by an adversary or "jammer," who intelligently uses the knowledge of either the input or the output of the channel to design a jamming strategy. Thus the problems may be viewed as those representing extended versions of the standard Gaussian Test Channel (Gallager [1968]), further including an intelligent jammer.

The organization of this chapter is as follows. In Section 7.2 we identify the problems to be analyzed in the sequel. In particular, three kinds of communication systems are identified, each of which is subsequently analyzed under a variety of fidelity criteria, which are also defined in Section 7.2. The three types of communication systems are analyzed in Sections 7.3, 7.4 and 7.5. The concluding remarks in Section 7.6 then end this chapter.

In what follows, we shall refer to the problems studied in this chapter alternatively as zero-sum games, since the jammer wishes to maximize the same criterion which the encoder-decoder pair is trying to minimize. A general discussion on zero sum games may be found in Basar and Olsder [1982].

7.2. Problem Description

N.

* . 5 6 . 1

8

8

8

X

Ś

ğ

X

Ŕ

7.2.1. Channel description

Following the formulation of Basar [1983], consider the communication systems depicted in Figures 7.1, 7.2 and 7.3 which represent extended versions of the standard Gaussian Test Channel, (Gallager [1968]), and include an intelligent jammer who has access either to the input or to the output of the encoder. The input to the encoder is a Gaussian random variable with zero mean and unit variance, denoted $u \sim N(0,1)$. The transmitter encodes the input signal u into a variable $y = \gamma(u)$ with the encoding policy γ being an







Figure 7.2. Schematics for games of Type 2.



Figure 7.3. Schematics for games of Type 3.

Ľ

222

1

V)

3

X

33

8

 $\frac{2}{2}$

X).

8

--33

3

8

8

, V

 $\langle \cdot \rangle$

<u>ين</u> بر

55 101

element of the space Γ_e of random mappings or an element of the space Γ_{ed} of deterministic mappings. The encoder is further restricted to satisfy certain power constraints as to be elucidated in the sequel. When the encoder map is allowed to be probabilistic we refer to the problem as a game of Type 1, in which case we also allow for a side channel that permits the transmission of information regarding the structure of the probabilistic encoder to the receiver (Figure 7.1). When the encoder map is necessarily deterministic it is a game of Type 2, for which the side channel is superfluous (Figure 7.2).

In games of Types 1 and 2 the jammer is allowed to observe the input u. The jammer then feeds back into the channel a random variable $v = \beta(u)$ so that the input to the decoder is

$$z = \gamma(u) + \beta(u) + w .$$

The jammers policy is an element in the space Γ_j of random mappings also restricted to satisfy certain power constraints.

At the receiver, the decoder $\delta(z)$ is chosen so as to cooperate with the encoder in minimizing a (generalized) quadratic distortion measure $J(\gamma, \delta, \beta)$. Here the noise w is assumed to be Gaussian with mean zero, variance σ_w^2 , and is independent of u.

Figure 7.3 represents the situation when the jammer has access to the output of the encoder instead of u. This situation shall be referred to as a game of Type 3. Games of Type 3 are investigated in Bansal and Basar [1989] under a variety of fidelity criteria which we next describe.

7.2.2. Fidelity criteria

The general communication game addresses the reproduction of the input u. with high fidelity, at the output of the decoder. This corresponds to minimizing the mean
square error $E\{(\delta(z) - u)^2\}$. In order that this minimization be well defined we restrict ourselves to using encoding and jamming policies which satisfy $E[\gamma^2(u)] \leq P_0^2$ and $E[v^2] \leq P_j^2$. These are the *hard* power constraints generally assumed in the communication literature, which lead to criterion C_1 below.

Criterion C₁:

$$J_{1}(\gamma,\delta,\beta) = E\{(\delta(z) - u)^{2}\}$$
(7.1a)

and

$$E[\gamma^{2}(u)] \leq P_{0}^{2}, E[v^{2}] \leq P_{J}^{2}.$$
(7.1b)

This is precisely the criterion adopted in Basar [1983] and Basar and Wu [1985], under which a complete characterization of minimax encoder decoder policies for games of Types 1 and 2 has been provided in Basar and Wu [1985]. We introduce here some classes of problems which have heretofore received less attention. These are problems where the power levels are not fixed a priori, but are determined as a result of the underlying optimization problem. This is particularly important in situations where it may be possible, for instance, to transmit a larger power at the encoder (at some cost) in order to further decrease the mean square error at the decoder. Mathematically this may be represented as a power constraint which is "implied" or "soft" and appears as an additional term in the cost functional. Based on the consideration of tradeoffs between power used and the magnitude of the mean square error term, we have criteria C_2 , C_3 and C_4 below where coefficient terms k_0 and α are positive scalars in all cases. Criterion C_2 :

$$J_{2}(\gamma,\delta,\beta) = E\{k_{0}\gamma^{2}(u) + (\delta(z) - u)^{2} - \alpha v^{2}\}$$
(7.2)

Criterion C₃:

$$J_{3}(\gamma,\delta,\beta) = E\{(\delta(z) - u)^{2} - \alpha v^{2}\}$$
(7.3a)

$$\mathbf{E}[\boldsymbol{\gamma}^{2}(\mathbf{u})] \leqslant \mathbf{P}_{0}^{2} \tag{7.3b}$$

Criterion C_4 :

$$I_{4}(\gamma,\delta,\beta) = E\{k_{0}\gamma^{2}(u) + (\delta(z) - u)^{2}\}$$
(7.4a)

$$\mathbf{E}[\mathbf{v}^2(\mathbf{u})] \leqslant \mathbf{P}_1^2 \,. \tag{7.4b}$$

Habessee

Under the different criteria above, the spaces in which the policies need to be restricted are characterized by using a superscript. For example, Γ_{ed}^{k} denotes the class of deterministic encoder policies which are permissible when criterion C_{k} is under consideration.

7.2.3. Solution concepts

We are faced with multiperson decision problems with conflicting objectives. We thus seek to obtain worst case solutions by minimax and maximin approaches, and further, saddle-point solutions if they exist.

Under the minimax approach the encoder-decoder pair is assumed to be careful and defensive, wishing to protect against any irrational behavior of the jammer. Thus, under the minimax criterion, we evaluate the upper value \overline{J} of the zero sum game with kernel J,

$$J = J(\gamma^{*}, \delta^{*}, \beta^{*}_{(\gamma^{*}, \delta^{*})}) = \min_{\substack{\gamma, \delta \\ \gamma, \delta \\ = \min_{\gamma, \delta} J(\gamma, \delta, \beta^{*}_{(\gamma, \delta)})}$$
(7.5)

ХÌ ХÌ

8

Ř

7

140

where

$$\beta^{*}_{(\gamma,\delta)} = \arg \max_{\beta} J(\gamma,\delta,\beta)$$
(7.6)

assuming that such a solution exists. The triple $J(\gamma^*, \delta^*, \beta^*_{(\gamma^*, \delta^*)})$ defined by (7.5) and (7.6) above is called the minimax solution for the corresponding game.

To obtain the maximin solution, we consider instead the lower value of the zero sum game:

$$\underline{J} = J(\gamma^{*}_{\beta*}, \delta^{*}_{\beta*}, \beta^{*}) = \max_{\beta} \min_{(\gamma, \delta)} J(\gamma, \delta, \beta)$$

= max $J(\gamma^{*}_{\beta}, \delta^{*}_{\beta}, \beta)$ (7.7)

where the pair $(\gamma {}^{\boldsymbol{\ast}}{}_{\boldsymbol{\beta}}, \delta {}^{\boldsymbol{\ast}}{}_{\boldsymbol{\beta}})$ is determined by

$$(\gamma^*_{\beta}, \delta^*_{\beta}) = \arg\min_{\substack{(\gamma, \delta) \\ (\gamma, \delta)}} J(\gamma, \delta, \beta) .$$
(7.8)

The triple $(\gamma_{\beta_*}, \delta_{\beta_*}, \beta^*)$ defined by (7.7) and (7.8) above is the maximin solution for the corresponding game.

A saddle-point solution $(\gamma^*, \delta^*, \beta^*)$ will exist if and only if

$$\underline{J} = \overline{J} = J(\gamma^*, \delta^*, \beta^*) . \tag{7.9}$$

A detailed study of the concepts of minimax and maximin strategies and saddle-point solutions may be found in Basar and Olsder [1982].

7.2.4. Notation

We shall use a double subscript notation and refer to a game as Game G_{ik} , with $i \in \{1,2,3\}$ and $k \in \{1,2,3,4\}$. The first subscript i refers to the game type (Section 7.2.1) while the second subscript k indicates that fidelity criterion C_k is under consideration. with the strategy spaces being Γ_{ed}^{k} or Γ_{e}^{k} , Γ_{r}^{k} and Γ_{j}^{k} for the encoder, decoder and jammer, respectively. Thus for i=3 the jammer has access to the output of the encoder, whereas for i=1 or 2 the jammer taps the input u. Further, for i=1 the encoder is allowed to use probabilistic mappings whereas for i=2, the encoder is necessarily deterministic.

7.2.5. Contributions of the chapter

The existence of a saddle-point solution for game G_{31} is proved in Basar [1983] in a more general framework where it is shown that the parameter space can be partitioned into regions with the saddle point depending *structurally* upon the region of operation. Games G_{i1} for i=1,2 are considered in Basar and Wu [1985] where the existence of a saddle-point solution for G_{11} is established and a complete characterization of minimax and maximin strategies for G_{21} is provided.

In this chapter we study the remaining nine games G_{ik} , $i \in \{1,2,3\}$ and $k \in \{2,3,4\}$. In Section 7.2 we provide saddle-point solutions for games G_{12} , G_{13} and G_{14} . In Section 7.4 we analyze games G_{32} , G_{33} and G_{34} . For games G_{22} , G_{23} and G_{24} saddle-point solutions do not exist; we provide both maximin and minimax solutions for these games in Section 7.5.

7.3. Saddle-Point Solutions for Games G_{12} , G_{13} and G_{14}

Consider the problem described in Section 7.1, with the encoder allowed to use probabilistic mappings (Figure 7.1), under Criteria C2, C3 and C4, respectively. The solution is provided in Theorem 7.1 after introducing some notation.

ЗХ Х

8

8

X

ς,

ŝ

8

<u>,</u>

15 γ,

Preliminary Notation for Theorem 7.1

We introduce scalar parameters P_0^* and P_J^* by dividing the parameter space into distinct regions as follows:

a. Game G₁₂

Define Regions R_{12}^{1} , R_{12}^{2} and R_{12}^{3} by

$$R_{12}^{1}: \frac{1}{k_{0}} \le \sigma_{w}^{2}$$
(7.10a)

$$R_{12}^{2}: \frac{k_{0}}{(k_{0}+\alpha)^{2}} \leq \sigma_{w}^{2} < \frac{1}{k_{0}}$$
(7.10b)

$$R_{12}^{3}: \sigma_{w}^{2} < \frac{k_{0}}{\left(k_{0}+\alpha\right)^{2}}.$$
(7.10c)

Now define P_0^* and P_J^* by

$$P_{0}^{*} = \begin{cases} 0 & \ln R_{12}^{-1} \\ \frac{\sigma_{w}}{k_{0}^{\frac{1}{2}}} - \sigma_{w}^{2} \\ \frac{\kappa_{0}^{\frac{1}{2}}}{k_{0}} & \ln R_{12}^{-3} \\ \frac{\alpha^{\frac{1}{2}}}{(k_{0} + \alpha)} & \ln R_{12}^{-3} \end{cases}$$
(7.11a)

$$P*_{J} = \begin{cases} 0 & \text{in } R_{12}^{1} \text{ and } R_{12}^{2} \\ \frac{k_{0}}{\left(\frac{k_{0}}{\left(k_{0}+\alpha\right)^{2}} - \sigma_{w}^{2}\right)^{\frac{1}{2}}} & \text{in } R_{12}^{3} \end{cases}$$
(7.11b)

b. Game G_{13}

Define Regions R_{13}^1 and R_{13}^2 by

$$R_{13}^{1}: \alpha \ge P_{0}^{2} / (P_{0}^{2} + \sigma_{w}^{2})^{2}$$
(7.12a)

PLANDER PARAMON 2004524

- 12 La 12 L

$$R_{13}^{2}: \alpha < P_{0}^{2} / (P_{0}^{2} + \sigma_{w}^{2})^{2}.$$
(7.12b)

Now define P_0^* and P_J^* by

$$P_0^* = P_0$$
 in R_{13}^1 and R_{13}^2 (7.13a)

$$P*_{J} = \begin{cases} 0 & \text{in } R_{13}^{-1} \\ (P_{0}/\alpha^{1/2}) - (P_{0}^{2} + \sigma_{w}^{2}) & \text{in } R_{13}^{-2} \end{cases}$$
(7.13b)

c. Game G_{14}

Define Regions R_{14}^1 and R_{14}^2 by

$$R_{14}^{1}: k_{0} \ge (P_{J}^{2} + \sigma_{w}^{2})^{-1}$$
(7.14a)

$$R_{14}^{1}: k_{0} < (P_{J}^{2} + \sigma_{w}^{2})^{-1}.$$
(7.14b)

Now define P_0^* and P_J^* by

$$P_{0}^{*} = \begin{cases} 0 & \text{in} R_{14}^{1} \\ (((P_{J}^{2} + \sigma_{w}^{2})^{\frac{1}{2}} / k_{0}^{\frac{1}{2}}) - (P_{J}^{2} + \sigma_{w}^{2})^{\frac{1}{2}} & \text{in} R_{14}^{2} \end{cases}$$
(7.15a)

$$P*_{J} = P_{J}$$
 in R_{14}^{1} and R_{14}^{2} . (7.15b)

Theorem 7.1: Consider the games G_{12} , G_{13} and G_{14} described in Section 1. For each of these games there exists a saddle-point solution ($\gamma *, \delta *, \beta *$) given as follows:

(7.16)

144

$$(\gamma^*, \delta^*) = \begin{cases} (P^*_0 u, (P^*_0 / (P^*_0^2 + \sigma_w^2 + P^*_J^2))z) & \text{w.p. } 0.5 \end{cases}$$

$$(-P*_0u,(-P*_0/(P*_0^2+\sigma_w^2+P*_J^2))_Z)$$
 w.p. 0.5

$$\beta^{\#}(\mathbf{u}) = \eta \tag{7.17}$$

where η is a zero mean Gaussian random variable with variance P_{J}^{*2} , which is independent of u and w.

Proof: We need to prove that the solutions given above satisfy the saddle-point inequalities:

$$J(\gamma^*, \delta^*, \beta) \leq J(\gamma^*, \delta^*, \beta^*) \leq J(\gamma, \delta, \beta^*)$$
(7.18)

for all permissible triplets (γ, δ, β) and under the stipulation that the side channel is used io carry structural information concerning the probabilistic encoder mappings. The proof is provided by establishing separately the validity of the right-hand side (RHS) and lefthand side (LHS) inequalities of (18) under each criterion for each of the regions in consideration.

(a) Game G_{12}

Region R_{12}^{1}

(i) The RHS inequality

The problem faced by the encoder-decoder pair, with the jammer's policy fixed as indicated is

$$\min_{\gamma,\delta} E[k_0 \gamma^2(u) + (\delta(z)-u)^2]$$

where $z = \gamma(u) + w$.

For each fixed power level $(E[\gamma^2(u)] = P_0^2)$ the problem above is the standard Gaussian test channel for which the optimal solution is known to be linear. The optimal power levels may be found by solving the parameter optimization problem:

$$\min_{\mathbf{P}_{0}} \left[k_{0} \mathbf{P}_{0}^{2} + \sigma_{\mathbf{w}}^{2} / (\mathbf{P}_{0}^{2} + \sigma_{\mathbf{w}}^{2}) \right]$$

BARANARE & BUSY /////X & PRAYMORE & BARANARE & BUSY ///// & D

which gives $P_0^{*2} = 0$ as in the statement of the theorem.

(ii) The LHS inequality

With the encoding-decoding policy fixed at zero, the problem faced by the jammer is

$$\max_{\beta(u)} (-\alpha \beta^2(u) + 1)$$

for which the optimal choice is $\beta^{*}(u) = 0$.

Region R_{12}^2

(i) The RHS inequality

As in the case of Region R_{12}^1 , we find that the optimal encoding-decoding policy is linear and the linear encoding coefficient may be found by solving the parameter optimization problem

$$\min_{P_{0}} [k_{0}P_{0}^{2} + \sigma_{w}^{2}/(P_{0}^{2} + \sigma_{w}^{2})]$$

which gives $P_0^{*2} = (\sigma_w / k_0^{1/2}) - \sigma_w^2$ as in the statement of the theorem.

(ii) The LHS inequality

With the encoding-decoding policy fixed as given, the cost to be maximized by the jammer is obtained by unconditioning the conditional value of the cost incurred for each

E

ŷ,

8

 Σ

ĝ

3

멍

х S

5

83

æ

146

realization of the equiprobable random variable which decides the sign of the linear encoding. In the expression thus obtained for the cost to be maximized, the jammer's policy only enters as a term containing $E[\beta^2(u)]$ with the coefficient

$$(((\sigma_{w}/k_{0}^{\frac{1}{2}}) - \sigma_{w}^{2})k_{0}/\sigma_{w}^{2}) - \alpha$$

This coefficient is nonpositive in R_{12}^2 , implying that $\beta(u) = 0$ is the jammer's optimal policy.

Region R_{12}^3

(i) The RHS inequality

When the jammer uses independent Gaussian noise with the indicated variance, the problem faced by the encoder-decoder pair is

$$\min_{\substack{\gamma,\delta}} E[k_0 \gamma^2(u) + (\delta(z) - u)^2 + K']$$

where $z = y(u) + w + \eta$ and K' is independent of the encoding-decoding policy. For each fixed power level, $E[\gamma^2(u)] = P_0^2$, the problem is the standard Gaussian test channel problem, and thus the coefficient of the optimal linear solution may be found by solving the parameter optimization problem

$$\min_{P_0} \left[\frac{k_0 P_0^2 + \frac{k_0}{(k_0 + \alpha)^2}}{P_0^2 + \frac{k_0}{(k_0 + \alpha)^2}} \right]$$

which gives P_0^* as in the statement of the theorem.

(ii) The LHS inequality

When the encoder-decoder policy is fixed as indicated, the cost function to be maximized by the jammer is obtained by unconditioning the conditional value of the cost which gives an expression independent of β , and thus the jamming policy may be chosen as in

pontos Operatoras Oucurantes Casaccopers Operatores

Deressed of the second deveraged deverage of the second

Secondo

(b) Game G_{13}

Region R_{13}^{1}

(i) The RHS inequality

the statement of the theorem.

With the jammer's policy fixed at zero, the encoder-decoder pair faces the standard Gaussian test channel problem for which the optimal linear solution is as indicated in the statement of the theorem.

(ii) The LHS inequality

When the encoder-decoder policy is fixed as indicated, the cost function to be maximized by the jammer is obtained by unconditioning the conditional value of the cost which gives

$$E[(-\alpha + P_0^2/(P_0^2 + \sigma_w^2)^2)\beta^2(u) + K'']$$

where K'' is independent of β . In the region under consideration the coefficient of $E[\beta^2(u)]$ is negative implying that $\beta(u) = 0$ is an optimum.

Region R_{13}^2

(i) The RHS inequality

With the jammer's policy fixed as given, the encoder-decoder pair faces the standard Gaussian test channel problem, and the solution is as given.

148

(ii) The LHS inequality

With the encoding-decoding policy fixed as indicated, the cost function to be maximized by the jammer, obtained by unconditioning the cost, becomes independent of $\beta(u)$ which may, therefore, be chosen as in the statement of the theorem.

(c) Game G_{14}

Region R_{14}^{1}

(i) The RHS inequality

With the jamming policy fixed as Gaussian noise, the problem faced by the encoderdecoder pair is

$$\min_{\substack{\gamma,\delta}} \mathbb{E}[k_0 \gamma^2(u) + (\delta(z) - u)^2]$$

with $z = \gamma(u) + w + \eta$.

The coefficient of the optimal linear solution to the above problem may be found by solving the parameter optimization problem

$$\min_{P_0} k_0 P_0^2 + \frac{\sigma_w^2 + P_J^2}{P_0^2 + P_J^2 + \sigma_w^2}$$

for which $P_0^{*2} = 0$ in the region under consideration.

(ii) The LHS inequality

With the encoder-decoder policy fixed at zero, the cost function is independent of $\beta(u)$ which may be chosen to be Gaussian noise as in the statement of the theorem.

Region R_{14}^2

(i) The RHS inequality

The problem faced by the encoder-decoder pair, with the jammer's policy fixed as Gaussian noise is

$$\min_{\substack{\gamma,\delta}} E[k_0 \gamma^2(u) + (\delta(z) - u)^2]$$

where $z = \gamma(u) + w + \eta$.

The coefficients for the optimal linear solution to the above problem may be obtained by solving the parameter optimization problem

$$\min_{P_0} k_0 P_0^2 + (\sigma_w^2 + P_J^2) / (P_0^2 + P_J^2 + \sigma_w^2)$$

which gives the optimal encoding-decoding policies as in the statement of the theorem.

(ii) The LHS inequality

With the encoding-decoding policy fixed as indicated, the cost to be maximized by the jammer, obtained by unconditioning, increases linearly with $E[\beta^2(u)]$. Thus the jammer may choose any random variable with second moment equal to P_J^2 ; i.e., $\beta(u)$ may be chosen to be independent Gaussian noise as in the statement of the theorem.

7.4. Analysis of Games G_{32} , G_{33} and G_{34}

7.4.1. Some useful results

The most relevant paper to this section is Basar [1983], where a game of Type 3 with fidelity Criterion C1 has been studied. Here we reproduce the main theorem of that paper (in the current less general framework) for future reference. First note that if $P_o \leq P_J$ the solution is trivial since the jammer can cancel out the signal component. If, however, $P_o > P_J$, then it is convenient to partition the parameter space into two regions

R1:
$$P_o > P_J$$
 and $P_oP_J - P_J^2 < \sigma_w^2$
R2: $P_o > P_J$ and $P_oP_J - P_J^2 \ge \sigma_w^2$

and to introduce

and

المقرار مراجع من اللارين

a the second of the second second

12290000

$$t^* = 1 - (P_J^2 + \sigma_w^2)^2 / P_o^2 P_J^2$$

 $\lambda^* = -P_1/P_0$

Lemma 7.1. Game G_{31} admits two saddle-point solutions $(\gamma^*, \delta^*, \mu^*)$ and $(-\gamma^*, -\delta^*, \mu^*)$, where

(i)
$$\gamma^*(u) = P_0 u$$

(ii) μ^* is the Gaussian probability measure associated with the random variable

$$\mathbf{v} = \boldsymbol{\beta}^{*}(\mathbf{y}) = \begin{cases} \lambda^{*}\mathbf{y} & \text{in } \mathbf{R}\mathbf{1} \\\\\\ \lambda^{*}(\mathbf{1}-\mathbf{t}^{*})^{\frac{1}{2}}\mathbf{y} + \eta^{*} & \text{in } \mathbf{R}\mathbf{2} \end{cases}$$

where $\eta^* \sim (0, t^* P_J^2)$

(iii) δ^* is the Bayes estimator for u under the least favorable distribution μ^* , computed as

$$\delta^{*}(z) = \begin{cases} (P_{o}(1+\lambda^{*})/[P_{o}^{2}(1+\lambda^{*})^{2}+\sigma_{w}^{2}]) z & \text{in } R1 \\ \\ (1/P_{o}) z & \text{in } R2 \end{cases}$$

Proof. See Basar [1983].

In order to establish similar results under Criteria C2, C3 and C4, we shall need the following lemmata:

Lemma 7.2. Consider the stochastic team problem with nonclassical information (which can be obtained from C2 by setting $\beta(\cdot) \equiv 0$)

$$\min_{\substack{\gamma,\delta}} E\{k_o \gamma^2(u) + (\delta(z) - u)^2\}$$

where $z = \gamma(u) + w$. The optimal solution to this problem is linear, i.e.,

 $\gamma^*(u) = P^*_{o}u$

and

88

F.88

80

335

8

8

G.

22

Ŷ,

8

10

$$\delta^{*}(z) = E[u | P^{*}_{o}u + w] = [P^{*}_{o}/(P^{*2}_{o} + \sigma^{2}_{w})] z$$

where

$$P_{o}^{*} \triangleq \arg\min_{P} \{k_{o}P^{2} + \sigma_{w}^{2}/(P^{2} + \sigma_{w}^{2})\}$$

Proof. See Bansal and Basar [1987a].

Lemma 7.3. Consider the stochastic team decision problem with nonclassical information (which can be obtained from C3 by setting $\beta(\cdot) \equiv 0$)

7

R

3

Ż

5

 \mathbf{v}_{i}

Ę

- 0

min
$$E\{(\delta(z) - u)^2\}$$

 γ, δ

where $z = \gamma(u) + w$ and $E[\gamma^2(u)] \leq P^2$. The optimal solution is linear and

$$\gamma^{*}(u) = Pu ,$$

$$\delta^{*}(z) = \left[P/(P^{2} + \sigma_{w}^{2}) \right] z$$

Proof. The problem may be viewed as the standard Gaussian Test Channel problem, from which the optimality of the linear solution follows. Alternatively, the proof follows in a straightforward fashion using a contradiction on Lemma 7.2.

We note here that because of the complex structure of the solution to the hard constraint problem (discussed in Basar [1983]), the solution given in Lemma 7.1 cannot readily be used to obtain the solution to the soft constraint versions. In fact, it turns out that, for each of the soft constraint versions, the saddle-point property has to be established individually.

7.4.2. Solutions to the minimax problem under the three criteria

(a) Game G_{32}

Preliminary notation for Theorem 7.2.

We decompose the parameter space into three regions R1, R2 and R3:

$$R_{32}^{1}: \alpha \leq k_{o}$$

$$R_{32}^{2}: \alpha > k_{o} \text{ and } \alpha k_{o} - k_{o}^{2} < \alpha^{3} \sigma_{w}^{2}$$

and

$$R_{32}^3$$
: $\alpha > k_o$ and $\alpha k_o - k_o^2 \ge \alpha^3 \sigma_w^2$.

We next introduce the scalar parameters $P^{*}_{\ o}$ and λ^{*} by

$$P_{o}^{*} = \begin{cases} 0 & \text{in } R_{32}^{1} \\ (Max\{0, \sigma_{w}((k_{o}^{-1} - \alpha^{-1})^{\frac{1}{2}} - \sigma_{w})/((1 - k_{o}/\alpha)^{2})\})^{\frac{1}{2}} & \text{in } R_{32}^{2} \\ \alpha^{-\frac{1}{2}} & \text{in } R_{32}^{3} \end{cases}$$

and

)

88

8

5

Ŋ

8

X

71

$$\lambda^* = \begin{cases} -1 & \text{in } \mathbb{R}_{32}^{-1} \\ \\ -k_0/\alpha & \text{in } \mathbb{R}_{32}^{-2} \text{ and } \mathbb{R}_{32}^{-3} \end{cases}$$

Theorem 7.2. Game G_{32} admits two saddle-point solutions $(\gamma^*, \delta^*, \mu^*)$ and $(-\gamma^*, -\delta^*, \mu^*)$ where

(i)
$$\gamma^*(u) = P^*_{a}u$$

(ii) μ^* is the Gaussian probability measure associated with the random variable

$$\mathbf{v} = \beta^{*}(\mathbf{y}) = \begin{cases} \lambda^{*}\mathbf{y} & \text{in } \mathbf{R}_{32}^{1} \text{ and } \mathbf{R}_{32}^{2} \\ \\ \lambda^{*}\mathbf{y} + \eta^{*} & \text{in } \mathbf{R}_{32}^{3} \end{cases}$$

where η^* is an independent Gaussian random variable, i.e.,

$$\eta^* \sim N(0, \sigma_{\eta^*}^2) = N(0, k_0 / \alpha^2 - k_0^2 / \alpha^3 - \sigma_w^2)$$

(iii) δ^* is the Bayes estimator for u under the least favorable distribution μ^* , computed as

X)

20

3

83

1946 - 1947 - 1948 1946 - 1948 - 1948

, V

ĸ٦ Ċ,

154

$$\delta^{*}(z) = p^{*}z = \begin{cases} (P^{*}_{o}(1+\lambda^{*})/(P^{*}_{o}^{2}(1+\lambda^{*})^{2}+\sigma_{w}^{2})) \ z \ \text{in } \mathbb{R}^{1}_{32} \text{ and } \mathbb{R}^{2}_{32} \\ \\ (1/P^{*}_{o}) \ z \ \text{in } \mathbb{R}^{3}_{32} \ . \end{cases}$$

Proof. We shall establish separately the validity of the right-hand side (RHS) and lefthand side (LHS) inequalities of (7.18) in each of the Regions R_{32}^1 , R_{32}^2 and R_{32}^3 .

Region R_{32}^{1}

(i) The RHS inequality:

With $\beta(y) = \beta^*(y) = -y$, the problem faced by the encoder-decoder pair is

min E{
$$(k_0 - \alpha)\gamma^2(u) + 1$$
}
 γ,δ

which implies $\gamma^*(u) = 0$ and $\delta^*(z) = E[u | w] = 0$

(ii) The LHS inequality:

With $\gamma^* = \delta^* = 0$, the problem faced by the jammer is

$$\max_{\beta} E\{1 - \alpha \beta^2(y)\}$$

and since y=0 a.s., any choice of β with $E[\beta^2(y)]=0$ attains the maximum (which is 1), one such policy being $\beta(y) = -y$.

Region R_{32}^2

(i) The RHS inequality:

With the jammer's policy fixed as $\beta^*(y) = (-k_0/\alpha)y$, the problem faced by the encoderdecoder pair is

$$\min_{\substack{\gamma,\delta}} E[(k_o - k_o^2/\alpha)\gamma^2(u) + (\delta(z) - u)^2]$$

with $z = (1-k_o/\alpha)\gamma(u) + w$.

8

82

88 88

×2

R,

22

Ŋ,

Š.

Defining $\gamma'(u) = (1-k_o/\alpha)\gamma(u)$, the problem may be rewritten as

$$\min_{\substack{\gamma',\delta}} E[k'(\gamma'(u))^2 + (\delta(z)-u)^2]$$

where $z = \gamma'(u)+w$. It follows from Lemma 7.2 that the optimal encoder-decoder policies are linear in the observations, and may be obtained by minimizing over $P_o^2 \ge 0$ the expression

$$J(P_o^2) = k_o(1-k_o/\alpha)P_o^2 + \sigma_w^2/(1-k_o/\alpha)^2P_o^2 + \sigma_w^2,$$

which yields P^*_{o} as in the statement of the theorem.

(ii) The LHS inequality:

With the encoder-decoder policy fixed as given, the problem faced by the jammer is

$$\max_{\beta} E[(p^{*2} - \alpha)\beta^{2}(y) + 2p^{*}(p^{*}P^{*}_{o} - 1)\beta(y)u].$$

Under the assumption $p^{*2}-\alpha < 0$, its solution is unique (because of strict concavity), and is given by

$$\beta(y) = [p^{*}(p^{*}P^{*}_{o}-1)/(P^{*}_{o}(\alpha-p^{*}))] y = \lambda^{*}y$$

With p* as indicated, this condition becomes

$$-\lambda^*(1+\lambda^*) < \sigma_w^2/P_o^*^2$$

which implies and is implied by

$$\alpha^3 \sigma_{\rm w}^2 > \alpha k_{\rm o} - k_{\rm o}^2$$

Region R_{32}^3

(i) The RHS inequality:

With the jammer's policy fixed as indicated, the problem faced by the encoder-decoder pair is

156

$$\min_{\substack{\gamma,\delta}} E\{(k_o - k_o^2/\alpha)\gamma^2(u) + (\delta(z) - u)^2 - \alpha \eta^{*2}\}.$$

Using Lemma 7.2 as in Region R_{32}^2 , we obtain the encoder policy to be linear, with the corresponding power level obtained by minimizing over P_0^2 the expression

$$J(P_o^2) = k_o(1-k_o/\alpha)P_o^2 + (\sigma_w^2 + \sigma_{\eta^*}^2)/[(1-k_o/\alpha)^2 P_o^2 + \sigma_w^2 + \sigma_{\eta^*}^2]$$

which requires (by differentiating with respect to P_o^2 and noting that the second-order condition for a minimum is always satisfied)

$$k_{o}(1-k_{o}/\alpha) - (\sigma_{w}^{2} + \sigma_{\eta^{*}}^{2})(1-k_{o}/\alpha)^{2} / [(1-k_{o}/\alpha)^{2}P_{o}^{*} + \sigma_{w}^{2} + \sigma_{\eta^{*}}^{2}]^{2} = 0$$

which is satisfied (uniquely) by $P_{o}^{*2} = 1/\alpha$. Furthermore,

$$\delta(z) = \mathrm{E}[\mathbf{u} \mid z] = \sqrt{\alpha} z \; .$$

(ii) The LHS inequality:

With the encoder-decoder policy fixed as indicated, the problem faced by the jammer is

$$\max_{\beta} E\{(p^{*2}-\alpha)\beta^{2}(y) + 2p^{*}(p^{*}P^{*}_{o}-1)\beta(y)u\}.$$

Since $p^{*2} = \alpha$ and $p^* = 1/P^*_{o}$, this cost is independent of β , and hence the jammer's policy may be chosen as indicated.

(b) Game G₃₃

Consider the game of Type 3, under Criterion C3. The solution is provided in Theorem 7.3 below, after introducing some notation.

Preliminary Notation for Theorem 7.3.

We decompose the parameter space into Regions R_{33}^{1} and R_{33}^{2} , as follows:

$$R_{33}^{1}: (P_{o}^{2} \neq 1/\alpha) \cup (P_{o}^{2} = 1/\alpha < 4\sigma_{w}^{2})$$
$$R_{33}^{2}: P_{o}^{2} = 1/\alpha \ge 4\sigma_{w}^{2}.$$

We next introduce the scalar parameter λ^* by

$$\lambda^{*} = \begin{cases} \arg \max_{\lambda} \left\{ \sigma_{w}^{2} / [(1+\lambda)^{2} P_{o}^{2} + \sigma_{w}^{2}] - \alpha \lambda^{2} P_{o}^{2} \right\} & \text{in } R_{33}^{1} \\ \\ -\frac{1}{2} & \text{in } R_{33}^{2} \end{cases}$$

Note that $\lambda^* \in (-1,0)$.

Theorem 7.3. Game G_{33} admits two saddle-point solutions $(\gamma^*, \delta^*, \mu^*)$ and $(-\gamma^*, -\delta^*, \mu)$ where

$$(i) \gamma^*(u) = P_o u$$

(ii) μ^* is the Gaussian probability measure associated with the random variable

$$\mathbf{v} = \boldsymbol{\beta}^{*}(\mathbf{y}) = \begin{cases} \lambda^{*}\mathbf{y} & \text{in } \mathbf{R}_{33}^{1} \\ \\ \lambda^{*}\mathbf{y} + \eta^{*} & \text{in } \mathbf{R}_{33}^{2} \end{cases}$$

where $\eta^* \sim N(0, \sigma_{\eta^*}^2) = N(0, (1/4\alpha - \sigma_w^2)^{1/2}).$

95 - 33

5

Ś

X

(iii) δ^* is the Bayes estimator for u under the least favorable distribution μ^* for v, computed as

$$\delta^{*}(z) = p^{*}z = \begin{cases} (P_{o}(1+\lambda^{*})/[P_{o}^{2}(1+\lambda^{*})^{2}+\sigma_{w}^{2}])z & \text{in } R_{33}^{1} \\ \\ (1/P_{o})z & \text{in } R_{33}^{2} \end{cases}$$

Proof. We shall establish separately the validity of the RHS and the LHS inequalities of (7.18) in Regions R_{33}^{1} and R_{33}^{2} .

Region R_{33}^{1}

(i) The RHS inequality:

With the jammer's policy fixed as indicated, the problem faced by the encoder-decoder pair is

$$\min_{\gamma,\delta} E\{(\delta(z)-u)^2 - \alpha \lambda^{*2} \gamma^2(u)\}, E[\gamma^2(u)] \leq P_o^2$$

where $z = (1+\lambda^*)\gamma(u)+w$. Noting that $\lambda^* \in (-1,0)$ it follows that

$$\min_{\substack{\gamma,\delta}} E\{(\delta(z) - u)^2\}$$

under $E[\gamma^{2}(u)] \leq P_{o}^{2}$ is achieved by $\gamma(u) = P_{o}u$.

Now $\alpha \lambda^{*2} E\{\gamma^2(u)\}$ is also maximized when $\gamma(u) = P_0 u$ and therefore we obtain the solution of the above problem as

$$\gamma(u) = P_0 u$$

and

$$\delta(z) = E[u | z] = [P_0(1 + \lambda^*)/(P_0^2(1 + \lambda^*)^2 + \sigma_w^2)] z$$

(ii) The LHS inequality:

With the encoder-decoder policies fixed as indicated, the problem faced by the jammer is

$$\max_{\beta} E[(p^{*2}-\alpha)\beta^{2}(y) + 2p^{*}(p^{*}P_{o}-1)\beta(y)u],$$

the unique solution to which is

$$\beta^{*}(y) = [p^{*}(p^{*}P_{o}-1)/(P_{o}(\alpha-p^{*}))] y = \lambda^{*}y ,$$

provided that

$$p^{*^2}-\alpha<0$$

or equivalently

$$-\lambda^*(1+\lambda^*) < \sigma_w^2/P_o^2 . \qquad (7.19)$$

We now show that the inequality (2) holds for all P_o^2 in R_{33}^1 .

(a) First consider the case $P_o^2 = 1/\alpha < 4\sigma_w^2$. Note that under this condition the equation $\lambda^{*2} + \lambda^* + \sigma_w^2/P_o^2 = 0$ has no real roots, which implies that

$$\lambda^{*^2} + \lambda^* + \sigma_w^2 / P_o^2 > 0$$

and hence $-\lambda^*(1+\lambda^*) < \sigma_w^2/P_o^2$.

(b) Now, consider the case $P_o^2 \neq 1/\alpha$. If $P_o^2 = 1/\alpha$ we know that λ satisfies

$$[(1+\lambda)^{2}P_{o}^{2} + \sigma_{w}^{2}]^{2} + (\sigma_{w}^{2}(1+\lambda)/\lambda)P_{o}^{2} = 0 , \qquad (7.20)$$

which after some manipulations yields

$$g(\lambda) = (\lambda(1+\lambda) + \sigma_{w}^{2}/P_{o}^{2})(\sigma_{w}^{2}/P_{o}^{2} + (1+\lambda)^{3}/\lambda) = 0 \quad .$$
 (7.21)

This function is illustrated in Figures 7.4a and 7.4b. The extrema are found at the intersections of the horizontal line at αA , the necessary condition being

159



5

161

$$\alpha = -\sigma_{\mathbf{w}}^{2}(1+\lambda)/[\lambda((1+\lambda)^{2}P_{o}^{2} + \sigma_{\mathbf{w}}^{2})]^{2} \underline{\Delta} f(\lambda) .$$

The quadratic term of (7.21) corresponds to the maximizing roots, whereas the cubic term in (7.21) has only one real root which is the minimizing solution.

Let

$$A(\lambda) \triangleq \lambda(1+\lambda) + \sigma_w^2/P_o^2$$
 (with roots λ_1^A and λ_2^A),

and

$$B(\lambda) \triangleq \sigma_w^2/P_o^2 + (1+\lambda)^3/\lambda$$
 (with real root λ^B),

so that Equation (7.21) may be rewritten as

 $A(\lambda) \cdot B(\lambda) = 0$.

From Figure 7.4 it is apparent that if $\alpha > 1/P_o^2$, we get a maximizing root smaller than λ_1^A , which satisfies (7.19), and if $\alpha < 1/P_o^2$, we get a maximizing root larger than λ_2^A , which satisfies (7.19), and thus the proof for Region R_{33}^1 is complete.

Region R_{33}^2

(i) The RHS inequality:

With the jammer's policy as indicated, the problem faced by the encoder-decoder pair is

$$\min_{\substack{\gamma,\delta}} E\{(\delta(z)-u)^2 - \alpha \lambda^{*2} \gamma^2(u) - \alpha \sigma_{\eta^*}^2\}, E[\gamma^2(u)] \leq P_o^2$$

with $z = (1+\lambda^*)\gamma(u) + w + \eta^*$.

It follows then, as in the proof for Region R_{33}^{-1} , that the required solution is

 $\gamma(u) = P_0 u$

and

$$\delta(z) = E[u + z] = (1/P_0) z$$

8

(ii) The LHS inequality:

$$J(\gamma^*,\delta^*,\beta) \leq J(\gamma^*,\delta^*,\beta^*)$$
.

With the encoder-decoder policies fixed as indicated we have $p^*P_o = 1$ and $p^{*2} = \alpha$. Therefore, the maximization problem faced by the jammer, i.e.,

$$\max_{\boldsymbol{\beta}(\mathbf{y})} E[(p^{*2} - \alpha)\boldsymbol{\beta}^{2}(\mathbf{y}) + 2p^{*}(p^{*}P_{o} - 1)\boldsymbol{\beta}(\mathbf{y})u],$$

has a cost function which is independent of β which may, therefore, be chosen as in the statement of the theorem.

Remark 7.1. This solution for the jamming policy in Region R_{33}^2 is not unique. In fact, any pair (λ^*, σ_n) which satisfies

$$(1+\lambda^{*})^{2} - (1+\lambda^{*}) + \alpha(\sigma_{w}^{2}+\sigma_{\eta^{*}}^{2}) = 0$$

provides a solution, since this gives

$$p^* = (1+\lambda^*)P_o/((1+\lambda^*)^2P_o^2 + \sigma_w^2 + \sigma_{\eta^*}^2) = 1/P_o ,$$

thus yielding a cost which is independent of β . The specific choice of the solution pair in the statement of the theorem is motivated by continuity of λ^* at the boundary of R_{33}^1 and R_{33}^{2} .

(c) Game G_{34}

It turns out that Game G_{34} does not admit a saddle-point solution and hence the general approach used so far is not applicable. This is because the maximin value of the problem is not well'defined. (In order to find the maximin value, we need to fix the jammer's policy and then minimize over the encoder-decoder policy; but the jamming policy itself depends on the encoder policy through the constraint, thus leading to ill-posedness.)

However, we can still obtain a minimax solution by proceeding in two stages. First, we formulate a related problem with an enlarged information structure for which we can show the existence of a saddle-point solution. We then show that the minimax value of the original problem and that of the problem with the enlarged information structure are identical, and hence the saddle-point solution found corresponds to the minimax solution desired.

Towards this end, suppose that in addition to having access to the random variable $y = \gamma(u)$, the jammer also has access to the policy γ , in other words, the jammer's policy $\tilde{\beta}$ belongs to the space of all Borel measurable mappings which are functions of both y and γ , i.e., $v = \tilde{\beta}(y,\gamma)$. The saddle-point solution to the new problem with this enlarged information structure is presented below in Proposition 7.1 after introducing some notation.

Preliminary notation for Proposition 7.1.

Let

$$P^*_{o} = \underset{P_{o} \ge 0}{\operatorname{arg min}} f(P_{o})$$

where

$$f(P_{o}) = \begin{cases} k_{o}P_{o}^{2} + \sigma_{w}^{2}/[(P_{o}-P_{J})^{2}+\sigma_{w}^{2}] & \text{if } P_{o} \ge P_{J} \\ \\ \\ k_{o}P_{o}^{2} + 1 & \text{if } P_{o} < P_{J} \end{cases}$$

Note that either $P_{o}^{*} = 0$ or $P_{o}^{*} > P_{J}$.

Furthermore, decompose the parameter space into Regions R_{34}^1 and R_{34}^2 characterized by

163

N N

N.

ζ.

2

マス

164

$$R_{34}^{1}$$
: $P_{0}^{*} > P_{J}$
 R_{34}^{2} : $P_{0}^{*} = 0$

Proposition 7.1. The problem with the enlarged information structure, just described, admits two saddle-point solutions $(\gamma^*, \delta^*, \mu^*)$ and $(-\gamma^*, -\delta^*, \mu^*)$ where

(i) $\gamma^*(u) = P^*_o u$

(ii) μ^* is the Gaussian probability measure associated with the random variable

$$\mathbf{v} = \tilde{\beta}^*(\mathbf{y}, \boldsymbol{\gamma}) = \begin{cases} -(\mathbf{P}_{\mathbf{j}} / \|\mathbf{y}\|) \ \mathbf{y} \ \text{in } \mathbf{R}_{\mathbf{34}}^1 \\ \\ -\mathbf{y} \ \text{in } \mathbf{R}_{\mathbf{24}}^1 \end{cases}$$

(iii) $\delta^*(z) = p^* z$

$$= \begin{cases} [(P_{o}^{*}-P_{J})/((P_{o}^{*}-P_{J})^{2}+\sigma_{w}^{2})]z & \text{in } R_{34}^{1} \\ \\ 0 & \text{in } R_{34}^{2} \end{cases}$$

Proof. We shall establish separately the validity of the RHS and the LHS inequalities of (7.18) in R_{34}^{1} and R_{34}^{2} . (The key observation here is that the jammer must know his opponent's policy in order to determine a minimax solution. The enlarged information structure enables the jammer to use the policies as in (ii) above.)

Region R_{34}^{-1} .

(i) The RHS inequality:

The problem faced by the encoder-decoder pair, with the jammer's policy fixed as indicated, is

 $\min_{\substack{\gamma,\delta}} E[k_{o}\gamma^{2}(u) + (\delta(z)-u)^{2}]$

where

$$z = (1 - P_J / ||y||)y + w$$

Let us first consider the Problem P1:

$$\min_{\substack{\gamma,\delta}} E\{(\delta(z) - u)^2\} ; E[\gamma^2(u)] \leq P_o^2.$$

We assert that the minimum value of this problem is achieved by some γ satisfying $E[\gamma^2(u)] = P_o^2$. To verify this, we first assume the contrary, that the minimum is achieved by a γ satisfying $E[\gamma^2(u)] = P^2$, where $0 < P_J < P < P_o$. The solution to the Problem P2:

min
$$E\{(\delta(z)-u)^2\}$$
; $E[\gamma^2(u)] = P^2$,
 γ,δ

is then the same as that for the Problem P1. However, for Problem P2 we know that

$$z = (1 - P_{I}/P)\gamma(u) + w .$$

By defining $\gamma'(u) = (1-P_J/P)\gamma(u)$, and using Lemma 7.3, we find that the solution to Problem P2 is linear and yields an optimal performance of

$$\sigma_{w}^{2}/((P-P_{1})^{2}+\sigma_{w}^{2})$$

By using a linear solution $y = P_0$ u for Problem P1, we obtain the performance

$$\sigma_{\rm w}^2 / ((P_{\rm o} - P_{\rm I})^2 + \sigma_{\rm w}^2)$$

which is clearly lower than the one obtained for Problem P2, implying that the minimum cannot be achieved for $E[\gamma^2(u)] = P^2 < P_o^2$. We now have

$$\min_{\substack{\gamma,\delta\\ (\mathbf{E}|\gamma^{2}(\mathbf{u})| = P_{o}^{2} < P_{J}^{2})}} E[k_{o}\gamma^{2}(\mathbf{u}) + (\delta(z) - \mathbf{u})^{2}]$$

$$\geq k_{o}P_{o}^{2} + \sigma_{w}^{2}/[(P_{o} - P_{J})^{2} + \sigma_{w}^{2}]$$

$$\geq \min_{P_{o} > P_{J}} (k_{o}P_{o}^{2} + \sigma_{w}^{2}/[(P_{o} - P_{J})^{2} + \sigma_{w}^{2})]$$

$$= k_{o}P_{o}^{*2} + \sigma_{w}^{2}/[(P_{o}^{*} - P_{J})^{2} + \sigma_{w}^{2}]$$

where $P_{o}^{*} > P_{J}$ necessarily exists in R_{34}^{1} . With $\gamma(u) = P_{o}^{*}u$ and $\delta(z) = E[u \mid z]$ the above inequalities are all tight, implying that the minimum is indeed achieved by the policies given in the statement of the theorem.

(ii) The LHS inequality:

With γ^* and δ^* fixed as indicated, the problem faced by the jammer is

$$\max_{\beta} \{p^{*2}\beta^{2}(y) + 2p^{*}(p^{*}P^{*}-1)\beta(y)u\},\$$

subject to the constraint $E[\beta^2(y)] \leq P_J^2$. Clearly the optimal policy for the jammer is to use a function, linear in u, at the maximum permissible power level, which, under the enlarged information structure, is

$$\tilde{\beta}^{*}(\mathbf{y},\boldsymbol{\gamma}) = -(\mathbf{P}_{\mathbf{j}}/||\mathbf{y}||) \mathbf{y} .$$

Region R_{34}^2

(i) The RHS inequality:

With $\beta^{*}(y) = -y$, the problem faced by the encoder-decoder pair is

$$\min_{\substack{\gamma,\delta}} E[k_{o}\gamma^{2}(u) + 1] ,$$

the solution to which clearly requires $\gamma(u) = 0$ and

$$\delta(z) = E[u \mid z] = 0$$

(ii) The LHS inequality:

With the encoder-decoder policies fixed at zero, the cost becomes independent of the jammer's policy, who may now use $\beta(y) = -y$ as the policy which yields the saddle-point solution.

We finally argue that the minimax values of the two problems with and without the enlarged information structure are the same. With γ fixed, let β belong to the space of Borel measurable functions of y satisfying $E[\beta^2(y)] \leq P_j^2$, and let $\tilde{\beta}$ be in the space of Borel measurable functions of y and γ satisfying $E[\tilde{\beta}^2(y,\gamma)] \leq P_j^2$. Then

$$\min_{\gamma,\delta} \max_{\beta} J(\gamma,\delta,\beta) = \min_{\gamma,\delta} \max_{\beta} J(\gamma,\delta,\beta) ,$$

$$\sum_{\gamma,\delta} \sup_{\beta} J(\gamma,\delta,\beta) = \min_{\gamma,\delta} \max_{\beta} J(\gamma,\delta,\beta) ,$$

since for fixed y and δ the two inner maximization problems are identical.

We therefore have Theorem 7.4 below.

Theorem 7.4. Game G_{34} admits a minimax solution which is equivalent to the saddle-point solution given in Proposition 7.1.

666661 • 3032669 • **320003**9

7.5. Minimax and Maximin Strategies for Games G_{22} , G_{23} and G_{24}

7.5.1. Minimax strategies

If the encoding policy γ is restricted to be deterministic, then to every jamming policy β that is a function of the encoder output $\gamma(u)$, there corresponds a jamming policy $\tilde{\beta} = \beta \cdot \gamma$ which represents the same random variable

$$\mathbf{v} = \boldsymbol{\beta}(\mathbf{u}) = \boldsymbol{\beta}(\boldsymbol{\gamma}(\mathbf{u})) \,. \tag{7.22}$$

਼ੁ

X

Ř

5

S.

Ň

168

Since γ is not necessarily invertible, we have the inequality

$$\sup_{\beta} J(\gamma, \delta, \beta(\gamma(u)) \leq \sup_{\tilde{\beta}} J(\gamma, \delta, \tilde{\beta}(u))$$
(7.23)

for every pair (γ, δ) , where γ is restricted to be deterministic. Taking the infimum of both sides over (γ, δ) we obtain for k=2,3,4,

$$\overline{J}_{3k} = \inf_{\substack{(\gamma,\delta) \\ \beta}} \sup_{\beta} J_k(\gamma,\delta,\beta(\gamma(u)) \leq \overline{J}_{2k}$$
(7.24)

where \overline{J}_{3k} is the upper value of a similar game with fidelity criterion C_k , with the difference that in this new game the jammer has access to the output of the encoder rather than the input. Thus the upper value of a game of Type 2 is bounded from below by the upper value of the corresponding game of Type 3. We shall next show that this inequality is in fact an equality, and provide for each $k \in \{2,3,4\}$, a set of minimax strategies that achieve this value.

Preliminary notation for Theorem 7.5

(a) $Game G_{2}$

Define Regions \overline{R}_{22}^{1} , \overline{R}_{22}^{2} and \overline{R}_{22}^{3} by

$\bar{\mathbf{R}}_{22}$: $\alpha \leq \mathbf{k}_0$	
$\overline{R}_{22}^2: \alpha > k_0 \text{ and } \alpha k_0 - k_0^2 < \alpha^3 \sigma_w^2$	(7.25)
$\overline{R}_{22}^{3}: \alpha > k_{0} \text{ and } \alpha k_{0} - k_{0}^{2} \ge \alpha^{3} \sigma_{w}^{2}.$	

169

Now define $P{\color{red}{\ast}_0}$ and $\lambda{\color{red}{\ast}}$ by

$$\mathbf{P}_{0}^{*} = \begin{cases} 0 & \text{in } \overline{R}_{22}^{1} \\ (\operatorname{Max}\{0, \sigma_{\mathbf{w}}^{-1} - \alpha^{-1})^{\frac{1}{2}} - \sigma_{\mathbf{w}}^{-1})^{\frac{1}{2}} - \sigma_{\mathbf{w}}^{-1})^{\frac{1}{2}} & \text{in } \overline{R}_{22}^{2} \\ \alpha^{-\frac{1}{2}} & \text{in } \overline{R}_{22}^{3} \end{cases}$$
(7.26)

$$\lambda^* = \begin{cases} -1 & \text{in } \overline{R}_{22}^1 \\ -k_0/\alpha & \text{in } \overline{R}_{22}^3 \text{ and } \overline{R}_{22}^3 \end{cases}$$
(7.27)

222222

• <u>5 5 55 1</u> • <u>55 22 23</u> •

Let $(\overline{\gamma}^*, \overline{\delta}^*, \overline{\beta}^*)$ be defined by

$$\bar{\gamma}^{*}(u) = P^{*}_{0}u \quad \text{in } \bar{R}^{1}_{22}, \bar{R}^{2}_{22} \text{ and } \bar{R}^{3}_{22}$$
 (7.28)

$$\delta^{*}(z) = p^{*}z = \begin{cases} (P^{*}_{0}(1+\lambda^{*})/(P^{*}_{0}^{2}(1+\lambda^{*})^{2}+\sigma_{w}^{2}))z & \text{ in } \mathbb{R}_{22}^{1} \text{ and } \mathbb{R}_{22}^{2} \\ (1/P^{*}_{0})z & \text{ in } \mathbb{R}_{22}^{3} \end{cases}$$
(7.29)

$$\overline{\beta}^{*}(u) = \begin{cases} 0 & \text{in } \overline{R}_{22}^{2} \\ -(k_{0}/\alpha)P^{*}_{0}u & \text{in } \overline{R}_{22}^{2} \\ -(\frac{k_{0}}{\alpha})P^{*}_{0}u + \eta^{*} & \text{in } \overline{R}_{22}^{3} \end{cases}$$
(7.30)

where η^* is an independent Gaussian random variable, $\eta^* \sim N(0,k_0/\alpha^2 - k_0^2/\alpha^3 - \sigma_w^2)$.

(b) Game G₂₃

Define Regions $\overline{R}_{23}^{\ 1}$ and $\overline{R}_{23}^{\ 2}$ by

$$\overline{R}_{23}^{1}: (P_{0}^{2} \neq \frac{1}{\alpha}) \cup (P_{0}^{2} = \frac{1}{\alpha} < 4\sigma_{w}^{2})$$

$$\overline{R}_{23}^{2}: P_{0}^{2} = \frac{1}{\alpha} \ge 4\sigma_{w}^{2}.$$
(7.31)

Now define $\lambda * by$

$$\lambda^{*} = \begin{cases} \arg \max_{\lambda} \left(\sigma_{w}^{2} / ((1+\lambda)^{2} P_{0}^{2} + \sigma_{w}^{2}) - \alpha \lambda^{2} P_{0}^{2} \right) & \text{in } \overline{R}_{23}^{1} \\ -\frac{1}{2} & \text{in } \overline{R}_{23}^{1} \end{cases}$$
(7.32)

Let $(\overline{\gamma}^*, \overline{\delta}^*, \overline{\beta}^*)$ be defined by

$$\overline{\mathbf{y}}^{*}(\mathbf{u}) = \mathbf{P}_{0}\mathbf{u}$$
 in $\overline{\mathbf{R}}_{23}^{1}$ and $\overline{\mathbf{R}}_{23}^{2}$ (7.33)

лу Х

يت بر

'n

22

Х Х

 $\mathbf{F}_{\mathbf{x}}^{\mathbf{x}}$

ğ

i.N

E.

 $\frac{1}{20}$

S.

ર્

5

্বাম আন
$$\delta^{*} = \begin{cases} (P_{0}(1+\lambda^{*})/(P_{0}^{2}(1+\lambda^{*})^{2}+\sigma_{w}^{2}))z & \text{in } \mathbb{R}_{23}^{1} \\ (1/P_{0})z & \text{in } \overline{\mathbb{R}}_{23}^{2} \end{cases}$$
(7.34)

$$\overline{\beta}^{*} = \begin{cases} \lambda^{*} P_{0} u & \text{in } \mathbb{R}_{23}^{1} \\ \lambda^{*} P_{0} u + \eta^{*} & \text{in } \overline{\mathbb{R}}_{23}^{2} \end{cases}$$
(7.35)

where η^* is an independent Gaussian random variable, $\eta^* \sim N(0, 1/4\alpha - \sigma_w^2)^{\frac{1}{2}})$.

(c) Game G₂₄

Let

$$P_0^* \equiv \arg\min_{\substack{P_0 \ge 0}} f(P_0) \tag{7.36}$$

where

170

 $f(P_0) \equiv \begin{cases} k_0 P_0^2 + \sigma_w^2 / ((P_0 - P_J)^2 + \sigma_w^2) & \text{if } P_0 \ge P_J \\ k_0 P_0^2 & \text{if } P_0 < P_J \end{cases}$ (7.37)

Note that either $P_0^* > P_J$ or $P_0^* = 0$.

Define Regions $\overline{R}_{24}^{\ 1}$ and $\overline{R}_{24}^{\ 2}$ by

$$\overline{R}_{24}^{1}: P_{0}^{*} > P_{J}$$

$$\overline{R}_{24}^{2}: P_{0}^{*} = 0.$$
(7.38)

Let $(\overline{\gamma}^*, \overline{\delta}^*, \overline{\beta}^*)$ be defined by

$$\overline{\mathbf{y}}^{*}(\mathbf{u}) = \mathbf{P}^{*}_{0}\mathbf{u} \tag{7.39}$$

$$\delta^{*}(u) = \begin{cases} ((P^{*}_{0} - P_{J})/((P^{*}_{0} - P_{J})^{2} + \sigma_{w}^{2}))z & \text{in } \mathbb{R}_{24}^{1} \\ 0 & \text{in } \mathbb{R}_{24}^{2} \end{cases}$$
(7.40)

$$\overline{\beta}^{*}(u) = \begin{cases} -P_{J}u & \text{in } R_{24}^{1} \\ 0 & \text{in } \overline{R}_{24}^{2} \end{cases}$$
(7.41)

Theorem 7.5. Sets of minimax solutions for games G_{22} , G_{23} and G_{24} are provided by $(\overline{\gamma}^*, \overline{\delta}^*, \overline{\beta}^*)$ and $(-\overline{\gamma}^*, -\overline{\delta}^*, -\overline{\beta}^*)$ with $\overline{\gamma}^*, \overline{\delta}^*$ and $\overline{\beta}^*$ defined above in the various regions under each of the three fidelity criteria.

Proof. We shall first establish that

$$\overline{J}_{2k} \leq \sup_{\beta} J_{k}(\overline{\gamma}^{*}, \overline{\delta}^{*}, \beta(u)) = J_{k}(\overline{\gamma}^{*}, \overline{\delta}^{*}, \overline{\beta}^{*}) = \overline{J}_{3k}.$$
(7.42)

ŝ Δ. 8 S 5. 19 , î 17.10C

S.

5

Ś

The first inequality in (7.42) is immediate because

$$\overline{J}_{2k} \leq \sup_{\beta} J_{k}(\gamma, \delta, \beta(u))$$

for all pairs (y, \delta). \overline{J}_{3K} is the minimax value for the corresponding game of Type 3, and we shall prove the equality in (7.42) by considering individually the various regions under each of the criteria and comparing with the minimax strategies for problems of Type 3 available from Section 7.4.

(a) Game G_{22}

(i) Region \overline{R}_{22}^{1}

$$J_{2}(\overline{\mathbf{y}},\overline{\mathbf{\delta}},\beta) = E[1-\alpha\beta^{2}(\mathbf{u})].$$

Therefore, $\overline{\beta}$ *(u) = 0 and

$$\sup_{\beta} J_2(\bar{\gamma}^*, \bar{\delta}^*, \beta(u)) = \bar{J}_{32} \text{ in } \bar{R}_{22}^1.$$

(ii) Region \overline{R}_{22}^{2}

The problem faced by the jammer is

$$\max_{\beta} E[(p^{*2}-\alpha)\beta^{2}(u)+2p^{*}(p^{*}P^{*}_{0}-1)\beta(u)u]$$

for which the solution is

$$\overline{\boldsymbol{\beta}}^{*}(\mathbf{u}) = -(\mathbf{k}_{0}/\alpha)\mathbf{P}^{*}_{0}\mathbf{u} .$$

Noting that this provides the same value as the minimax val e for Game G_{32} in Region $\overline{R}_{22}^{(2)}$, we have

$$\sup_{\beta} J_2(\overline{\gamma}^*,\overline{\delta}^*,\beta(n)) = \overline{J}_{32} \quad \text{in } \overline{R}_{22}^2.$$

(iii) Region \overline{R}_{22}^{3}

The problem faced by the jammer is

$$\max_{\beta} E[(p^{*2} - \alpha)\beta^{2}(u) + 2p^{*}(p^{*}P^{*}_{0} - 1)\beta(u)u],$$

and the cost is independent of β , the value being the same as the minimax value for Game G_{32} in \overline{R}_{22}^{3} , i.e.,

$$\sup_{\beta} J_2(\overline{\gamma}^*, \overline{\delta}^*, \beta(u)) = \overline{J}_{32} \text{ in } \overline{R}_{22}^3.$$

We remark that while β may be chosen arbitrarily to satisfy the equality, the specific choice of $\overline{\beta}^*$ as indicated provides the saddle-point solution for Game G₂₃ and further provides continuity of the jamming policy at the boundary of Regions \overline{R}_{22}^2 and \overline{R}_{22}^3 .

- (b) Game G_{23}
- (i) Region \overline{R}_{23}^{1}

The problem faced by the jammer is

$$\max_{\beta} E[(p^{*2}-\alpha)\beta^{2}(u)+2p^{*}(p^{*}P_{0}-1)\beta(u)u],$$

the solution to which is

$$\bar{\beta}^{*}(u) = \frac{p^{*}(p^{*}P_{0}-1)}{(\alpha-p^{*})} u = \lambda^{*}P_{0}u.$$

Since this solution provides the same value as the minimax strategy for $\rm G_{33}$ we have

$$\sup_{\beta} J_3(\overline{\gamma}^*,\overline{\delta}^*,\beta(u)) = \overline{J}_{33} \text{ in } \overline{R}_{23}^1.$$
The problem faced by the jammer is to minimize a cost function which is independent of β as in the case of Region \overline{R}_{22}^3 and we have

$$\sup_{\beta} J_3(\overline{\gamma}^*, \overline{\delta}^*, \beta(u)) = \overline{J}_{33} \text{ in } \overline{R}_{23}^2.$$

Again β may be chosen arbitrarily; the specific choice here is due to the same reasons as remarked for Region \overline{R}_{22}^{3} .

(c) Game G_{24}

(i) Region \overline{R}_{24}^{1}

The problem faced by the jammer is

$$\max_{\beta} E\{p^{*2}\beta^{2}(u)+2p^{*}(p^{*}P^{*}_{0}-1)\beta(u)u\},\$$

subject to $E[\beta^2(u)] \leq P_J^2$.

Clearly, the optimal policy for the jammer is to use a linear function of u at the maximum permissible power level, which gives

$$\overline{\beta}^{*}(u) = -P_{1}u$$

and provides the same value as the minimax strategy for Game G_{34} , implying that

$$\sup_{\beta} J_4(\overline{\gamma}^*, \overline{\delta}^*, \beta(u)) = \overline{J}_{34} \text{ in } \overline{R}_{24}^1.$$

(ii) Region \overline{R}_{24}^{2}

With the encoding-decoding policy pair fixed at zero, the cost is independent of the jammer's policy which may be chosen arbitrarily to provide

$$\sup_{\beta} J_4(\overline{\gamma}^*, \overline{\delta}^*, \beta(u)) = \overline{J}_{34} \text{ in } \overline{R}_{34}^2.$$

We now see that in all cases we have

$$\sup_{\beta} J_{k}(\overline{\gamma}^{*},\overline{\delta}^{*},\beta(u)) = \overline{J}_{3k},$$

and thus (7.42) holds.

Using (7.42) along with (7.24) it follows that $\overline{J}_{2k} = \overline{J}_{3k}$ for $k \in \{2,3,4\}$ and that the strategy sets given are sets of minimax policies.

11772222C

7.5.2. Maximin Strategies

Since $\Gamma_e \supset \Gamma_{ed}$ we have

$$\inf_{\gamma \in \Gamma_{k}, \delta \in \Gamma_{k}} J_{k}(\gamma, \delta, \beta) \leq \inf_{\gamma \in \Gamma_{k}, \delta \in \Gamma_{k}} J_{k}(\gamma, \delta, \beta)$$
(7.43)

for all $\beta \in \Gamma_j$. Taking the supremum of both sides over $\beta \in \Gamma_j$, we get

$$\underline{J}_{1k} \leq \underline{J}_{2k} , \qquad (7.44)$$

i.e., the maximin value for a game of Type 2 is bounded from below by the maximin value of the corresponding game of Type 1 (which is also the saddle-point value of the game of Type 1).

Preliminary notation for Theorem 7.6

(a) Game G_{22}

Define Regions R_{12}^{1} , R_{12}^{2} and R_{12}^{3} as in (7.10), and define (P_{0}^{*}, P_{J}^{*}) by (7.11).

÷ •

, , ,

٧,

(b) Game G_{23}

Define Regions R_{13}^{1} and R_{13}^{2} as in (7.12) and define (P_{0}^{*}, P_{J}^{*}) by (7.13).

(c) Game G_{24}

Define Regions R_{14}^{1} and R_{14}^{2} as in (7.14) and define (P_{0}^{*}, P_{J}^{*}) by (7.15).

Theorem 7.6. Sets of maximin solutions for games G_{22} , G_{23} and G_{24} are provided by $(\underline{\gamma}^*, \underline{\delta}^*, \underline{\beta}^*)$ and $(-\underline{\gamma}^*, -\underline{\delta}^*, \underline{\beta}^*)$ where

$$\underline{\boldsymbol{\gamma}}^{\ast}(\mathbf{u}) = \mathbf{P}^{\ast}_{0}\mathbf{u} \tag{7.45a}$$

$$\underline{\delta}^{*}(z) = (P_{0}^{*}/(P_{0}^{*2} + P_{J}^{*2} + \sigma_{w}^{2}))z$$
(7.45b)

and

$$\boldsymbol{\beta^{\ast}(u) = \eta^{\ast}} \tag{7.45c}$$

where η^* is a zero mean Gaussian random variable with variance $P^{*_J^2}$ which is independent of u and w.

Proof. Note that

$$\frac{J_{2k}}{\beta \in \Gamma_{i}^{k}} = \sup_{\gamma \in \Gamma_{ed}^{k}} \inf_{\substack{\beta \in \Gamma_{i}^{k} \ \gamma \in \Gamma_{ed}^{k} \\ \delta \in \Gamma_{i}^{k}}} \inf_{\substack{\beta \in \Gamma_{i}^{k} \ \gamma \in \Gamma_{ed}^{kl} \\ \delta \in \Gamma_{i}^{k}}} \inf_{\substack{\beta \in \Gamma_{ed}^{kl} \\ \delta \in \Gamma_{i}^{kl}}} (7.46)$$

where Γ_{ed}^{kl} and Γ_{r}^{kl} are the sets of all permissible *linear* policies for the encoder and the decoder, respectively, and the inequality follows because the infimum on the right side is taken over a smaller set. We next show that the expression on the right side of the inequality in (7.46) is \underline{J}_{1k} , i.e., the maximin (or saddle-point) value for the corresponding game of Type 1.

(a) Game G_{22}

Let $\gamma(u) = \in u$ and $\delta(z) = \Delta z$. Then

$$J_{2}(\gamma,\delta,\beta) = (1-\Delta\epsilon)^{2} + \Delta^{2}\sigma_{w}^{2} + \Delta^{2}E[\beta^{2}(u)] + 2\Delta(\epsilon\Delta-1)E[u\beta(u)] - \alpha E[\beta^{2}(u)] + k_{0}\epsilon^{2}.$$

Consider the minimization of the above functional over (ϵ, Δ) . Using $\frac{\partial J_2}{\partial \Delta}|_{\Delta *} = 0$ and

noting that $\frac{\partial^2 J_2}{\partial \Delta^2} > 0$, we obtain

$$\Delta^* = \frac{\epsilon + E[u\beta(u)]}{\epsilon^2 + E[\beta^2(u)] + \sigma_w^2 + 2\epsilon E[u\beta(u)]}$$

as the unique minimizing Δ^* for every fixed \in . Therefore,

$$J_{2}|_{\Delta *} = K_{0} \epsilon^{2} + \frac{E[\beta^{2}(u)] + \sigma_{w}^{2} - (E[u\beta(u)])^{2}}{\epsilon^{2} + E[\beta^{2}(u)] + \sigma_{w}^{2} + 2\epsilon E[u\beta(u)]} - \alpha E[\beta^{2}(u)] + \sigma_{w}^{2} + 2\epsilon E[u\beta(u)]$$

We next minimize $J_2|_{\Delta_*}$ with respect to \in . In particular, if $E[u\beta(u)] = 0$ we get

$$\in *^{2} = Max \left\{ 0, \left(\frac{E[\beta^{2}(u)] + \sigma_{w}^{2}}{k_{0}} \right)^{\frac{1}{2}} - (E[\beta^{2}(u)] + \sigma_{w}^{2}) \right\}$$

and, in general, we have

$$J|_{\Delta^{*}, \epsilon^{*}} = k_{0} \epsilon^{*^{2}} + \frac{E[\beta^{2}(u)] + \sigma_{w}^{2} - (E[u\beta(u)])^{2}}{\epsilon^{*^{2}} + E[\beta^{2}(u)] + \sigma_{w}^{2} + 2\epsilon^{*}E[u\beta(u)]} - \alpha E[\beta^{2}(u)]$$

and we note that $sgn(\in *) = -sgnE[u\beta(u)]$.

To maximize $J|_{\Delta^{\bullet}, \epsilon^{\bullet}}$ as a function of $E[u\beta(u)]$ we differentiate with respect to $E[u\beta(u)]$ and find that

ĥ

(5. 38

ŝ

5

3

٣,

178

 $\frac{dJ|_{\Delta^{\bullet}, \epsilon^{\bullet}}}{d(E[u\beta(u)])}$ if $E[u\beta(u)] < 0$ > 0

and

$$\frac{dJ[_{\Delta^{\bullet},\varepsilon^{\bullet}}}{d(E[u\beta(u)])} < 0 \quad \text{if } E[u\beta(u)] > 0.$$

Therefore, $J|_{\Delta^{\bullet},\alpha^{\bullet}}$, which is continuous in $E[u\beta(u)]$, admits its maximum at $E[u\beta(u)] = 0$ and we next maximize over $E[\beta^2(u)]$ to find that

$$E[\beta^{2}(u)] = Max\{0, \frac{\epsilon^{*}}{\alpha^{\frac{1}{2}}} - (\epsilon^{*} + \sigma_{w}^{2})\}$$

Note that if $\in *=0$, then $E[\beta^2(u)]$ is necessarily zero, which is the case if $\frac{1}{k_0} \leq \sigma_w^2$, i.e.,

Region R_{12}^{1} .

Next note that \in * could be nonzero with $E[\beta^2(u)] = 0$, which occurs if

$$\varepsilon^{*2} = \frac{\sigma_{w}}{k_{0}^{\frac{1}{2}}} - \sigma_{w}^{2}$$

along with

$$\frac{\epsilon^{*^2}}{\alpha} < (\epsilon^{*} + \sigma_w^2)^2$$

which together yield

$$\frac{k_0}{\left(k_0+\alpha\right)^2} \leqslant \sigma_w^2 < \frac{1}{k_0}$$

i.e., we have Region R_{12}^2 .

Finally note that in Region R_{12}^3 where

 $\sigma_w^2(k_0+\alpha)^2 < k_0,$

both $\in *$ and $E[\beta^2(u)]$ are nonzero.

Observing that the jammer's maximization problem is solved by using any secondorder random variable with the appropriate variance, it follows that a Gaussian random variable as indicated is an appropriate choice. Thus, with the encoding-decoding policy restricted to be linear, the cost obtained by using the jammer's maximinimizing policy is the same as the maximin value of Game G_{12} ,

(b) Game G_{23}

With $\gamma(u) = \in u$ and $\delta(z) = \Delta z$

$$J_{3}(\gamma,\delta,\beta) = (1-\Delta\epsilon)^{2} + \Delta^{2}\sigma_{w}^{2} + \Delta^{2}E[\beta^{2}(u)] + 2\Delta(\epsilon\Delta-1)E[u\beta(u)] - \alpha E[\beta^{2}(u)]$$

and we obtain as before

$$J_{3}|_{\Delta^{*}} = \frac{E[\beta^{2}(u)] + \sigma_{w}^{2} - (E[u\beta(u)])^{2}}{\epsilon^{2} + E[\beta^{2}(u)] + \sigma_{w}^{2} + 2\epsilon E[u\beta(u)]} - \alpha E[\beta^{2}(u)]$$

and minimization of $J|_{\Delta \ast}$ with respect to \in yields

$$\epsilon^{*} = \begin{cases} P_{0} & \text{if } E[u\beta(u)] > 0 \\ -P_{0} & \text{if } E[u\beta(u)] < 0 \\ P_{0} \text{ or } -P_{0} & \text{if } E[u\beta(u)] = 0 \end{cases}$$

(Recall that we require $|\epsilon^2| \leq P_0^2$). Now,

 $J_{3}|_{\Delta^{\bullet}, \epsilon^{\bullet}} = \begin{cases} \frac{E[\beta^{2}(u)] + \sigma_{w}^{2} - (E[u\beta(u)])^{2}}{P_{0}^{2} + E[\beta^{2}(u)] + \sigma_{w}^{2} - 2P_{0}E[u\beta(u)]} - \alpha E[\beta^{2}(u)] & \text{if } E[u\beta(u)] \leq 0\\ \\ \frac{E[\beta^{2}(u)] + \sigma_{w}^{2} - (E[u\beta(u)])^{2}}{P_{0}^{2} + E[\beta^{2}(u)] + \sigma_{w}^{2} + 2P_{0}E[u\beta(u)]} - \alpha E[\beta^{2}(u)] & \text{if } E[u\beta(u)] \geq 0 \end{cases}$

Noting that the derivative of this expression with respect to $E[u\beta(u)]$ is increasing for negative $E[u\beta(u)]$ and decreasing for positive $E[u\beta(u)]$, we find that the unique maximum is at $E[u\beta(u)] = 0$.

To further maximize with respect to $E[\beta^2(u)]$ we take the derivative and get

$$E[\beta^{2}(u)] = \frac{P_{0}}{\alpha^{\frac{1}{2}}} - (P_{0}^{2} + \sigma_{w}^{2}) \quad \text{if } (\frac{P_{0}}{\alpha^{\frac{1}{2}}}) > (P_{0}^{2} + \sigma_{w}^{2})$$

(i.e., Region R_{13}^2), and zero otherwise (i.e., Region R_{13}^1).

Also note that with the encoding-decoding policy restricted to be linear, the cost obtained by using the jammer's maximizing policy is the same as the maximin value for Game G_{13} .

(c) Game G_{24}

With $\gamma(u) = \in u$ and $\delta(z) = \Delta z$

$$J_{4}|_{\Delta \bullet} = k_{0} + \frac{E[\beta^{2}(u)] + \sigma_{w}^{2} - (E[u\beta(u)])^{2}}{\epsilon^{2} + E[\beta^{2}(u)] + \sigma_{w}^{2} + 2\epsilon E[u\beta(u)]}$$

and

$$I_{4|\Delta^{*}, \epsilon^{*}} = k_{0} \epsilon^{*^{2}} + \frac{E[\beta^{2}(u)] + \sigma_{w}^{2} - (E[u\beta(u)])^{2}}{\epsilon^{*^{2}} + E[\beta^{2}(u)] + \sigma_{w}^{2} + 2\epsilon^{*}E[u\beta(u)]}$$

with $sgn(\in *) = -sgn E[u\beta(u)]$.

ð

X

X

Ŕ

X

3.42

2

Ŕ

We differentiate $J_4|_{\Delta^*, \in^*}$ with respect to $E[u\beta(u)]$ to find that the maximizing value is $E[u\beta(u)] = 0$. Further since $J_4|_{\Delta^*, \in^*}$ is an increasing function of $E[u\beta(u)]$, its maximum over $\beta(u)$ subject to $E[u\beta(u)] \leq P_J^2$ is attained at $E[\beta^2(u)] = P_J^2$. Any second-order random variable with the appropriate variance may be used as the maximinimizing solution, yielding a cost which is the same as the maximin value for Game G_{14} .

We have thus shown that for each of the games the expression on the right side of the equality in (7.46) is J_{1K} , i.e., we have

$$\mathbf{J}_{\mathbf{2K}} \leq \mathbf{J}_{\mathbf{1K}} \,. \tag{7.47}$$

Now using (7.47), along with (7.45), we find that $J_{2K} = J_{1K}$, and the strategies indicated are indeed the maximin strategies.

_	_
÷.	- 1
	_

7.6. Conclusion

In this chapter general classes of communication games have been studied, which include situations where the encoder mapping may be random or restricted to be deterministic, the jammer may tap the input to the encoder or the input to the channel, and the power constraints on the encoder and the jammer may be hard or soft. For each of the above cases, minimax, maximin or saddle-point solutions have been provided.

The important results obtained are

Ş Ņ 10 Ŕ Σð 277 X 22 $\frac{7}{3}$ Ŗ X 73 •

182

- (i) Games of Type 1 where the encoder structure is allowed to be probabilistic and the jammer taps the input to the encoder, admit saddle-point solutions under all fidelity criteria considered.
- (ii) Games of Type 2 where the encoder structure is restricted to be deterministic do not admit a saddle-point solution; however, both minimax and maximin strategies may be found in this case.
- (iii) The minimax value for a game of Type 2 coincides with the minimax value of the corresponding game of Type 3, and the maximin value for a game of Type 2 coincides with the maximin (saddle-point) value of the corresponding game of Type 1.

The least favorable jamming noise for games of Type 1 is Gaussian, and this adds to the list of results previously available in the literature where the Gaussian distribution has been shown to be extremal. However, it is important to note that for games of Type 1 the least favorable jamming noise is independent of the input, whereas for games of Type 3 this least favorable noise is correlated with the output of the encoder.

It is noteworthy that the analysis here cannot be trivially extended when the input sequence is vector valued or the number of channels is more than one, since the counterpart of the standard Gaussian test channel does not admit a simple linear coding scheme in the vector case.

DEDECKING CONTRACTOR

CHAPTER 8

RECAPITULATION AND CONCLUSIONS

In this thesis we considered the problem of simultaneously designing communication and control strategies for decentralized systems. The main thrust was towards identifying classes of problems with linear dynamics, quadratic loss functionals and Gaussian statistics for which the optimality of linear strategies could be established. The general approach used consisted of first finding a lower bound on the cost, and then constructing strategies which achieved this lower bound. For some instances of the cases in which linear strategies failed to provide globally optimal solutions, explicit nonlinear strategies were obtained to demonstrate the inferiority of linear designs.

In Chapter 1 we introduced scenarios in which the simultaneous design of communication strategies and control policies may be desired, and provided a discussion of various types of information structures associated with stochastic team problems. Some issues of computational complexity, which arise when the numerical derivation of the optimal team solution is attempted, were also discussed in Chapter 1.

In Chapter 2 we formulated and analyzed some fundamental classes of stochastic team problems with two decentralized agents. We identified those instances of the general problem for which the optimal solutions are linear. It was shown that if the first agent observes an uncorrupted version of the variable to be transmitted, or if the channel noise is uncorrelated with the input, then the decentralized team problem admits an optimal solution which is linear in the observation variables, and the linear coefficients may be found by solving for the roots of the fifth-order polynomial. For some instances where the first agent observes a noise-corrupted version of the variable to be transmitted, with

183

į,

R

ŝ

1

নিয

~

5

184

the channel noise also correlated with this variable, we provided nonlinear strategies that outperform the optimal linear strategies. We also commented on some aspects of the difficulties associated with Witsenhausen's problem, which is one of the most important and most referenced counterexamples in stochastic control, refuting the common belief (prior to 1968) that all linear quadratic Gaussian control problems admit linear solutions.

In Chapter 3 we considered stochastic dynamic team problems where at each step two consecutive decisions must be taken, one being what information-bearing signal to transmit, and the other regarding what control action to exert. Such problems arise in the simultaneous optimization of both the observation and the control sequences in stochastic systems. We solved the problem completely for first-order systems under quadratic cost criteria. This was done by first constructing an equivalent problem having a cost function consisting of a sum of squared differences, and then solving this equivalent problem by using some bounds from Information Theory. For cases with hard power constraints, it was shown that the optimum measurement strategy is to linearly amplify the innovation at each stage to the maximum permissible power level. For cases with soft power constraints, the structure of the solution was found to be similar, with the optimum power levels being found via solving a nonlinear optimal control problem, this in turn being done by using a dynamic program. The results were then extended to cases with an infinite time horizon and a discounted cost functional, and the existence of optimal stationary policies for these problems was established.

In Chapter 4 we considered stochastic dynamic decision problems requiring simultaneous optimization of both the observation and the control sequences for second- and higher-order systems under quadratic cost criteria. We showed that for some of the simplest of such problems, involving a second-order system, the optimal linear strategies may problem.

be outperformed by appropriately chosen nonlinear strategies. We considered optimality over the affine class for problems involving a general j-th order model, and showed that within this class, the optimal strategy for the hard power constraint problem consists of transmitting the innovation linearly, at each stage, at the maximum permissible power level. For the soft power constraint version the structure of the solution was found to be similar, with the optimal power levels being found via solving a nonlinear optimal control

In Chapter 5 we generalized the results on decentralized, two-person teams obtained in Chapter 2, by allowing the action of one agent to be transmitted to the other agent through a number of noisy channels simultaneously, instead of being transmitted through a : ingle noisy channel. We showed that if all channel noises are independent of the input variable, then linear strategies are optimal, even if the first agent observes a noise corrupted version of the input, the linear coefficients being found through a related parameter optimization problem. We further showed that when the channel noises are indeed correlated with the input variable, then there are instances in which the strategies which are optimal over the affine class may be outperformed by nonlinear strategies, even when the first agent observes an uncorrupted version of the input variable.

In Chapter 6, the results of Chapter 2 were generalized to cases with more than two decision makers. We considered problems with a single transmitting agent and multiple receiving agents, problems with multiple transmitting as well as receiving agents, and finally problems with multiple transmitting agents and a single receiving agent. For problems involving a single transmitting agent and multiple receiving agents, we showed that the optimal strategies are linear when either the first agent observes an uncorrupted version of the input, or when all channel noises are independent of the input. For problems

185

ᠣᡆᡯᡊᡆᠴᡄᢌᠣᢛᠬᡄᢌᡗᠣᠷᢉᠴᡊᠺᡯᡗᠴᡪᡗᠴᡪᢉᢧᡪᡗᠴᠺᡘᢣ᠋ᠺᠰᡗ᠘ᠰᡗ᠘ᠰᡗᢣ᠘ᢣ᠘ᢣᠫᢣ᠕ᢣᠺᢣᠺᠽᡬᡚᡚᡬ᠘ᢢᡚᢢᡚᢓ

2 8 X Ŕ R <u>y</u> Ľ, 67 3 ст, . Ċ, Я R kî, Ś ٨.

186

with multiple transmitting as well as receiving agents we again found that the optimal strategies are linear when either al' transmitting agents observe an uncorrupted version of the variable to be transmitted or when all channel noises are independent of this variable. However, for the simplest classes of problems involving multiple transmitting agents and a single receiving agent, we found instances in which the strategies which are optimal over the affine class may be outperformed by appropriately chosen nonlinear strategies, even when the first agent observes an uncorrupted version of the input to be transmitted, and the channel noises are independent of the input. For this case, we provided strategies which are optimal within the affine class.

C

In Chapter 7, we allowed incomplete statistical description of the channel used to transmit measurements between the decentralized agents, and sought optimal solutions under a worst-case scenario. Assuming the unknown part of the channel noise to be controlled by an adversary or "jammer," we viewed the problems as zero-sum games. We considered a number of cases depending on whether there were hard power constraints or soft power constraints on the decision variables. The unknown channel noise was allowed to be correlated with either the input or the output of the encoder, i.e., the jammer was assumed to have the ability to tap the channel. We found that if the encoder is probabilistic and the jammer taps the input to the encoder, then saddle-point solutions exist under all fidelity criteria. If, however, the encoder structure is restricted to be deterministic, then saddle-point solutions do not exist, but both maximin and minimax strategies may be found. The maximin value here was found to be the same as the maximin value of a corresponding game where the encoder is allowed to be random, with the jammer still tapping the input to the encoder; and the minimax value here was found to be

8 X 8 8 8 ŝ Æ 3 F ł 8 X 0

be outperformed by appropriately chosen nonlinear strategies. We considered optimality over the affine class for problems involving a general j-th order model, and showed that within this class, the optimal strategy for the hard power constraint problem consists of transmitting the innovation linearly, at each stage, at the maximum permissible power level. For the soft power constraint version the structure of the solution was found to be similar, with the optimal power levels being found via solving a nonlinear optimal control problem.

In Chapter 5 we generalized the results on decentralized, two-person teams obtained in Chapter 2, by allowing the action of one agent to be transmitted to the other agent through a number of noisy channels simultaneously, instead of being transmitted through a single noisy channel. We showed that if all channel noises are independent of the input variable, then linear strategies are optimal, even if the first agent observes a noise corrupted version of the input, the linear coefficients being found through a related parameter optimization problem. We further showed that when the channel noises are indeed correlated with the input variable, then there are instances in which the strategies which are optimal over the affine class may be outperformed by nonlinear strategies, even when the first agent observes an uncorrupted version of the input variable.

In Chapter 6, the results of Chapter 2 were generalized to cases with more than two decision makers. We considered problems with a single transmitting agent and multiple receiving agents, problems with multiple transmitting as well as receiving agents, and finally problems with multiple transmitting agents and a single receiving agent. For problems involving a single transmitting agent and multiple receiving agents, we showed that the optimal strategies are linear when either the first agent observes an uncorrupted version of the input, or when all channel noises are independent of the input. For problems

the same as the minimax value of a corresponding game where the jammer taps the output of the encoder.

The problems studied in this thesis can be viewed as important prototype problems, which could be considered essential building blocks for a general theory of multistage distributed decision making under nonclassical information, and possibly partial statistical description. A major focus has been on the question: Are all stochastic team problems which involve simultaneous communication and control, and hence exhibit nonclassical information patterns, inherently difficult and complex? We have shed some light on this question, obtained fundamental results for two-stage stochastic teams, and have made some contributions towards the development of a general theory for multistage (finite and infinite horizon) stochastic control and team problems with nonclassical information, where the control (decision) variable affects not only the state trajectory but also the quality of information available to the decision makers.

 \overline{N}

188

APPENDIX A

REFORMULATION OF THE TEAM PROBLEM

In this appendix we show how the squares may be completed to obtain the equivalent problems PS and PH from the problems PS° and PH°, respectively.

We first note that for k > i we may write

$$x_{k}x_{i} = \sum_{m=0}^{i} \delta_{k,i,m}x_{m}^{2} + \sum_{n=i}^{k-1} p_{k,i,n,i}v_{n}x_{i} + \sum_{n=0}^{i-1} (\sum_{m=n-(j-2)}^{n} p_{k,i,n,m}v_{n}x_{m})$$

where the δ 's are defined recursively as follows:

$$\delta_{100} = \rho_{10}$$

$$\delta_{m,0,0} = \rho_{m,0} + \sum_{k=1}^{m-1} \rho_{m,k} \delta_{k,0,0} \text{ for } 1 < m \le j-1$$

$$\delta_{k,m,p} = \sum_{n=m+1}^{k-1} \rho_{k,n} \delta_{n,m,p} + \sum_{n=p}^{m-1} \rho_{k,n} \delta_{m,n,p} \text{ for } p < m$$

$$\delta_{k,m,m} = \rho_{k,m} + \sum_{n=m+1}^{k-1} \rho_{k,n} \delta_{n,m,m}$$

and the p's are defined by

$$p_{k,m,k-1,m} = -1$$

$$p_{k,m,k-n,m} = \sum_{i=1}^{n-1} \rho_{k,k-i} p_{k-i,m,k-n,m} \text{ for } 2 \le n \le k-m$$

$$p_{k,m,n,i} = \sum_{t=m+1}^{k-1} \rho_{k,t} p_{t,m,n,i} + \sum_{t=0}^{m-1} \rho_{k,t} p_{m,t,n,i} .$$

We define $p_{a,b,c,d}$ to be zero whenever any of the following conditions holds:

(i) d < 0

- (ii) d > Min(b,c)
- (iii) d = b and c > k-1
- (iv) c < b and d < c (j-2)

and $\delta_{a,b,c}$ is defined to be zero whenever c > b.

We therefore have

$$(\overline{a}_{i}x_{i} + \overline{a}_{i-1}x_{i-1} + \dots + \overline{a}_{i-(j-1)}x_{i-(j-1)})^{2}$$

= $\sum_{k=0}^{i-1} v_{k}(\sum_{m=k-(j-2)}^{k} 2r_{i+1,k,m}x_{m}) + \sum_{k=0}^{i} q_{i+1,k}x_{k}^{2}$

• 2222239 • 2222239 • 2222239

where the r's and q's are given as follows:

$$\mathbf{r}_{i+1,n,p} = \sum_{\mathbf{k}=i-(j-2)}^{i} \sum_{\mathbf{m}=i-(j-1)}^{\mathbf{k}-1} \overline{\mathbf{a}}_{\mathbf{k}} \overline{\mathbf{a}}_{\mathbf{m}} \mathbf{p}_{\mathbf{k},\mathbf{m},n,p},$$

$$q_{i+1,n} = \sum_{k=i-(j-2)}^{i} \sum_{m=i-(j-1)}^{k-1} \overline{a}_k \overline{a}_m \delta_{k,m,n} \text{ for } 0 \leq n \leq i-j$$

and

$$q_{i+1,n} = a_n^2 + \sum_{k=i-(j-2)}^{i} \sum_{m=i-(j-1)}^{k-1} \overline{a}_k \overline{a}_m \delta_{k,m,n} \text{ for } i-j < n < i.$$

We can now write

$$\left(\sum_{n=i-(j-1)}^{i} \rho_{i+1,n} x_n\right)^2 = \sum_{k=0}^{i-1} v_k \left(\sum_{m=k-(j-2)}^{k} 2s_{i+1,k,m} x_m\right) + \sum_{m=0}^{i} t_{i+1,m} x_m^2$$

and

α

190

$$a_{i}\left(\sum_{n=i-(j-1)}^{i} b_{i,n}x_{n}\right)^{2} = \sum_{k=0}^{i-1} v_{k}\left(\sum_{m=k-(j-2)}^{k} 2s'_{i+1,k,m}x_{m}\right) + \sum_{m=0}^{i} t'_{i+1,m}x_{m}^{2}.$$

Here $s_{i+1,k,m}$ and $t_{i+1,n}$ are given by expressions identical to those for $r_{i+1,k,m}$ and $q_{i+1,n}$, respectively, with the \overline{a}_n being replaced by $\rho_{i+1,n}$, and $s'_{i+1,k,m}$ and $t'_{i+1,n}$ are given similarly by replacing \overline{a}_n by $b_{i,n}$ and further multiplying by a_i .

The completing of the squares now proceeds as follows. At the first step we have

$$c_{N+1}x_{N+1} + d_{N}v_{N}^{2}$$

= $d_{N}v_{N}^{2} + c_{N+1}(v_{N}^{2} - 2v_{N}(\sum_{m=N-(j-1)}^{N} \rho_{N+1,m}x_{m}) + m_{N}^{2})$
+ $\sum_{n=1}^{N-1} v_{k}(2 - \sum_{n=1}^{k} s_{N+1,N} - x_{m}) + \sum_{n=1}^{N} t_{N+1,m}x_{m}^{2})$

m=0

i.e., we can complete the first square to obtain

k = 0

$$a_{N} = d_{N} + c_{N+1}$$

$$b_{N,N} = \frac{c_{N+1}\rho_{N+1,N}}{a_{N}}$$

$$\vdots$$

$$b_{N,N-(j-1)} = \frac{c_{N+1}\rho_{N+1,N-(j-1)}}{a_{N}}$$

m = k - (j - 2)

Coming to the second step now, we have the following terms containing \boldsymbol{v}_{N-1} :

$$d_{N-1}v_{N-1}^{2} + c_{N}x_{N}^{2} + (t_{N+1,N} - t'_{N+1,N})x_{N}^{2} + 2v_{N-1}(\sum_{m=N-1-(j-2)}^{N-1} (s_{N+1,N-1,m} - s'_{N+1,N-1,m})x_{m}.$$

Letting

$$\mathbf{c'}_{N} \equiv \mathbf{c}_{N} + (\mathbf{t}_{N+1,N} - \mathbf{t'}_{N+1,N})$$

we obtain

Ľ

T

8

S.

8

8

88 98

įΩ.

8

$$a_{N-1} = d_{N-1} + c'_{N}$$

and expanding x_N^2 we get the following expressions for $b_{N-1,n}$'s:

$$b_{N-1,N-1} = \frac{c'_{N}\rho_{N,N-1} - (s_{N+1,N-1} - s'_{N+1,N-1,N-1})}{a_{N-1}}$$

$$\vdots$$

$$b_{N-1,N-j+1} = \frac{c'_{N}\rho_{N,N-j+1} - (s_{N+1,N-1,N-j+1} - s'_{N+1,N-1,N-j+1})}{a_{N-1}}$$

$$b_{N-1,N-j} = \frac{c'_{N}\rho_{N,N-j}}{a_{N-1}}.$$

Moving on to the third step, we have the following expression for terms containing v_{N-2} :

$$\begin{split} d_{N-2}v_{N-2}^{2} + c_{N-1}x_{N-1}^{2} + (t_{N+1,N-1} - t'_{N+1,N-1})x_{N-1}^{2} + (t_{N,N-1} - t'_{N,N-1})x_{N-1}^{2} \\ &+ 2v_{N-2}(\sum_{m=N-2-(j-2)}^{N-2} (s_{N+1,N-2,m} - s'_{N+1,N-2,m})x_{m} \\ &+ 2v_{N-2}(\sum_{m=N-2-(j-2)}^{N-2} (s_{N,N-2,m} - s'_{N,N-2,m})x_{m} . \end{split}$$

Letting

$$c'_{N-1} \equiv c_{N-1} + (t_{N+1,N-1} - t'_{N+1,N-1}) + (t_{N,N-1} - t'_{N,N-1})$$

we obtain



Proceeding in a similar fashion we can obtain the general expression for the a's and b's as follows:

$$a_{k-1} = c'_{k} + d_{k-1}$$

where

$$c'_{k} = c_{k} + \sum_{n=k+1}^{N+1} (t_{n,k} - t'_{n,k});$$

for
$$k-j+1 \le m \le k-1$$
 we have

$$b_{k-1,m} = \frac{c'_{k}\rho_{k,m} - \sum_{n=k+1}^{N+1} (s_{n,k-1,m} - s'_{n,k-1,m})}{a_{k-1}}$$

and

This completes the first step of the transformation and we now have a cost functional of the form

 $b_{k-1,k-j} = \frac{c'_k \rho_{k,k-j}}{2}$.

 $E[\sum_{i=0}^{N} a_{i}(v_{i} - \sum_{k=0}^{j-1} b_{i,i-k} x_{i-k})^{2}] + \bar{c}_{N}$

where \overline{c}_N is a constant given by

$$\overline{c}_{N} = \sum_{i=0}^{N} c'_{i+1} \sigma_{m_{i}}^{2} + c'_{0} \sigma_{m_{0}}^{2}$$

8 88 50 8 X Ś R

To complete the transformation, we now need a redefinition of the v's. Towards this end, we first note that with \tilde{x} 's defined by

$$\tilde{\mathbf{x}}_{i+1} = \sum_{k=0}^{j-1} \rho_{i+1,i-k} \tilde{\mathbf{x}}_{i-k} + \mathbf{m}_i$$
,

we have for i=0,1,2,...

 $\mathbf{x}_{i+1} = \tilde{\mathbf{x}}_{i+1} - \overline{\mathbf{v}}_i$

where

$$\overline{\mathbf{v}}_{i} = \mathbf{v}_{i} + \sum_{k=0}^{j-1} \rho_{i+1,i-k} \overline{\mathbf{v}}_{i-(k+1)} \,.$$

Therefore, we have

$$v_{i} - \sum_{\substack{k=0 \ j=1}}^{j-1} b_{i,i-k} x_{i-k}$$

= $v_{i} - \sum_{\substack{k=0 \ j=1}}^{j-1} b_{i,i-k} (\tilde{x}_{i-k} - \bar{v}_{i-k-1})$
= $v_{i} + \sum_{\substack{k=0 \ j=1}}^{j-1} b_{i,i-k} \bar{v}_{i-k-1} - \sum_{\substack{k=0 \ k=0}}^{j-1} b_{i,i-k} \tilde{x}_{i-k}$
= $\sum_{\substack{k=0 \ k=0}}^{j-1} b_{i,i-k} \tilde{x}_{i-k}$

and hence, the new \tilde{v} 's are defined in terms of the original v's by

 $\sum_{i=1}^{N}$

3

න0 දැර

88

3

<u>Š</u>

8

Ş

кН ----

Å,

78

8

Ş

Š

2 M

ŋ]

N

\$*****\$*******\$*******\$*******\$******\$*****\$\$*****\$**

$$\tilde{\mathbf{v}}_{i} = \mathbf{v}_{i} + \sum_{k=0}^{j-1} \mathbf{b}_{i,i-k} \overline{\mathbf{v}}_{i-k-1} ,$$

which is an invertible transformation.

APPENDIX B

THE GAUSSIAN CHANNEL WITH SIDE INFORMATION AT THE DECODER

Consider Problem P1 defined below where the objective is to simultaneously design the measurement policy γ and the control policy δ to minimize a quadratic cost.

Problem P1

Minimize
$$J(u,v) = E[k_0 u^2 + (v-x)^2]$$
 (B.1)

where

$$\mathbf{u} = \mathbf{y}(\mathbf{x}) \tag{B.2a}$$

$$\mathbf{v} = \delta(\mathbf{x}_1, \mathbf{x}_2) \tag{B.2b}$$

and

$$\mathbf{x}_1 = \mathbf{x} + \mathbf{w}_1 \tag{B.3a}$$

$$\mathbf{x}_2 = \mathbf{u} + \mathbf{w}_2 \,. \tag{B.3b}$$

Here x, w_1 and w_2 are zero mean, independent, Gaussian random variables, with variances σ_x^2 , $\sigma_{w_1}^2$ and $\sigma_{w_2}^2$, respectively.

To facilitate the study of Problem P1, we formulate a related problem, P2, which is the hard constraint version of Problem P1, i.e., the term $E[k_0u^2]$ is removed from the cost functional and the further stipulation is made that

$$\mathbf{E}[\boldsymbol{\gamma}^2(\mathbf{x})] \leqslant \mathbf{P}^2 \,. \tag{B.4}$$

We therefore have Problem P2 below.

Problem P2

$$\underset{\gamma,\delta}{\text{Minimize } J_2(u,v) = E[(v-x)^2]}$$
(B.5)

given (2), (3) and (4).

Problem P2 is represented schematically in Figure B.1 below. Note that this problem may be viewed as one of designing encoding and decoding strategies in the presence of side information at the decoder. If the side information were absent, then this would be a special case of the problem studied in Bansal and Basar [1987a], without a product term between the decision variables, and the linear solution would be optimal. For the above problem, however, the linear solution is not optimal as we shall elucidate in the sequel.

Encoding-decoding problems with side information have been the subject of some previous investigations. The rate distortion for source coding with side information at the decoder has been studied for general sources in Wyner [1978]. It has been shown that if the source and side information are jointly Gaussian, then the minimum rate (in the usual Shannon sense) required for encoding the source at a distortion level about d, is equal to



Figure B.1. Schematics of Problem P2.

the rate which would be required if the encoder (as well as the decoder) had access to this side information. Since the coding theorem uses the fact that long ergodic sequences are asymptotically typical and therefore can be encoded into sequences that have the distribution which achieves capacity, Wyner's result implies that if block encoding and arbitrary delays were permissible, then the least mean square error for Problem P2 would be the same as that for the system depicted in Figure B.2 below. Note that the only difference between Figures B.1 and B.2 is that in the latter the encoder too has access to the side information.

SULUI .

The situations we envisage are control theoretic applications, where the encoder output represents system measurements and the decoder has to control the system in real time. Thus, bounds which are asymptotically tight are of little use in our application, where sample by sample transmission is required. Real-time coding-decoding problems have also been studied earlier, notably in Walrand and Varaiya [1983], where *finite* sources are considered and feedback information has been shown to be useful for the



Figure B.2. The encoder has side information also.

197

K

8

ß

N N

ale P

Č,

22

ŝ

R

2

<u>ارد</u>

Property of

198

causal encoding-decoding problem. In the problems of interest to us here, feedback is not permitted.

Wyner's [1978] result implies that if block encoding were permitted, then the least mean square error for Problem P2 would be the same as in the case illustrated by Figure B.2. For the situation depicted in Figure B.2 we have

$$I(x;x_{1}) = \frac{1}{2} \log(\frac{\sigma_{x}^{2} + \sigma_{w_{1}}^{2}}{\sigma_{w_{1}}^{2}})$$
(B.6)

and

ķ

ſ

ſ

$$I(x;x'_{2}|x_{1}) \leq \frac{1}{2} \log(\frac{P^{2} + \sigma_{w_{2}}^{2}}{\sigma_{w_{2}}^{2}}).$$
(B.7)

Now since

$$I(x;x_1,x_2) = I(x;x_1) + I(x;x_2 + x_1),$$
(B.8)

we have

$$I(x;x_{1},x_{2}') \leq \frac{1}{2} \log(\frac{(\sigma_{x}^{2} + \sigma_{w_{1}}^{2})(P^{2} + \sigma_{w_{2}}^{2})}{\sigma_{w_{1}}^{2}\sigma_{w_{2}}^{2}}).$$
(B.9)

The rate distortion function for a Gaussian source of variance σ_x^2 under a mean square distortion criterion is

$$R(d) = Max(0, \frac{1}{2} \log \frac{\sigma_x^2}{d}).$$
 (B.10)

Further,

 $I(x;x_{1},x_{2}') \ge I(x;v') \ge \frac{1}{2} \log \frac{\sigma_{x}^{2}}{E((x-v')^{2})}$ (B.11)

which implies that

 $E[(x-v')^{2}] \ge \frac{\sigma_{x}^{2} \sigma_{w_{1}}^{2} \sigma_{w_{2}}^{2}}{(\sigma_{x}^{2} + \sigma_{w_{1}}^{2})(P^{2} + \sigma_{w_{2}}^{2})} \triangleq \bar{d}.$ (B.12)

Now if we use the causal policy

$$u(x) = \lambda'(x - E(x | x_1))$$
(B.13)

Section of the sectio

where λ' satisfies

$$(\lambda')^{2} = \frac{P^{2}(\sigma_{x}^{2} + \sigma_{w_{1}}^{2})}{(\sigma_{x}^{2} \sigma_{w_{1}}^{2})}$$
(B.14)

then this least possible distortion \overline{d} is attained with equality, implying that the linear strategy indicated is optimal for the problem with feedback.

In Problem P2 feedback is not available, and hence the optimal linear strategy consists of using

$$u(x) = \lambda x \tag{B.15a}$$

$$v = E[x | x_1, x_2]$$
 (B.15b)

which yields the mean square distortion d* given by

$$d^* = \frac{\sigma_x^2 \sigma_{w_1}^2 \sigma_{w_2}^2}{P^2 \sigma_{w_1}^2 + \sigma_{w_2}^2 (\sigma_x^2 + \sigma_{w_1}^2)}.$$
 (B.16)

Clearly $d^* > \overline{d}$, since $P^2 \sigma_x^2 > 0$. This is not unexpected since in the case of \overline{d} , additional information was used at the encoder. As discussed earlier, if arbitrarily large delays are allowed, then the least distortion for P2 is also \overline{d} , hence strategies may be constructed via block encoding which would improve upon the case of linear sample by sample transmission.

The important issue now is whether a reduction in the least mean square error d* (obtained via linear coding) is possible *without* block encoding. We now show that the optimal causal linear strategy may be outperformed by a causal nonlinear strategy, i.e., without block encoding and thus with no delay.

We first provide an example where the optimum linear strategy for Problem P2 may be outperformed by an appropriately chosen nonlinear policy. Towards this end, we propose the following design:

$$u = x + \epsilon \operatorname{sgn} x$$
 (B.17)

$$\mathbf{v} = \begin{cases} (\mathbf{x}_1 + \mathbf{x}_2 - \epsilon)/2 & \text{if } \mathbf{x}_1 \ge 0\\ \\ (\mathbf{x}_1 + \mathbf{x}_2 + \epsilon)/2 & \text{if } \mathbf{x}_1 < 0 \end{cases}$$
(B.18)

We next find the mean square error using the policy proposed in (B.17) and (B.18). We have

$$E[(x-v)^{2}] = E[(x-v)^{2} | x_{1} \ge 0] \operatorname{Prob}[x_{1} \ge 0] + E[(x-v)^{2} | x_{1} < 0] \operatorname{Prob}[x_{1} < 0]$$

= E[(x-v)^{2} | x_{1} \ge 0].

Now,

$$E[(x-v)^2 | x_1 \ge 0]$$

 201

$$= E[(x-v)^2 | x_1 \ge 0, x \ge 0] \operatorname{Prob}[x \ge 0 | x_1 \ge 0]$$

+
$$E[(x-v)^2 | x_1 \ge 0, x < 0] \operatorname{Prob}[x < 0 | x_1 \ge 0].$$

Since

 $E[(x-v)^{2} | x_{1} \ge 0, x \ge 0] = \frac{\sigma_{w_{1}}^{2} + \sigma_{w_{2}}^{2}}{4}$

and

$$E(x-v)^{2} | x_{1} \ge 0, x < 0] = \frac{\sigma_{w_{1}}^{2} + \sigma_{w_{2}}^{2}}{4} + \epsilon^{2},$$

we have

$$E[(x-v)^{2} | x_{1} \ge 0] = \frac{\sigma_{w_{1}}^{2} + \sigma_{w_{2}}^{2}}{4} + \epsilon^{2} \operatorname{Prob}[x < 0 | x_{1} \ge 0].$$

Now $Prob(x < 0 | x_1 \ge 0)$ may be easily computed, since the joint density of x and x_1 is known:

$$p_{x,x_1}(\cdot,\cdot) = \frac{1}{2\pi\sigma_x\sigma_{w_1}} \exp\left\{-\frac{(x_1-x)^2}{2\sigma_{w_1}^2} - \frac{x^2}{2\sigma_x^2}\right\}$$

Further,

$$Prob(x < 0 | x_1 \ge 0) = 1 - Prob(x > 0 | x_1 \ge 0)$$

$$\operatorname{Prob}(x \ge 0 \mid x_1 \ge 0) = \frac{\operatorname{Prob}(x \ge 0, x_1 \ge 0)}{\operatorname{Prob}(x_1 \ge 0)} \stackrel{\Delta}{=} \frac{1}{1/2} = 2I$$

where

$$I = \int_{0}^{\infty} \int_{0}^{\infty} p_{x,x_1} dx dx_1 .$$

Letting $\sigma_{w_1}^2 = 1$ we get

$$I = \int_{0}^{\infty} \frac{1}{2\pi\sigma_{x}} \exp(-\frac{x^{2}}{2\sigma_{x}^{2}}) dx \int_{0}^{\infty} \exp(-\frac{(y-x)^{2}}{2}) dy$$

= $\int_{0}^{\infty} \frac{1}{\sqrt{2\pi}\sigma_{x}} \exp(\frac{-x^{2}}{2\sigma_{x}^{2}}) dx \frac{1}{\sqrt{2\pi}} \int_{-x}^{\infty} e^{-t^{2}/2} dt$
= $\int_{0}^{\infty} \frac{1}{\sqrt{2\pi}\sigma_{x}} \exp(\frac{-x^{2}}{2\sigma_{x}^{2}}) \phi(-x) dx$
= $\int_{0}^{\infty} \frac{1}{\sqrt{2\pi}} \exp(\frac{-z^{2}}{2}) \phi(-\sigma_{x}z) dz$.

Thus, I may be evaluated and the error using nonlinear policies:

$$C_{NL} \triangleq \frac{1 + \sigma_{w_2}^2}{4} + \epsilon^2 (1 - 2I)$$

may be found, and hence the mean square error using linear strategies and that using this nonlinear strategy may now be compared.

We now provide some examples where the optimum linear strategy for Problem P2 is outperformed by a nonlinear strategy of the form given by (B.17) and (B.18). We then illustrate how these results may be used to construct instances of Problem P1 where a nonlinear strategy may be used to outperform the optimal linear strategy.

Consider Problem P2 with parameter values $\sigma_x^2 = 100.0$, $\sigma_{w_1}^2 = \sigma_{w_2}^2 = 1.0$. Letting $P^2 = 85.0423$, we get $\epsilon = -1$ and the nonlinear strategy achieves a mean square error of 0.53172. The optimum linear strategy at the above power level attains a mean square error of 0.53751, which is inferior to that attained by the nonlinear policy.

Table B.1 below gives other instances of Problem P2 where the nonlinear policy outperforms the optimum linear policy, the third column indicating the improvement (C_L-C_{NL}) , i.e., the difference between the mean square errors for the two cases, and the last column indicating percentage improvement over the optimum linear design. In all the cases considered in Table B.1, we have used $\sigma_{w_1}^2 = \sigma_{w_2}^2 = 1.0$.

TABLE B.1. SOME INSTANCES OF PROBLEM P2 WHERE NONLINEAR STRATEGIES OUTPERFORM OPTIMUM LINEAR STRATEGIES.

σ	-e	Improvement	% change
4.0	1.00	0.001375	0.24
6.0	1.10	0.001736	. 0.31
6.0	0.90	0.009627	1.74
6.0	0.80	0.011973	2.19
6.0	0.70	0.013262	2.46
6.0	0.50	0.012691	2.41
10.0	0.80	0.009148	1.73
10.0	0.60	0.009994	1.92
10.0	0.55	0.009813	1.89

We now construct an instance of Problem P1 by finding k_0 which corresponds to the appropriate hard power constraint. Returning to the example with $\sigma_x^2 = 100.0$, $\sigma_{w_1}^2 = \sigma_{w_2}^2 = 1.0$, we note that when the policies are linear, the optimum power level is given by

$$P^{*2} = \arg \min_{P^2 \ge 0} \left[k_0 P^2 + \frac{\sigma_x^2 \sigma_{w_1}^2 \sigma_{w_2}^2}{P^2 \sigma_{w_1}^2 + \sigma_{w_2}^2 (\sigma_x^2 + \sigma_{w_1}^2)} \right]$$

203

Ø

Ň

0

X

R

5

ĝ

<u>.</u>6

8

Ċ.

which implies that k_0 and P^{*2} satisfy

$$k_0 - \frac{100}{(P^{*2} + 101)^2} = 0,$$

and for the case in consideration with $P^2 = 85.0423$, we may let

$$k_0 = \frac{100}{\left(186.0423\right)^2} = 0.0028892 \; .$$

Therefore, for Problem P1 with parameter values $k_0 = 0.0028892$, $\sigma_x^2 = 100.0$, $\sigma_{w_1}^2 = \sigma_{w_2}^2 = 1.0$, the optimum cost over the linear class is given by (0.0028892 x 85.0423 + 0.53751 =) 0.78321 whereas a nonlinear policy of the form (B.17) and (B.18) with $\epsilon = -1$ attains 0.77742.

REFERENCES

Abramson, N. [1963], Information Theory and Coding, McGraw-Hill, NY.

Ash, R. [1965], Information Theory, Interscience Publishers, NY.

1

R

1. A.A.

R,

È.

K

ġ

- Athans, M. [1972], "On the determination of optimal costly measurement strategies for linear stochastic systems," Automatica, 8, 397-412.
- Bagchi, A. and T. Başar [1980], "Team decision theory for linear continuous time systems," *IEEE Trans. Automat. Contr.*, AC-25, 1154-1161.
- Bansal, R. and T. Basar [1987a], "Stochastic teams with nonclassical information revisited — when is an affine law optimal?," *IEEE Trans. Automat. Contr.*, AC-32, 554-559.
- Bansal, R. and T. Basar [1987b], "Solutions to a class of linear-quadratic-Gaussian (LQG) stochastic team problems with nonclassical information," Systems and Control Letters, 9, 125-130.
- Bansal, R. and T. Basar [1989], "Communication games with partially soft power constraints," JOTA, 61, to appear.
- Basar, T. [1978], "Decentralized multicriteria optimization of linear stochastic systems," *IEEE Trans. Automat. Contr.*, AC-23, 233-243.
- Basar, T. [1980], "A trace minimization problem with applications in joint estimation and control under nonclassical information," JOTA, 31, 343-359.
- Basar, T. [1983], "The Gaussian test channel with an intelligent jammer," *IEEE Trans.* Inform. Theory, IT-29, 152-157.
- Basar, T. and J. B. Cruz, Jr. [1982], "Concepts and methods in multiperson coordination and control," in Optimization and Control of Dynamic Operational Research Models. S. G. Tzafestas (ed.), North Holland, Chapter 11, 351-394.
- Basar, T. and G. J. Olsder [1982], Dynamic Noncooperative Game Theory, Academic Press, NY.
- Basar, T. and Y. W. Wu [1985], "A complete characterization of minimax and maximin encoder-decoder policies for communication channels with incomplete statistical description," *IEEE Trans. Inform. Theory*, IT-31, 482-489.
- Basar, T. Ü., B. Sankur and H. Abut [1980], "Performance bounds and optimal linear coding for discrete-time multichannel communication systems," *IEEE Trans. Inform. Theory*, IT-26, 212-217.

Berger, T. [1971], Rate Distortion Theory, Prentice Hall, NJ.

Bertsekas, D. [1987], Dynamic Programming, Prentice Hall, NJ.

A CONTRACTOR O STATE O

- Chu, K. C. [1978], "Designing information structures for quadratic decision problems," JOTA, 25, 139-160.
- Dobrushin, R. and B. Tysbakov [1962], "Information transmission with additional noise," *IRE Trans. Inform. Theory*, IT-8 293-304.

Gallager, R. G. [1968], Information Theory and Reliable Communication, John Wiley, NY.

- Herring, K. D. and J. L. Melsa [1974], "Optimum measurements for estimation," *IEEE Trans. Automat. Contr.*, AC-19, 264-266.
- Ho, Y. C. [1980], "Team decision theory and information structures," *Proc. IEEE*, 68, 644-654.
- Ho, Y. C. and T. S. Chang [1980], "Another look at the nonclassical information problem," *IEEE Trans. Automat. Contr.*, AC-25, 537-340.
- Ho, Y. C. and K. C. Chu [1972], "Team decision theory and information structures in optimal control problems Part I," *IEEE Trans. Automat. Contr.*, AC-17, 15-22.
- Kagan, A. M., V. Y. Linnik and C. R. Rao [1973], Characterization Problems in Mathematical Statistics, Wiley, NY, 408-410.
- Kumar, P. R. and P. P. Varaiya [1986], Stochastic Systems: Estimation, Identification and Adaptive Control, Prentice Hall, NJ.
- Kurtaran, B. [1975], "A concise derivation of the LQG one-step-delay sharing problem solution," *IEEE Trans. Automat. Contr.*, AC-20, 808-810.
- Lafortune, S. [1985], "On stochastic optimal control problems with selection among different costly observations," Elec. Res. Lab., U. of California, Berkeley, Mem. No. UCB/ERL M85/99.
- Mehra, R. K. [1976], "Optimization of measurement schedules and sensor designs for linear dynamic systems," *IEEE Trans. Automat. Contr.*, AC-21, 55-64.
- Meier, L., J. Peschon and R. M. Dressler [1967], "Optimum control of measurement subsystems," *IEEE Trans. Automat. Contr.*, AC-12, 528-336.
- Ovseevich, I. A. and M. S. Pinsker [1958], "Information transmission rate, capacity of a multipath system and reception by a linear-operator transformation method," *Radiotechnika*, 14, 9-21.
- Papadimitriou, C. H. and J. N. Tsitsiklis [1982], "On the complexity of designing distributed protocols," Information and Control, 53, 211-218.

- Papadimitriou, C. H. and J. N. Tsitsiklis [1986], "Intractable problems in control theory," SIAM J. Contr. and Optimiz., 24, 639-654.
- Papavassilopoulos, G. P. [1983], "On the optimal choice of measurements in linear quadratic Gaussian team problems," in Proc. 1983 American Contr. Conf., San Francisco, CA, 693-701.
- Pinsker, M. S. [1972], "The information content of observations and asymptotically sufficient statistics," *Problems of Information Transmission*, (Translated from *Problemy Peredachi Informatsii*,) 8, 45-61.
- Radner, R. [1962], "Team decision problems," Ann. Math. Statist., 33, 857-881.

8

8

8

ß

B

Ś

Ŕ

R

80

<u>P</u>

- Sandell, N. and M. Athans [1974], "Solutions of some nonclassical LQG stochastic decision problems," *IEEE Trans. Automat. Contr.*, AC-19, 108-116.
- Shannon, C. E. [1949], "Communication in the presence of noise," Proc. IRE, 37, 10-21.
- Shannon, C. E. [1959], "Coding theorems for a discrete source with a fidelity criterion," *IRE National Convention Record*, Part 4, 142-163.
- Tsitsiklis, J. N. and M. Athans [1985], "On the complexity of decentralized decision making and detection problems," *IEEE Trans. Automat. Contr.*, AC-30, 440-446.
- Varaiya, P. and J. Walrand [1983], "Causal coding and control for Markov chains," Systems and Control Letters, 3, 189-192.
- Walrand, J. C. and P. Varaiya [1983], "Optimal causal coding-decoding problems," *IEEE Trans. Inform. Theory*, IT-29, 814-820.
- Whittle, P. and J. F. Rudge [1976], "Simultaneous communication and control," Adv. Appl. Prob., 8, 365-384.
- Witsenhausen, H. S. [1968], "A counterexample in stochastic optimal control," SIAM J. Contr., 6, 131-147.
- Witsenhausen, H. S. [1971], "On information structures, feedback and causality," SIAM J. Contr., 9, 149-160.
- Witsenhausen, H. S. [1980], "Indirect rate distortion problems," *IEEE Trans. Inform. Theory*, **IT-26**, 518-321.
- Witsenhausen, H. S. [1988], "Equivalent stochastic control problems," Mathematics of Control, Signals and Systems, 1, 3-11.
- Wolf, J. K. and J. Ziv [1970], "Transmission of noisy information to a noisy receiver with minimum distortion," *IEEE Trans. Inform. Theory*, **IT-16**, 406-411.

X 23 <u>v</u>. **5**45 2 $\overline{\mathcal{X}}$ $\hat{\mathbf{x}}$ 53 <u>8</u>2 8 2 S S 64 3 33 N \sim ۶ ۱

- Wonham, W. M. [1968], "On the separation theorem of stochastic control," SIAM J. Contr., 6, 312-326.
- Wozencraft, J. M and I. M. Jacobs [1965], Principles of Communication Engineering, Wiley, NY.
- Wyner, A. D. [1970], "Another look at the coding theorem of information theory, A Tutorial," *Proc. IEEE*, 58, 894-913.
- Wyner, A. D. [1978], "The rate distortion function for source coding with side information at the decoder - II: general sources," Inform. Contr., 38, 60-80.
- Yoshikawa, T. [1975], "Dynamic programming approach to decentralized control problems," *IEEE Trans. Automat. Contr.*, AC-20, 796-797.

NAMON ALCOLO
VITA

Rajesh C. Bansal was the end of the received his B.Tech. in Electrical Engineering from the indian Institute of Technology at Kanpur in 1984, and his M.S. in Electrical Engineering from the University of Illinois at Urbana-Champaign in 1986. Since August 1984 he has been a research assistant in the Decision and Control Laboratory of the Coordinated Science Laboratory at the University of Illinois. His research interests include multiperson optimization, stochastic control and computational complexity.

Rajesh C. Bansal is a member of the Institute of Electrical and Electronics Engineers and the honor society Tau Beta Pi. He has co-authored the following papers:

R. Bansal and T. Basar, "Stochastic Teams with Nonclassical Information Revisited – When is an Affine Law Optimal," *IEEE Transactions on Automatic Control*, AC-32, June 1987, pp 554-559; also in *Proc. 1986 American Control Conference*, Seattle, Washington, June 1986.

R. Bansal and T. Basar, "Solutions to a Class of Linear Quadratic Gaussian Stochastic Team Problems with Nonclassical Information," *Systems and Control Letters*, 9, 2, 1987, pp 125-130; also presented at the 26th IEEE Conference on Decision and Control, Los Angeles, California, Dec. 9-11, 1987.

R. Bansal and T. Basar, "Communication Games with Partially Soft Power Constraints," *Journal of Optimization Theory and Applications*, to appear in 1989.

T. Basar and R. Bansal, "The Theory of Teams: A Selective Annotated Bibliography," Presented at *The Summer School on Dynamic Optimization*, Dublin, Ireland, July 6-17, 1987.

R. Bansal and T. Basar, "Simultaneous Design of Communication and Control Strategies in Stochastic Systems with Feedback," to appear in *Proc. of the 8th International Conference on the Analysis and Optimization of Systems*, Antibes, France, June 8-10, 1988.

R. Bansal and T. Basar, "Solutions to some Stochastic Team Problems and Zero Sum Games with Nonclassical Information Arising in Communication Systems." *International Symposium on Information Theory*, Ann-Arbor, Michigan, Oct. 6-9, 1986.

209

R. Bansal and T. Basar, 'Joint Design of Communication Channels and Decision Policies in Stochastic Teams," *IEEE Control Systems Society Workshop on Current Issues in Decentralized and Distributed Control*, Columbus, Ohio, Sept. 14-15, 1987.

R. Bansal and T. Basar, "Optimum Encoding and Decoding Policies for Jammed Communication Channels with Soft and Partially Soft Power Constraints," 22nd Annual Conference on Information Sciences and Systems, Princeton, New Jersey, March 16-18, 1988.

R. Bansal and T. Basar, "On Problems Exhibiting a Dual Role of Control," to be presented at 27th IEEE Conference on Decision and Control, Austin, Texas, Dec. 7-9, 1988.

R. Bansal and T. Basar, "Joint Estimation and Control for a Class of Stochastic Dynamic Teams," to be presented at 27th IEEE conference on Decision and Control, Austin, Texas, Dec. 7-9, 1988.

Ð.

33

đ۵

ψų,

Ň

3

275

8

X

2

5

3

Ŷ,

ų K

 $\mathcal{C}_{\mathcal{C}}$

3

ल